TARGET CANADA CO.

as Tenant

- and -

HILLCREST HOLDINGS INC., MONTEZ HILLCREST INC., OXFORD PROPERTIES RETAIL HOLDINGS II INC., CPPIB UPPER CANADA MALL INC., OMERS REALTY MANAGEMENT CORPORATION, SQUARE ONE PROPERTY CORPORATION, KINGSWAY GARDEN HOLDINGS INC., LES GALERIES DE LA CAPITALE HOLDINGS INC., IVANHOE CAMBRIDGE II INC., CARREFOUR DE L'ESTRIE HOLDINGS, INC., PLACE LAURIER HOLDINGS INC., MIC MAC MALL LIMITED PARTNERSHIP, OAKRIDGE CENTRE VANCOUVER HOLDINGS, INC., collectively as the Landlord Entities

LEASE TRANSACTION AGREEMENT FEBRUARY 26, 2015

OSLER, HOSKIN & HARCOURT LLP

Article 1 LE	ASE TRANSACTION	2
1.1	Landlord Entity	2
1.2	Object and Intent	2
1.3	Assignments of Certain Agreements	3
1.4	Post-Filing Obligations	
Article 2 CC	ONSIDERATION	4
2.1	Consideration	4
2.2	Surrender of Leases	5
2.3	Landlord Claims	
2.4	Tenant Claims	6
Article 3 AI	DJUSTMENTS	6
3.1	Statement of Adjustments	6
3.2	Adjustments and Absence of Post-Closing Adjustments	6
3.3	Items of Adjustments under the Leases.	7
3.4	Realty Tax Appeals	7
3.5	Utilities	8
Article 4 LA	ANDLORD ENTITIES' ACKNOWLEDGEMENTS	8
4.1	"As Is, Where Is"	8
4.2	Monitor's Capacity	
4.3	Approval Order and other Orders.	9
Article 5 LA	ANDLORD ENTITIES AND TENANT COVENANTS	10
5.1	Assignment until Vacancy Date	10
5.2	Interim Period	10
5.3	Absence of Tenant Estoppels	
5.4	Guarantee	
5.5	Wind-Down or Non-Operation	
5.6	Leasebacks	11
Article 6 RI	EPRESENTATIONS, WARRANTIES & COVENANTS	14
6.1	Tenant's Residency	14
6.2	Landlord Entities' Representations and Warranties	14
6.3	Landlord Entities and Tenant Covenants	15
6.4	Tax Matters	
6.5	Survival of Covenants, Representations and Warranties	17
Article 7 C	ONDITIONS TO CLOSING	17
7.1	Conditions of Closing for the Benefit of the Landlord Entities	17
7.2	Conditions of Closing for the Benefit of the Tenant	18
7.2	Conditions of Closing for the Mutual Benefit of the Parties	18
	osing	
	Closing Date	
8.1 8.2	Closing Documents	19
0.2	Ologing Lovullions	

8.3	Confirmation of Satisfaction of Conditions	20
8.4	1 Fscrow Closing	20
8.5	5 Monitor's Certificate	21
8.6		
Article 9	GENERAL PROVISIONS	22
9.1	Filings and Authorizations	22
9.2	Court Matters	
9.3	3 Termination	
9.4	Automatic Termination - Outside Date	23
9.5		23
Article 10	0 OTHER PROVISIONS	23
10	0.1 Confidentiality	23
10) 2 Time of the Essence	
10) 2 Finitement	
10) A Scionability	
10) 5 Entire Agreement	
10	0.6 Waiver	26
10	0.7 Further Assurances	26
	0.8 Severability	26
	0.9 Governing Law	26
	a man and a second a second and	26
		27
		27
_	0.13 Statute References	27
	0.15 References	27
_	0.16 Number and Gender	
	0.17 Use of the word "including" and "or" etc	
1	0.18 Rusiness Days	
1	0.10 Currency and Payment Obligations	21
1	0.20 Accounting Principles	28
1	0.21 Notice	40
1	0.22 Risk of Loss	28
SCHED	ULE "A" DEFINED TERMS	1
SCHED	OULE "B" LOCATIONS & CORRESPONDING LANDLORD ENTITI	ES 1
SCHED	OULE "B1" LEASES	1
SCHED	OULE "C" REAL PROPERTY INTERESTS	1
SCHED	OULE "D" EXCLUDED ASSETS	1
SCHED	OULE "E" INTENTIONALLY DELETED	
SCHED	OULE "F" SITE PLAN AGREEMENT	
SCHED	OULE "G" PRE-PAID LEASEBACK RENT	I

SCHEDULE "H" INTENTIONALLY DELETED	1
SCHEDULE "I" FORM OF LEASE AMENDMENT AND SURRENDER AGREEMENT	1
SCHEDULE "J1" ITEMS WHICH SHALL BE REMOVED FROM THE PREMISES	1
SCHEDULE "J2" ITEMS WHICH CANNOT BE REMOVED FROM THE PREMISES	
SCHEDULE "K" FORM OF LEASEBACK	
SCHEDULE "L" FORM OF APPROVAL ORDER	1
SCHEDULE "M" ADDRESSES FOR NOTICE	1
SCHEDULE "N" RELEASE OF LANDLORD CLAIMS	1
SCHEDULE "O" RELEASE OF TENANT CLAIMS	1
SCHEDULE "P" RELEASE OF GUARANTEES	1
SCHEDULE "O" SOA	1

LEASE TRANSACTION AGREEMENT

Lease Transaction Agreement dated February 26, 2015 among TARGET CANADA CO. (the "Tenant"), HILLCREST HOLDINGS INC., MONTEZ HILLCREST INC., OXFORD PROPERTIES RETAIL HOLDINGS II INC., CPPIB UPPER CANADA MALL INC., OMERS REALTY MANAGEMENT CORPORATION, SQUARE ONE PROPERTY CORPORATION, KINGSWAY GARDEN HOLDINGS INC., LES GALERIES DE LA CAPITALE HOLDINGS INC., IVANHOE CAMBRIDGE II INC., CARREFOUR DE L'ESTRIE HOLDINGS, INC., PLACE LAURIER HOLDINGS INC., MIC MAC MALL LIMITED PARTNERSHIP, OAKRIDGE CENTRE VANCOUVER HOLDINGS, INC. (collectively, but subject to Section 1.1, the "Landlord Entities").

RECITALS

- A. The Landlord Entities are, respectively, the owner and landlord or ground lessee and sublandlord or head tenant and landlord of the Locations and the Tenant's Leases and lease the Premises at the Locations to the Tenant as more particularly set out in Schedules "B" and "B1" hereto.
- B. The Tenant and certain of its Affiliates applied for and together with the limited partnerships listed on Schedule "A" to the Initial Order (collectively, for the purposes of this Agreement described as the "Applicants") were granted protection by the Ontario Superior Court of Justice (Commercial List) (the "Court") under the Companies' Creditors Arrangement Act, R.S.C. 1985, c. C-36, as amended (the "CCAA"), and Alvarez & Marsal Canada Inc. (the "Monitor") was appointed the monitor of the Tenant, pursuant to an Order of the Court dated January 15, 2015, as amended and restated on February 11, 2015, and as further amended, restated and/or amended and restated from time to time (collectively, the "Initial Order").
- C. The Landlord Entities desire to accept a surrender from the Tenant of each of the Real Property Interests at the Locations, on the terms and conditions set out herein, subject to a Leaseback of all or some of the Premises to the Tenant to allow it to complete the sale and removal of Inventory and FF&E for varying periods of time in accordance with the Initial Order and the Sale Order or as otherwise contemplated herein.
- D. This Agreement and the completion of the Transaction are subject to the issuance of the Approval Order and other conditions, as more particularly discussed herein.
- E. Unless otherwise provided for herein, all capitalized terms when used in this Agreement have the meaning ascribed thereto in Schedule "A".

In consideration of the foregoing and the mutual agreements contained in this Agreement (the receipt and adequacy of which are acknowledged), the parties agree as follows.

ARTICLE 1 LEASE TRANSACTION

1.1 Landlord Entity

- (a) Each Landlord Entity is entering into this Agreement and delivering the documents in connection with the Closing: (A) as an individual party on its own behalf, and not on behalf of another Landlord Entity; (B) on a several basis (as to the respective interest of such Landlord Entity in the Real Property Interests) and not on a joint or joint and several basis.
- (b) In no event shall a Landlord Entity have any obligation or liability in respect of any representations or warranties or any covenants, liabilities, obligations or indemnities made or incurred by another Landlord Entity in or pursuant to this Agreement and/or any of the Closing Documents. In the event of any conflict, inconsistency or ambiguity between the provisions of this Section 1.1 and any other provisions of this Agreement and/or any of the documents delivered in connection with the Closing, the provisions of this Section 1.1 shall prevail.
- Notwithstanding the foregoing or anything else contained herein or elsewhere, each of the Landlord Entities acknowledges and agrees that: (i) each of the Lease Amendment and Surrender Agreements must be completed at the same time as it is agreed that the Tenant's agreement to surrender, and the Landlord Entity's agreement to accept such surrender, apply and extend to all of the Leases and Real Property Interests and do not constitute an agreement in respect of discreet Leases and Real Property Interests independently of the other Leases and/or Real Property Interests; and (ii) the Tenant's recourse is to the entire Deposit and all interest thereon and not to a specific Landlord Entity's interest therein, even if such Landlord Entity is not responsible for a breach or default hereunder.

1.2 Object and Intent

- (a) Subject to the terms and conditions hereof, on the Closing Date:
 - (i) the Tenant hereby agrees to surrender, and the Landlord Entities hereby agree to accept a surrender of, each and every one of the Real Property Interests and the Leases;
 - the Tenant and the Landlord Entities hereby agree to amend, as the Landlord Entities may reasonably require and the Tenant may reasonably agree, any Lease, which amendments shall be conditional upon the successful Closing of the Transaction, and shall take effect immediately prior to the surrender of such Lease, or shall form part of the Lease Amendment and Surrender Agreement for such Lease, provided that:

 (A) such amendments may only relate to the release or cancellation of rights to renew or extend the Lease term; (B) such amendments do not create any greater obligations on the Tenant or on Target which are not released on Closing by the Release of the Landlord Claims pursuant to Section 2.3 and the Release of the Guarantees pursuant to Section 5.4; (C)

the relevant Landlord Entity indemnifies the Tenant and Target in respect of any Claims by any Governmental Authority arising from or relating to such amendments; and (D) such amendments shall not apply to the Leasebacks;

- (iii) the Tenant agrees to terminate, prior to Closing, any sublease, licence or other agreement granted by Tenant in favour of any Person and providing for a right of occupancy in all or any portion of the Premises or otherwise affecting the Real Property Interests. However, during the Interim Period and the Leaseback Period, the Tenant shall be permitted (A) to have existing third party pharmacists occupy a portion of the Premises and remain in such Leaseback Premises no later than March 30, 2015, all in accordance with the accommodation confirmed in the Endorsement of R.S.J. Morawetz dated February 18, 2015, which accommodation remains subject to the Landlord Entities' rights, and (B) to conduct through its Agent the Sale on the Premises and the Leaseback Premises as defined in the Sale Order in accordance with such Order and the Initial Order the whole as more fully provided for in Section 5.6; and
- (iv) The Landlord Entities and the Tenant shall enter into the Leasebacks on Closing, as provided for in Section 5.6.
- (b) Upon execution of this Agreement by the Tenant and the Landlord Entities, this Agreement shall constitute a binding agreement for the surrender of all of the Real Property Interests and the Leasebacks, subject to the issuance of the Approval Order and other conditions, as more particularly contained herein.
- (c) For greater certainty, the Real Property Interests will not include any of the Excluded Assets or the Tenant's other assets relating to the Premises.

1.3 Assignments of Certain Agreements

- (a) Nothing in this Agreement shall be construed as an attempt to assign to any Landlord Entity any contract or other agreement which, as a matter of law or by its terms, is not assignable in whole or in part without the consent of the other party or parties to such contract or other agreement, unless such consent has been given or the assignment has been approved by the Court.
- (b) No later than thirty (30) days following Closing, the Landlord Entity for the Conestoga Mall Location (#3672) shall cause the letter of credit posted as security for the Site Plan Agreement in relation to the Premises at such Location to be released to the Tenant without further drawings thereunder and the Landlord Entity shall post replacement letters of credit or other security such that the letter of credit posted by the Tenant is released to the Tenant. The Tenant shall cooperate with such Landlord Entity to provide it with documentation in the Tenant's possession and control relating to the performance by the Tenant of the work described in the Site Plan Agreement, including the certificates of the Tenant's professionals.

1.4 Post-Filing Obligations

Each and every one of the obligations, undertakings, covenants, representations and warranties of the Tenant to the Landlord Entities contained herein or in any Closing Document shall constitute "post-filing" obligations of the Tenant within the CCAA Proceedings.

ARTICLE 2 CONSIDERATION

2.1 Consideration

- (a) Subject to only the adjustments provided for in Section 3.2, the aggregate consideration (the "Consideration") payable by the Landlord Entities to the Tenant for the Transaction, exclusive of all Taxes, shall be One Hundred and Thirty-Eight Million Dollars (CDN\$138,000,000).
- (b) The Consideration shall be paid by the Landlord Entities to the Monitor, as escrow agent, on the Closing Date by wire transfer of immediately available funds payable to or to the order of Monitor, as escrow agent, as it may otherwise direct in writing in accordance with the provisions of Section 8.4.
- (c) In addition, the Landlord Entities shall:
 - (i) waive any fees or charges payable to the Landlord under any of the Leases in respect of the Transaction, including, without limitation, any fees, penalties, or charges payable to any Landlord in respect of a surrender of a Lease and any obligation in respect of the payment of accelerated and/or increased rent which arises solely as a result of a surrender of a Lease; and
 - (ii) pay all applicable Taxes payable in connection with the Transaction, including, without limitation, any Taxes payable on the surrender of any of the Leases by the Tenant to the Landlord Entities.

The provisions of this Section 2.1(c) shall not merge upon Closing but shall survive Closing.

(d) Concurrently with the execution of this Agreement, the Landlord Entities shall have paid or shall have caused to be paid to the Monitor, by wire transfer of immediately available funds payable to or to the order of the Monitor, as escrow agent, as it may otherwise direct in writing as escrow agent, a deposit in the amount of fifteen percent (15%) of the Consideration (the "Deposit"), to be held by the Monitor in accordance with Section 8.4. The Monitor shall have no obligation to earn or maximize any interest on the Deposit. The Deposit (plus accrued interest, if any) will be applied on Closing in satisfaction of an equivalent amount of the Consideration. The Deposit is refundable (plus accrued interest, if any) if the Approval Order is not obtained on or before the Outside Date or if the Closing does not occur on or before the Closing Date for any reason except only in the circumstances provided for in Section 2.1(e). In the event that the Closing has not occurred for any reason other than in the circumstances provided for in

- Section 2.1(e), the Monitor shall, without the requirement for notice, release and disburse the Deposit to a single law firm designated by the Landlord Entities two (2) Business Days following the Closing Date.
- (e) If the Closing does not occur solely by reason of a breach by any of the Landlord Entities of its representations, warranties or covenants or other default of any of the Landlord Entities under this Agreement, the Monitor shall disburse the full amount of the Deposit and all interest thereon to the Tenant (without allocation and notwithstanding Section 1.1) regardless of which Landlord Entity(ies) default(s), and the Deposit together with all interest thereon shall become the property of, and may be retained by, the Tenant as liquidated damages (and not as a penalty) to compensate the Tenant for the expenses incurred and the delay caused and opportunities foregone as a result of the failure of the Transaction to close. The entitlement of the Tenant to the Deposit in such circumstances shall be the Tenant's sole and only remedy in respect of such breach or default and, upon the Deposit being released to the Tenant, each of the Landlord Entities and the other parties hereto shall be automatically released of any and all liability pursuant to this Agreement. This Section 2.1(e) shall survive the termination of this Agreement or if the Transaction is not successfully completed for any reason.

2.2 Surrender of Leases

- (a) The Tenant and the Landlord Entities shall, effective as of 11:59 p.m. on the day immediately preceding the Closing Date, for each Lease, surrender and accept surrender and execute a Lease Amendment and Surrender Agreement for each Lease and a Leaseback for the Real Property Interests under each such Lease other than the Oakridge Location (store #3711). For greater certainty, no surrender of Lease shall take effect unless and until Closing occurs.
- (b) The Landlord Entities shall not assume any obligations or liabilities of the Tenant to third parties with respect to the Real Property Interests, whether in respect of the period on, before or after the Closing Date, or otherwise arising, incurred or accrued on or after the Closing Date whether in respect of the period on, before or after the Closing Date. The foregoing shall be further incorporated in each Lease Amendment and Surrender Agreement. This Section 2.2(b) shall not merge upon Closing but shall survive Closing.

2.3 Landlord Claims

On Closing, the Landlord Entities shall execute and deliver a release in the form attached as Schedule "N" (each a "Release of Landlord Claims") in favour of the Tenant Releasees, whereby each Landlord Entity on its own behalf and on behalf of its Affiliates that own or owned a direct or indirect interest in the real properties of which any of the Premises form a part (collectively, the "Properties" and individually, a "Property") and on behalf of each beneficial owner that now owns an interest in its respective Properties will: (a) fully and unconditionally release and forever discharge each of the Tenant Releasees of and from any and all Claims which each Landlord Entity ever had, now has or hereafter can, shall or may have against any of the Tenant Releasees in any way relating to or arising from any of the Release Matters, but excluding Claims against the Tenant in any way relating to or arising from: (i) the Tenant's

obligations or liabilities under this Agreement which are expressly stated to survive the Closing of this Agreement, and (ii) the Leasebacks or any other Closing Documents; and (b) fully and unconditionally release and forever discharge each of the Tenant Affiliate Releasees of and from any and all Claims which each Landlord Entity ever had, now has or hereafter can, shall or may have against any of the Tenant Affiliate Releasees in any way relating to or arising from any of the Properties. "Release Matters" means the Real Property Interests, the Leases and the Premises. None of the Landlord Entities shall have any Claims in the CCAA Proceedings in connection with the Release Matters or in connection with any matter relating to or arising from the properties of which any of the Premises form a part. This Section 2.3 shall not merge upon Closing but shall survive Closing.

2.4 Tenant Claims

On Closing, the Tenant and Target shall execute and deliver a release in the form attached as Schedule "O" (the "Release of Tenant Claims") in favour of the Landlord Releasees whereby the Tenant and Target will fully and unconditionally release and forever discharge each of the Landlord Releasees of and from any and all Claims which each or Tenant and Target ever had, now has or hereafter can, shall or may have against any of the Landlord Releasees, whether in respect of the period prior to or after Closing, in any way relating to or arising from: (a) any of the Release Matters; and (b) the properties of which any of the Premises form a part, but excluding Claims in any way relating to or arising from: (i) the Landlord Entities' obligations or liabilities under this Agreement which are expressly stated to survive the Closing of this Agreement; and (ii) the Leasebacks or any other Closing Documents. This Section 2.4 shall not merge upon Closing but shall survive Closing.

ARTICLE 3 ADJUSTMENTS

3.1 Statement of Adjustments

The Landlord Entities have prepared and delivered to the Tenant, with the cooperation of the Tenant, a statement of adjustments with supporting documentation on a Location by Location basis in accordance with the Leases with reasonable particulars, which statement is attached at Schedule "Q" (the "SOA"). The Landlord Entities and the Tenant shall co-operate and provide any documentation and information necessary for verifying the SOA.

3.2 Adjustments and Absence of Post-Closing Adjustments

On Closing, the Tenant and the Landlord Entities shall adjust, as of the Closing Date (such date being allocated to the Landlord Entities), only those items listed in Sections 3.3 and 3.4(b) as such adjustments are shown on the SOA. For the avoidance of doubt, there shall not be any adjustments for categories of items not shown on the SOA and the amounts set out on the SOA are final. Except as otherwise provided for in Section 3.3, the Tenant shall be responsible for all expenses relating to the Real Property Interests and the Leases for the period ending on the day before the Closing Date, and the Landlord Entities shall, subject to the Leasebacks, be responsible for all expenses relating to the properties wherein the Premises are situated for the period from and after and including the Closing Date. If the amount of any adjustments required herein cannot be reasonably determined as of the Closing Date, an estimate agreed to by the Tenant and the Landlord Entities, each acting reasonably, shall be made as of the Closing Date

and shall serve as final determination. Notwithstanding anything to the contrary herein, after Closing, there shall be no adjustments, or re-adjustments of the adjustments made at Closing, whether in respect of the period prior to or after the Closing Date (including for greater certainty for the Leaseback Period). This Section 3.2 shall not merge upon Closing but shall survive Closing.

3.3 Items of Adjustments under the Leases.

There shall be no adjustments for any periods prior to January 1, 2013 and no accelerated rent or default rent or default charges payable by or on behalf of the Tenant. The Closing Date itself shall be allocated to the Landlord Entities. Subject to the foregoing, the following items shall be adjusted at Closing, in the manner set out in each of the Leases, but without duplication:

- (a) basic rent or minimum rent;
- (b) reconciliations of CAM, operating expenses, realty taxes and other additional rent payable by the Tenant to the Landlord Entities pursuant to the Leases, including any utilities invoiced by the Landlord Entities to or paid by the Tenant to the Landlord Entities as part of additional rent for 2014 and the stub period in 2015 up to and not including the Closing Date, based on actual or estimated amounts agreed to by the Tenant and the Landlord Entities, each acting reasonably;
- (c) those adjustments set out in the SOA at Schedule "Q" which shall include the items in Sections 3.3(a) and 3.3(b); and
- (d) an adjustment in favor of the relevant Landlord Entities in the amount of CDN\$1,763,454.27 inclusive of Taxes with respect to Location 3666 Hillcrest Mall (as described on Schedule "B").

3.4 Realty Tax Appeals

- (a) The Tenant and the Landlord Entities acknowledge that with respect to the Premises the Tenant and/or the Landlords may have instituted certain appeals and/or claims in respect of realty taxes or assessments for certain periods prior to the Closing Date and possibly including the tax year in which the Closing Date occurs (all such appeals, claims and any associated reassessments are hereinafter collectively referred to as the "Appeals").
- (b) On Closing, in consideration of the additional adjustment in favour of the Tenant in the amount set out in Schedule "Q", the Tenant shall assign to the relevant Landlord Entity of each of the Locations which are subject to the Appeals all of its right, title and interest, if any, in and to any credit, refund and/or rebate which may arise from any of the Appeals (collectively, the "Tax Refunds") for any period that is prior to the Closing Date.
- (c) From and after the Closing Date, such Landlord Entities may, at their sole cost and expense but without any obligation to do so, assume or retain the carriage of the Appeals and continue as the appellant in the Appeals. The Tenant and such Landlord Entities agree to co-operate with the other with respect to the Appeals

and to provide the other with access to any reasonably necessary documents or materials required to continue any Appeals. If the Appeals may only be prosecuted in the name of the Tenant, the Tenant shall cooperate with the relevant Landlord Entity, including granting such authorizations as may be reasonably required, to enable the relevant Landlord Entity to pursue and prosecute such Appeals, at the Landlord Entity's sole cost and expense. This Section 3.4(c) shall not merge upon Closing but shall survive Closing.

3.5 Utilities

The Landlord Entities shall not assume any contracts or agreements entered into by or on behalf of the Tenant for the supply of any utilities (including electricity, gas, water, fuel, telephone service, internet services, security and surveillance services or otherwise) at the Premises. On or before the Vacancy Date, the Tenant shall terminate all its contracts and agreements for the supply of any utilities to the Leaseback Premises. For greater certainty, the Landlord Entities shall only be required to provide such utilities to the Premises during the Leaseback Periods as the Landlord is currently required to provide pursuant to the relevant Leases, and the Tenant shall remain responsible to procure any and all other utilities required for its activities in the Leaseback Premises during the Leaseback Periods, subject always to the provisions of the Leasebacks. For the avoidance of doubt, there shall be no adjustment at Closing in respect of the payment of any utilities at the Premises. The provisions of this Section 3.5 shall not merge on Closing and shall survive the Closing.

ARTICLE 4 LANDLORD ENTITIES' ACKNOWLEDGEMENTS

4.1 "As Is, Where Is"

Notwithstanding the foregoing or anything else contained herein or elsewhere, but expressly subject to the terms of the Initial Order, the Sale Order and the Approval Order and any other Order of the Court made in the CCAA Proceedings, each Landlord Entity hereby acknowledges and agrees, on behalf of itself, that:

- (a) Except as otherwise expressly set out in Section 6.1 and except for the Tenant's covenant to leave the Premises in a broom-swept condition on the Vacancy Date as set out in Section 5.6(1), the Landlord Entities are accepting a surrender of the Real Property Interests on the Closing Date and accepting the Premises on an "as is, where is" basis and subject to the Permitted Encumbrances;
- (b) There are no written or oral statements, representations, warranties, promises or guarantees of any nature or kind whatsoever, either express or implied (by operation of law or otherwise) made by the Tenant and/or the Monitor and/or their respective legal counsel, advisors or other representatives with respect to the Real Property Interests and/or the Permitted Encumbrances, including, without limiting the generality of the foregoing, any written or oral statements, representations, warranties, promises, indemnities or guarantees of any nature or kind whatsoever, either express or implied (by operation of law or otherwise) as to the condition of any of the Real Property Interests, the status of any of the Leases, the existence of any default on the part of the tenant or landlord under any of the Leases, the

condition of, in, on, under or in the vicinity of any of the Premises or the use permitted at any of the Premises, and/or the existence of any Encumbrance on any of the Real Property Interests and/or any of the Premises and/or any of the Leases;

- (c) Any disclosure in respect of any of the Real Property Interests and/or any of the Premises and/or any of the Leases and/or any Encumbrances was made available to the Landlord Entities solely as a courtesy but the Landlord Entities are not entitled to rely on such disclosure, and it is expressly acknowledged by the Landlord Entities that no statement, representation, warranty, promise or guarantee of any nature or kind whatsoever, either express or implied (by operation of law or otherwise), is made by the Tenant and/or the Monitor and/or their respective legal counsel, advisors or other representatives as to the accuracy, currency or completeness of any such disclosure, and each of them expressly disclaims any and all liabilities with respect to such disclosure and any and all errors therein or omissions therefrom;
- (d) The Landlord Entities were given the opportunity to conduct their own independent review, inspection, diligence and investigations and to form their own independent opinions and conclusions in respect of the Real Property Interests and/or any of the Premises and/or any of the Leases and the Encumbrances. The Landlord Entities' decision to enter into this Agreement and complete the Transaction was made of their own accord without reference to or reliance upon any disclosure in respect of any of the Real Property Interests, and/or any of the Premises and/or any of the Leases and/or any Encumbrances; and
- (e) The Landlord Entities have received copies of the Leases and are familiar with the terms, agreements, covenants, obligations and conditions therein.

The provisions of this Section 4.1 shall not merge on Closing and shall survive the Closing.

4.2 Monitor's Capacity

Notwithstanding the foregoing or anything else contained herein or elsewhere, each of the Tenant and the Landlord Entities acknowledges and agrees that: (a) the Monitor's obligations hereunder are limited to those specifically set out in Sections 5.6(h), 8.4, 8.5, 9.5 and 10.1 hereof; and (b) Alvarez & Marsal Canada Inc. is acting solely in its capacity as the Court appointed monitor of the Tenant in the CCAA Proceedings and not in its personal or corporate capacity, and has no liability in connection with this Agreement whatsoever, in its personal or corporate capacity or otherwise, save and except for and only to the extent of the Monitor's gross negligence or willful misconduct.

4.3 Approval Order and other Orders.

Notwithstanding anything else contained herein, or the Closing of the Transaction, the Landlord Entities shall be bound by and benefit from the Approval Order, the Initial Order, the Sale Order and any other Orders of the Court made in the CCAA Proceedings, for the duration of the shorter of the following periods: (a) the Leaseback Periods on a Leaseback Premises by Leaseback Premises basis; or (b) the period during which any other owners, operators, managers or

landlords of commercial shopping centres or other commercial properties in which there is a store, office or warehouse owned or operated by Target, is bound by or obtains any benefit from same. Without limiting the generality of the foregoing, during such period, the Landlord Entities shall benefit from the stay of proceedings provision provided for at paragraph 18 of the Initial Order.

ARTICLE 5 LANDLORD ENTITIES AND TENANT COVENANTS

5.1 Assignment until Vacancy Date.

Each Landlord Entity covenants not to assign its rights, obligations and interests in any Leaseback, except if such assignment is to an Affiliate of such Landlord Entity, in which case the Landlord Entity shall cause any such Affiliate assignee to be bound by the covenants, obligations and liabilities of such Landlord Entity under the Leaseback, and to comply with the terms of the Leaseback for the relevant Leaseback Premises until the Vacancy Date. However, the foregoing shall not prohibit a Landlord Entity from assigning its interest in a Leaseback and or the Leaseback Premises as security to a *bona fide* lender which is taking security. The provisions of this Section 5.1 shall not merge on Closing and shall survive the Closing.

5.2 Interim Period

- (a) During the Interim Period, the Landlord Entities and the Tenant shall comply with each and every term and condition of the Leases as currently applicable in the CCAA Proceedings, subject only to the provisions of the Initial Order and the Sale Order and the provisions of this Section 5.2 and Section 5.5.
- (b) During the Interim Period, the Tenant by itself or through its Agent under the Agency Agreement shall be entitled to sell or remove, or permit any other parties to remove, any and all chattels, personal property, Inventory, FF&E and any other Excluded Assets or item of the type listed in Schedule "J1", but excluding any Landlord Assets and any items in Schedule "J2", from all or any of the Premises in accordance with the Initial Order and the Sale Order. The Tenant's obligations to repair or re-instate the relevant Premises shall be as required by the terms and conditions of the Initial Order and the Sale Order.
- (c) The Landlord Entities' rights and remedies in respect of the Leases shall be subject to the provisions of the Initial Order and the Sale Order.

5.3 Absence of Tenant Estoppels.

Notwithstanding anything to the contrary in any of the Leases or any of the Leasebacks, the Tenant shall not provide any acknowledgement, status certificate or estoppel in respect of any of the Leasebacks. The provisions of this Section 5.3 shall not merge on Closing and shall survive the Closing.

5.4 Guarantee

On Closing, each Landlord Entity, on its own behalf and on behalf of its Affiliates that own or

owned a direct or indirect interest in any of the Properties and on behalf of each beneficial owner which now owns an interest in its respective Properties, and all of their respective successors and assigns, which holds a guarantee and/or indemnity from any of the Tenant Affiliate Releasees in connection with any Lease including, without limitation, the Guarantees, shall execute and deliver full and final releases in favour of the Tenant Affiliate Releasees in respect of all covenants, obligations, liabilities, Claims, indemnities and guarantees of the Tenant Affiliate Releasees in any way relating to or arising from or under the Guarantees, the Release Matters and/or the properties of which any of the Premises form a part. Such releases shall be in the form attached hereto as Schedule "P" (each a "Release of Guarantees"). The provisions of this Section 5.4 shall not merge on Closing and shall survive the Closing.

5.5 Wind-Down or Non-Operation

The Landlord Entities acknowledge and agree that, during the Interim Period and the Leaseback Periods, notwithstanding anything to the contrary in any of the Leases or any of the Leasebacks, the Tenant shall have no obligation to stock, open, staff, operate or continuously operate any business at the Premises, and may, in its sole discretion, by itself or through its Agent continue the Sale as defined in and as approved or permitted by the Initial Order and the Sale Order, provided that the Tenant shall continue to maintain its existing insurance coverage (including any self-insurance, if currently existing) on the Premises until the Vacancy Date. As part of such winding-down, Tenant shall not sell or remove or permit the sale or removal from the Premises of any of the items listed in Schedule "J2". The provisions of this Section 5.5 shall not merge on Closing and shall survive the Closing.

5.6 Leasebacks

Each Landlord Entity covenants, on Closing to lease, sublease, or sub-sublease back, as the case may be, to the Tenant, and the Tenant covenants to lease, sublease, or sub-sublease (collectively, the "Leasebacks" and each a "Leaseback") each and every one of the Premises excluding the Premises at the Oakridge Location (store #3711) (each a "Leaseback Premises") for the Leaseback Period on the same terms and conditions as in the applicable Leases as currently applicable under the CCAA Proceedings for the Premises, including under all Court Orders made in the CCAA Proceedings to date, including the Initial Order and the Sale Order, each of which Leaseback shall be substantially in the form attached hereto as Schedule "K", provided that during the Leaseback Period of any Leaseback:

- (a) the Tenant shall accept the Leaseback Premises on an "as is, where is" basis;
- (b) Sections 5.1, 5.3 and 5.5 hereof shall apply without any of the amendments in Section 1.2(a)(ii) and without any of the Guarantees;
- during the Leaseback Period, the Tenant may occupy the Leaseback Premises for the purpose of selling or removing, either directly or through its agents, any and all chattels, personal property, Inventory, FF&E and any other Excluded Assets or item listed on Schedule "J1", but excluding any items listed in Schedule "J2", from the Leaseback Premises in accordance with the Initial Order and the Sale Order and the provisions of this Section 5.6;

- (d) the Tenant shall have no less rights, and subject to this Section 5.6, no less obligations, under the Leasebacks to use and occupy each of the Leaseback Premises than in the Leases as currently applicable under the CCAA Proceedings for the Premises, and notwithstanding anything to the contrary, no Landlord Entity which is a landlord/sub-landlord/sub-sublandlord pursuant to any of the Leasebacks shall, subject to the Initial Order, be entitled to terminate any of the Leasebacks prior to the Vacancy Date by any reason other than a default or event of default following the Closing Date beyond the applicable cure period giving rise to a right of termination pursuant to the terms and conditions of the Leaseback;
- (e) the Landlord Entities shall have no obligation under the Leasebacks to pay any allowance or inducement to Tenant, to carry out any work or improvements in the Premises (except for any maintenance, repair and replacement which is the responsibility of the landlord/sub-landlord/sub-sublandlord under the Leasebacks) or to pay any commissions;
- (f) the Tenant shall not require any consent from the landlord/sub-landlord/sub-sublandlord of the Leasebacks to allow pharmacists to operate in the Leaseback Premises until no later than March 30, 2015; subject to the foregoing, the Tenant shall not have any rights to assign its rights, title, interest or obligations pursuant to the Leasebacks nor to grant any sublease or licence in respect of the Leaseback Premises unless permitted by the Initial Order, the Sale Order, the Approval Order or any other Order of the Court made on notice to the Landlord Entities;
- (g) Section 5.5 shall apply to the Leasebacks during their respective Leaseback Periods and the Tenant shall deliver to the Landlord Entity certificates of insurance attesting the existence of the insurance coverage required pursuant to the terms and conditions of the Leasebacks or shall confirm that the existing insurance coverage pursuant to the Leases remains in full force and effect;
- (h) the Tenant shall pre-pay to the Monitor, for each Leaseback Premises and for the entire Leaseback Period, a gross rent on account of all basic rent, minimum and additional rent, without adjustment, plus applicable GST/HST, QST or BCPST, corresponding to the amounts set out on Schedule "G" in respect of each Leaseback Premises for the period from the Closing Date to June 30, 2015 (collectively, for all Leaseback Premises, the "Pre-Paid Leaseback Rent"). The Pre-Paid Leaseback Rent shall be paid out of the Consideration payable hereunder and shall be directed to and be held by the Monitor, as escrow agent, in accordance with Section 8.4. The Monitor shall make regular gross rent payments to the landlord/sub-landlord/sub-sublandlord of the Leasebacks payable under the Leasebacks from the Pre-Paid Leaseback Rent as per the Initial Order up to the Vacancy Date. Any part of such Pre-Paid Leaseback Rent not paid or not required to be paid under any Leaseback shall be refunded by the Monitor to the Tenant on the Business Day following the Vacancy Date on a Leaseback Premises by Leaseback Premises basis;
- (i) notwithstanding the duration of the Leaseback Period, the Tenant may, on an individual Leaseback Premises basis, without penalty or being in default, provide

notice (an "Early Vacancy Notice") to the respective landlord/sub-landlord/sub-sublandlord of the Leaseback Premises, on or before the date indicated below to advance the Vacancy Date to the date indicated below in respect of each Early Vacancy Notice date:

Early Vacancy Notice to be given on or before:	For a Vacancy Date effective on:	
April 15, 2015	April 30, 2015	
April 30, 2015	May 15, 2015	
May 15, 2015	May 30, 2015	
May 30, 2015	June 15, 2015	

For the avoidance of doubt, the Tenant may give an Early Vacancy Notice in respect of each Leaseback Premises independently. If the Tenant has not given an Early Vacancy Notice in respect of any Leaseback Premises on or before May 30, 2015, then the Leaseback Period for such Leaseback Premises shall end on June 30, 2015;

- on the Vacancy Date of each Leaseback Premises (as such Vacancy Date may be advanced pursuant to an Early Vacancy Notice), the Tenant shall vacate the entire Leaseback Premises, said Leaseback shall be at an end and all of the obligations of the Tenant, including to maintain insurance or pay rent for such Leaseback Premises, shall terminate;
- on or before the earlier of the expiry of the Leaseback Period or the Vacancy Date of any Leaseback, the Tenant shall vacate the relevant Leaseback Premises, remove any chattels, personal property, Inventory, FF&E and any other Excluded Assets in accordance with the Initial Order and the Sale Order; for greater certainty, the Tenant or the Agent shall remove each and every item of the type listed in Schedule "J1" and shall not remove any of the items of the type listed in Schedule "J2". If the Tenant fails to remove any item of the type listed in Schedule "J1", then the Landlord Entity which is the landlord/sub-landlord/sub-sublandlord of that Leaseback may, at their election, either exercise any and all remedies to cause Tenant to remove same, or obtain possession and ownership of same without notice or compensation to Tenant;
- (l) on the earlier of the expiry of the Leaseback Period or the Vacancy Date of any Leaseback, the Tenant shall leave each Leaseback Premises in a broom-swept condition as required by the provisions of the Initial Order and the Sale Order and the Landlord Entity which is the landlord/sub-landlord/sub-sublandlord of that Leaseback shall accept the Leaseback Premises in such condition, and notwithstanding anything to the contrary in the Leaseback, the Tenant shall not be responsible for making any repairs, replacements, renovations, alterations,

improvements or upgrades in or to any of the Leaseback Premises except as provided for in the Initial Order or the Sale Order and other than to leave the Leaseback Premises in a broom-swept condition and in the condition required by the provisions of the Initial Order and the Sale Order; and

(m) within five (5) Business Days of a Vacancy Date, the Landlord Entity which is the landlord/sub-landlord/sub-sublandlord of a Leaseback shall execute and deliver to the Tenant a full and final release in respect of any Claims relating to or arising from the Leasebacks in a form substantially similar to the Release of Landlord Claims attached as Schedule "N" and the Tenant shall execute and deliver to the Landlord Entity which is the landlord/sub-landlord/sub-sublandlord of such Leaseback, a release in respect of any Claims relating to or arising from the Leasebacks in a form substantially similar to the Release of Tenant Claims attached as Schedule "O", which releases may be subject to any Claims identified therein relating to or arising from the Leasebacks notice of which has been delivered to the other party prior to the delivery of such releases.

The provisions of this Section 5.6 shall not merge on Closing and shall survive the Closing.

ARTICLE 6 REPRESENTATIONS, WARRANTIES & COVENANTS

6.1 Tenant's Residency

The Tenant represents and warrants each Landlord Entity that and acknowledges and confirms that the Landlord Entities are relying on such representations and warranties in connection with the entering into of this Agreement that:

- (a) the Tenant is not a non-resident of Canada within the meaning of the *Income Tax Act* (Canada);
- (b) Tenant is duly registered under the ETA with respect to the GST/HST and under the ARQST with respect to the QST and under the BCPSTA with respect to the BC PST and its tax numbers are listed as follows:
 - (i) GST/HST: 835536608RT0001;
 - (ii) QST: 1217234367TQ0001; and
 - (iii) BC PST: PST-1013-9272.

The provisions of this Section 6.1 shall not merge on Closing and shall survive the Closing.

6.2 Landlord Entities' Representations and Warranties

Subject to Section 1.1, each Landlord Entity represents and warrants on behalf of itself only as to all of the following matters, and acknowledges and confirms that the Tenant is relying on such representations and warranties in connection with the entering into of this Agreement:

- (a) Each Landlord Entity has been duly incorporated or constituted and is validly subsisting under the Laws of the jurisdiction of its incorporation or constitution, and has all requisite corporate capacity, power and authority to carry on its business as now conducted by it and to own its properties and assets and is qualified to carry on business under the Laws of the jurisdictions where it carries on a material portion of its business.
- (b) The execution, delivery and performance by the Landlord Entities of this Agreement:
 - (i) has been duly authorized by all necessary corporate action on the part of each Landlord Entity; and
 - (ii) does not (or would not with the giving of notice, the lapse of time or the happening of any other event or condition) require any consent or approval under, result in a breach or a violation of, or conflict with, any of the terms or provisions of each of their respective constating documents or by-laws.
- (c) This Agreement has been duly executed and delivered by the Landlord Entities and constitutes legal, valid and binding obligations of each of the Landlord Entities, enforceable against each of them in accordance with their respective terms subject only to any limitation under applicable Laws relating to (i) bankruptcy, winding-up, insolvency, arrangement and other similar Laws of general application affecting the enforcement of creditors' rights, and (ii) the discretion that a court may exercise in the granting of equitable remedies such as specific performance and injunction.
- (d) The Landlord Entities have and will have at Closing, available all funds on hand necessary to pay the Consideration.
- (e) The Landlord Entities constitute, or the Landlord Entities have the full power, right and capacity to bind, 100% of all entities constituting the Landlords pursuant to each Lease and each Guarantee and each of the beneficial owners referred to in Section 6.4(d). Except for the beneficial owners referred to in Section 6.4(d), no Person other than a Landlord Entity has a beneficial, registered, legal or other interest as fee holder/owner, ground lessee (or emphyteutic lessee) or head tenant, as the case may be, in and to the Leases or Guarantees.

The provisions of this Section 6.2 shall not merge on Closing and shall survive the Closing.

6.3 Landlord Entities and Tenant Covenants

(a) The Landlord Entities shall take all commercially reasonable actions as are within their respective power to control, and shall use commercially reasonable efforts to cause other actions to be taken which are not within their respective power to control, so as to ensure compliance with each of the conditions and covenants set forth in Article 5 and Article 6 and to fulfill the conditions set forth in Article 7.

- (b) The Tenant shall take all commercially reasonable actions as are within its power to control, and shall use commercially reasonable efforts to cause other actions to be taken which are not within its power to control, so as to ensure compliance with each of the conditions and covenants set forth in Article 5 and Article 6 and to fulfill the conditions set forth in Article 7.
- (c) The Landlord Entities will promptly notify the Tenant and the Tenant will promptly notify the Landlord Entities upon:
 - (i) instituting a suit or proceeding, or becoming aware of any Order or any motion requesting an Order, in the CCAA Proceedings or otherwise restraining or enjoining the consummation of the Transaction contemplated under this Agreement; or
 - (ii) receiving any notice from any Governmental Authority of its intention to nullify or render ineffective this Agreement or such Transaction if consummated.

6.4 Tax Matters

In addition to the representations and warranties set forth in Section 6.2, each Landlord Entity, on behalf of itself, further warrants, represents and covenants to the Tenant, and acknowledges and confirms that the Tenant is relying on such representations and warranties and covenants in connection with the entering into of this Agreement, that:

- (a) each Landlord Entity or beneficial owner referred to in Section 6.4(d), as the case may be, is duly registered under Subdivision (d) of Division V of Part IX of the ETA with respect to the GST/HST and if applicable, under Division I of Chapter VIII of Title I of the ARQST with respect to the QST, each of which registrations are and shall remain in full force and effect and shall not have been cancelled or revoked on the Closing Date. The relevant registration numbers shall be provided in the Tax Certificate, Undertaking and Indemnity;
- (b) in accordance with subsection 221(2) of the ETA and section 423 of the ARQST, each Landlord Entity or beneficial owner referred to in Section 6.4(d), as the case may be, shall self-assess and remit directly to the appropriate Governmental Authority any GST/HST imposed under the ETA and any QST imposed under the ARQST payable in connection with the Transaction, including the surrender of any of the Leases, and each Landlord Entity or beneficial owner referred to in Section 6.4(d), as the case may be, shall have executed and delivered a certificate, undertaking and indemnity which includes a certification of all appropriate registration numbers issued under the ETA and/or the ARQST and incorporates the provisions of this Section 6.4 (the "Tax Certificate, Undertaking and Indemnity");
- (c) notwithstanding the foregoing, if the Tenant is required to collect any Taxes including GST/HST imposed under the ETA, QST imposed under the ARQST, BC PST imposed under the BCPSTA, and any similar value added or multistaged tax or sales tax exigible on the Transaction on Closing, each Landlord

Entity will pay, in addition to the Consideration, and the Tenant will collect, such Taxes on the Closing Date;

- (d) each Landlord Entity, as the case may be, shall warranty and represent, in its Tax Certificate, Undertaking and Indemnity that it is taking a surrender of the Real Property Interests on the Closing Date, as principal for its own account and not as an agent, trustee or otherwise on behalf of another person, or if it is acting as an agent, trustee or otherwise on behalf of another person, it shall cause the beneficial owner of the fee interest to the real property underlying the Real Property Interests so surrendered to execute and deliver a Tax Certificate, Undertaking and Indemnity to the Tenant on Closing;
- (e) each Landlord Entity or beneficial owner referred to in Section 6.4(d), as the case may be, shall make and file all required return(s) in accordance with the requirements of subsection 228(4) of the ETA and Section 438 of the ARQST; and
- (f) except to the extent of the Tenant's failure to remit the Taxes collected on Closing, if any, to the applicable Governmental Authority, each Landlord Entity shall indemnify and save the Tenant harmless from and against any and all Taxes including GST/HST under the ETA, QST under the ARQST, BC PST under the BCPSTA and any similar value added or multi-staged tax or sales tax, penalties, costs and/or interest which may become payable by or assessed against the Tenant in connection with the Transaction, or as a result of any inaccuracy, misstatement or misrepresentation made by or breach of covenant of any Landlord Entity on the Closing Date in connection with any matter raised in this Section 6.4 or contained in its Tax Certificate, Undertaking and Indemnity. This indemnity shall not merge on but shall survive Closing.

The provisions of this Section 6.4 shall not merge on Closing and shall survive the Closing.

6.5 Survival of Covenants, Representations and Warranties.

The covenants, representations and warranties of the Tenant and the Landlord Entities set forth in this Agreement which, by their terms, are to be performed before or on Closing, will merge on Closing except for those which are expressly provided in this Agreement to survive Closing. The other covenants, representations and warranties of the Tenant and the Landlord Entities set forth in this Agreement shall survive Closing.

ARTICLE 7 CONDITIONS TO CLOSING

7.1 Conditions of Closing for the Benefit of the Landlord Entities

The Landlord Entities' obligation to complete the Transaction is subject to the following conditions to be fulfilled or performed, on or before the Closing Date, which conditions are for the exclusive benefit of the Landlord Entities and may be waived, in whole or in part, by the Landlord Entities in their sole discretion:

- (a) the representations and warranties of the Tenant in Section 6.1 shall be true and correct as of the Closing Date in all material respects, as if such representations and warranties were made on and as of such date; and
- (b) the Tenant shall have performed and complied with all of the material terms and conditions in this Agreement on its part to be performed or complied with at or before Closing and shall have executed and delivered or caused to have been executed and delivered to the Landlord Entities at Closing all the Closing Documents of the Tenant.

7.2 Conditions of Closing for the Benefit of the Tenant

The Tenant's obligation to complete the Transaction is subject to the following conditions to be fulfilled or performed, on or before the Closing Date, which conditions are for the exclusive benefit of the Tenant and may be waived, in whole or in part, by the Tenant in its sole discretion:

- (a) the representations and warranties of any Landlord Entity in Sections 6.2 and 6.4 be true and correct as of the Closing Date with the same force and effect as if such representations and warranties were made on and as of such date; and
- (b) the Consideration plus all Taxes thereon required to be collected by the Tenant shall have been paid in its entirety (but subject to adjustments permitted hereunder) to the Monitor, as escrow agent, in accordance with the escrow closing provisions of Section 8.4, and the Landlord Entities shall have performed and complied with all of the material terms and conditions in this Agreement on its part to be performed or complied with at or before Closing and shall have executed and delivered or caused to have been executed and delivered to the Tenant at Closing all the Closing Documents of the Landlord Entities.

7.3 Conditions of Closing for the Mutual Benefit of the Parties

The obligations of either the Tenant or the Landlord Entities to complete the Transaction are also subject to the following condition to be fulfilled or performed, on or before the Closing Date, which condition is for the mutual benefit of each of the parties and may only be waived, in whole or in part, by agreement of the parties to this Agreement:

(a) the Approval Order shall have been obtained, and no Order shall have been issued which restricts or prohibits the completion of the Transaction.

ARTICLE 8 CLOSING

8.1 Closing Date

The completion of the Transaction (the "Closing") shall take place at 10:00 a.m. (Toronto time) at the Toronto office of Osler, Hoskin and Harcourt LLP, on March 6th, 2015, or on such other date and at such other time as may be agreed upon in writing by the parties (the "Closing Date"). The Closing shall take place in accordance with the provisions of this Article 8 and shall be deemed to be effective as of the date and time set out on the Monitor's Certificate.

8.2 Closing Documents

On or before Closing, subject to the provisions of this Agreement, the Landlord Entities shall and the Tenant shall, execute or cause to be executed and shall deliver or cause to be delivered into escrow (in a sufficient number of original copies or counterparts for the Tenant and each Landlord Entity and, where applicable, in registerable form), as provided for in Section 8.4, the following, which shall be in form and substance reasonably satisfactory to the Tenant and the Landlord Entities and their respective solicitors:

- (a) By the Landlord Entities and the Tenant:
 - (i) The amendment to the Leases and the indemnity relating thereto in accordance with Section 1.2(a)(ii), unless they are incorporated in the Lease Amendment and Surrender Agreements;
 - (ii) The Lease Amendment and Surrender Agreements in respect of each of the Leases:
 - (iii) The Leasebacks in accordance with Section 5.6; and
 - (iv) Such other documents as each party or each party's solicitors shall reasonably require in good faith or as may be required under applicable Laws.

(b) By the Tenant:

- (i) All master keys and duplicate keys relating to the Premises, if any, all security cards and access cards relating to the Premises, if any, and all combinations and passwords to vaults and combination locks and other security features located in the Premises, if any, in the possession of the Tenant, provided that such items shall not be provided to the Landlord Entities until the Vacancy Date for such Premises and subject to the Tenant's rights of removal;
- (ii) The Release of Tenant Claims required pursuant to Section 2.4;
- (iii) If requested by Landlord Entities, duly executed discharges, in registrable form, of each and every notice of Lease or similar notice, memorandum or other deed or document of record or registered against title to any of the properties on which the Premises are located, which shall be prepared by the solicitors of the Landlord Entities at the sole cost and expense of the Landlord Entities and approved by the Tenant's solicitors acting reasonably;
- (iv) The Approval Order; and
- (v) Such other documents as the Landlord Entities or the Landlord Entities' solicitors shall reasonably require in good faith or as may be required under applicable Laws.

- (c) By the Landlord Entities:
 - (i) The amount of the Consideration plus all Taxes thereon required to be collected by the Tenant, plus or minus the net amount of adjustments contemplated herein and minus the amount of the Deposit and interest accrued thereon, if any (collectively, the "Net Consideration");
 - (ii) The SOA evidencing the adjustments made at Closing;
 - (iii) a Release of Landlord Claims from each of the Landlord Entities required pursuant Section 2.3;
 - (iv) a Release of Guarantees from each relevant Landlord Entity in respect of any guarantees and indemnities including the Guarantees in accordance with Section 5.4;
 - (v) each Tax Certificate, Undertaking and Indemnity as required pursuant to Section 6.4;
 - (vi) a beneficial owner direction given by each beneficial owner referred to in Sections 2.3, 5.4 and 6.4(d) to the relevant Landlord Entity(ies) directing such Landlord Entity(ies) to execute and deliver the relevant Closing Documents on behalf of such beneficial owner, which direction shall state that the relevant Landlord Entity(ies) can bind the beneficial owner and that the Tenant Affiliate Releasees shall be entitled to rely on same as confirmation that the beneficial owner has indeed given said direction and is bound by the relevant Closing Documents; and
 - (vii) such other documents as the Tenant or the Tenant's solicitors shall reasonably require in good faith or as may be required under applicable Laws.

8.3 Confirmation of Satisfaction of Conditions

On the Closing Date, subject to satisfaction or waiver by the relevant party or parties, as applicable, of the conditions of Closing in its favour contained in Article 7, the parties shall confirm to the Monitor the satisfaction of all conditions to Closing, whereupon the Monitor shall deliver copies thereof to the Landlord Entities, and thereafter file the Monitor's Certificate with the Court.

8.4 Escrow Closing

(a) Subject always to Section 4.2 hereof, the Deposit and the Net Consideration shall be held by the Monitor, as escrow agent, in trust in a separate interest bearing account, pending completion of the Transaction or earlier termination of this Agreement. In holding and dealing with the funds paid to the Monitor in trust and any interest earned thereon pursuant to this Agreement, the Monitor is not bound in any way by any agreement other than this Section 8.4 and its obligation to pay and/or refund the Pre-Paid Leaseback Rent in accordance with Section 5.6(h), and

the Monitor shall not assume or be deemed to assume any duty, liability or responsibility other than to hold the trust funds and any interest earned thereon in accordance with the provisions of this Section 8.4 and to pay the funds, and any interest earned thereon, to the party becoming entitled thereto in accordance with the terms of this Agreement, except in the event of a dispute between the parties as to entitlement to the trust funds, of which the Monitor has been given notice in writing, the Monitor may, in its sole, subjective and unreviewable discretion, or shall, if requested by either of the parties, pay the trust funds and any and all interest earned thereon into court, whereupon the Monitor shall have no further obligations relating to the trust funds or any interest earned thereon or otherwise hereunder.

- (b) The Monitor shall not, under any circumstances, be required to verify or determine the validity of any written notice or other document whatsoever delivered to the Monitor in connection with the trust funds and the Monitor is hereby relieved of any liability or responsibility for any loss or damage which may arise as a result of the acceptance by the Monitor of any such written notice or other document in good faith, provided that the Monitor shall not be relieved of any liability or responsibility for any loss or damage which may arise if the Monitor releases the trust funds or any interest thereon to a party after having received prior written notice from the other party hereto claiming entitlement to such trust funds or a dispute to such entitlement.
- (c) The Monitor shall be entitled to rely upon any written instructions received from the Tenant in respect of the investment of the trust funds, provided any and all such investments shall be in guaranteed investment certificates or segregated accounts issued by or held at Schedule 1 Canadian chartered bank(s).
- (d) The parties acknowledge that the Monitor may rely upon the provisions of Section 4.2 hereof and this Section 8.4 notwithstanding that the Monitor is a party to this Agreement solely for the purpose of complying with this Section 8.4 and its obligation to pay and/or refund the Pre-Paid Leaseback Rent in accordance with Section 5.6(h) and the terms and conditions of Section 8.5, 9.5 and 10.1.
- On or before Closing, the parties' respective solicitors shall exchange the Closing Documents in escrow and the Net Consideration shall be delivered to or paid to the order of the Monitor, as escrow agent, in trust, and the Deposit and the Net Consideration shall remain in escrow with the Monitor until the Monitor has delivered the Monitor's Certificate to the Landlord Entities, upon the occurrence of which the escrow shall be lifted, the Closing Documents shall take effect as of the date and time set out in the Monitor's Certificate, the entire amount of the Deposit and the Net Consideration net only of the Pre-Paid Leaseback Rent shall be forthwith released to the Tenant and the Closing shall be deemed to have occurred as of such date and time and two (2) sets of fully signed Closing Documents shall be released to each of the Tenant and the Landlord Entities.
- (f) This Section 8.4 shall survive the Closing or termination of this Agreement.

8.5 Monitor's Certificate

The parties hereby acknowledge and agree that the Monitor shall be entitled to file the Monitor's Certificate with the Court without independent investigation upon receiving written confirmation from the Tenant and the Landlord Entities that all conditions of Closing have been satisfied or waived, and the Monitor shall have no liability to the Tenant or the Landlord Entities or any other person as a result of filing the Monitor's Certificate. The Monitor shall file the Monitor Certificate upon the Approval Order having been obtained and the Landlord Entities and the Tenant confirming to the Monitor that the conditions to Closing have been satisfied or waived.

8.6 Data Room

Immediately upon Closing, all documents and information specifically relating to the Leases, the Premises and the Real Property Interests shall be removed from the electronic data room set up for the purposes of the Real Property Portfolio Sales Process undertaken by the Tenant in the CCAA Proceedings.

ARTICLE 9 GENERAL PROVISIONS

9.1 Filings and Authorizations

- (a) Each of the Tenant and the Landlord Entities, as promptly as practicable after the execution of this Agreement, will make, or cause to be made, all such filings and submissions under all Laws applicable to it, as may be required for it to consummate the Transaction in accordance with the terms of this Agreement. The Tenant and the Landlord Entities shall coordinate and cooperate with one another in exchanging such information and supplying such assistance as may be reasonably requested by each in connection with the foregoing including, without limitation, providing each other with all notices and information supplied to or filed with any Governmental Authority (except for notices and information which the Tenant or the Landlord Entities, in each case acting reasonably, considers highly confidential and sensitive which may be filed on a confidential basis), and all notices and correspondence received from any Governmental Authority.
- (b) The Landlord Entities waive compliance with the *Bulk Sales Act* (Ontario) and any other similar bulk sales laws.
- (c) The provisions of this Section 9.1 shall not merge on Closing and shall survive the Closing.

9.2 Court Matters

- (a) The Tenant shall consult and co-ordinate with the Landlord Entities and their respective legal advisors regarding the parties upon whom the motion seeking the issuance of the Approval Order by the Court will be served. In this regard, the Tenant shall provide the Landlord Entities with draft motion materials (including for purpose of clarity any affidavit in support thereof) prior to serving same.
- (b) The Landlord Entities shall provide such information and take such actions as may be reasonably requested by the Tenant to assist the Tenant in obtaining the

Approval Order and any other Order of the Court necessary to consummate the Transaction.

9.3 Termination

This Agreement may, by notice in writing given at or prior to Closing, be terminated:

- (a) by mutual consent of the Tenant and the Landlord Entities;
- (b) by the Landlord Entities if any of the conditions in Section 7.1 have not been satisfied on or before the time specified for the satisfaction of such condition and the Landlord Entities has not waived such condition;
- (c) by the Tenant if any of the conditions in Section 7.2 have not been satisfied on or before the time specified for the satisfaction of such condition and the Tenant has not waived such condition; or
- (d) by either party if any of the conditions in Section 7.3 have not been satisfied on or before the time specified for the satisfaction of such condition and the parties have not waived such condition.

9.4 Automatic Termination - Outside Date

This Agreement shall automatically terminate at 11.59 p.m. on March 5, 2015 (the "Outside Date") if the Approval Order has not been obtained on or prior to such Outside Date, and the Deposit shall be released in accordance with Section 2.1.

9.5 No Shop

From the date of execution of this Agreement until the earlier of the Closing or termination of this Agreement pursuant to Section 9.3, neither the Tenant, Target, the Monitor or Lazard Frères & Co. ("Lazard") or Northwest Atlantic (Canada) Inc. ("Northwest") shall solicit or approach another party or enter into or participate in any negotiations or discussions with another party, or invite another party to submit any offer, regarding any transaction for all or some of the Real Property Interests or in respect of the assignment of any of the Leases or Premises without the prior written consent of the Landlord Entities; provided however that nothing in this Section 9.5 shall otherwise prohibit the Tenant, Target, the Monitor or Lazard from undertaking the Real Property Portfolio Sales Process as may be modified, supplemented or amended by the Court from time to time, including the distribution of process letters and other instructions from potential bidders and provision of access to the data room (subject to Section 8.6).

ARTICLE 10 OTHER PROVISIONS

10.1 Confidentiality

(a) From the date of execution of this Agreement until the service of motion materials to obtain the Approval Order, the Tenant, Target, the Monitor, Lazard, Northwest and other Representatives of the Tenant in a role similar to Lazard and Northwest

- and the Landlord Entities and their Representatives shall keep in confidence this Agreement, its existence and its terms and conditions.
- (b) Subject to Section 10.1(e), the Tenant, Target, the Monitor, Lazard, Northwest and other Representatives of the Tenant in a role similar to Lazard and Northwest and the Landlord Entities and their Representatives shall keep in confidence any confidential information exchanged between the parties or their respective representatives or advisors (collectively, "Confidential Information") and shall not use any such information except for the purpose of consummating the Transaction in accordance with the provisions of this Agreement and obtaining the Approval Order.
- (c) Nothing herein contained shall restrict or prohibit the Tenant, Target and the Landlord Entities from disclosing the Confidential Information (i) to such party's Representatives (x) who need to know the Confidential Information for the purposes of evaluating or completing the Transaction, (y) who are informed by the parties of this Agreement and of the confidential nature of the Confidential Information and (z) who are directed by the parties to treat the Confidential Information in a manner consistent with the terms of this Agreement; or (ii) in Court for the purpose of obtaining the Approval Order, such disclosure being subject to the provisions of Section 10.1(e). The parties agree to be responsible for any breach of the provisions of this Agreement by any of their respective Representatives. The Tenant shall use commercially reasonable efforts to cause Lazard Frères to comply with the provisions of this Section 10.1
- (d) In the event that the parties or their Representatives should be required, by law or regulation or by legal process, including the CCAA Proceedings, to disclose any Confidential Information, it is agreed that the relevant party shall provide the other parties with prompt notice of any such request, so that the other parties may seek an appropriate protective order or other remedy. The parties shall cooperate to obtain any such order or remedy.
- (e) The Tenant and the Monitor shall use their commercially reasonable efforts to obtain a sealing order with respect to any disclosure of the quantum of the Consideration payable under this Agreement and of Schedules "Q" and "G" made in Court or pursuant to the CCAA Proceedings, which sealing order shall be effective up to Closing. Only a redacted copy of this Agreement (to redact the quantum of the Consideration and Schedules "Q" and "G") shall be filed or served as part of the CCAA Proceedings and the parties acknowledge an unredacted copy of this Agreement shall be made public following Closing.
- (f) Other than statements made in the Court (or in pleadings filed therein), the Tenant, Target, the Monitor and the Landlord Entities shall not issue (prior to or after Closing) any press release or make any public statement or public communication with respect to this Agreement or the Transaction without the prior written consent of the other party, which shall not be unreasonably withheld or delayed, provided, however, that a party may, without the prior consent of the other party, issue such press release or make such public statement as may, upon

the advice of counsel, be required by applicable Laws or by any Governmental Authority with competent jurisdiction.

- (g) Nothing herein contained shall restrict or prohibit the Monitor from disclosing the Confidential Information if required to do so by a Court Order.
- (h) The provisions of this Section 10.1 shall not merge on Closing and shall survive the Closing.

10.2 Time of the Essence

Time shall be of the essence of this Agreement.

10.3 Enurement

This Agreement shall become effective when executed by the Tenant and the Landlord Entities and after that time shall be binding upon and enure to the benefit of the parties and their respective successors and permitted assigns.

10.4 Assignability

Except for assignments by any Landlord Entity to its Affiliate, in each case where such Landlord Affiliate has executed and delivered a written covenant in favour of the Tenant to assume and be jointly and severally responsible with the applicable Landlord Entity for any and all covenants, obligations and liabilities of the Landlord Entities under this Agreement in respect of such interest assigned, neither this Agreement nor any of the rights or obligations under this Agreement shall be assignable or transferable by either party without the consent of the other party.

10.5 Entire Agreement

This Agreement constitute the entire agreement between the parties with respect to the Transaction contemplated in this Agreement and supersedes all prior agreements, understandings, negotiations and discussions, whether oral or written, of the parties with respect to the subject matter of this Agreement. There are no representations, warranties, covenants, conditions or other agreements, express or implied, collateral, statutory or otherwise, between the parties in connection with the subject matter of this Agreement, except as specifically set forth in this Agreement. The parties have not relied and are not relying on any other information, discussion or understanding in entering into and completing the Transaction contemplated by this Agreement.

10.6 Waiver

- (a) No waiver of any of the provisions of this Agreement shall be deemed to constitute a waiver of any other provision (whether or not similar), nor shall such waiver be binding unless executed in writing by the party to be bound by the waiver.
- (b) No failure on the part of the Tenant or the Landlord Entities to exercise, and no delay in exercising any right under this Agreement shall operate as a waiver of

such right; nor shall any single or partial exercise of any such right preclude any other or further exercise of such right or the exercise of any other right.

10.7 Further Assurances

Each of the parties covenants and agrees to do such things, to attend such meetings and to execute such further conveyances, transfers, documents and assurances as may be deemed necessary or advisable from time to time in order to effect the Transaction and carry out the terms and conditions of this Agreement in accordance with their true intent. The provisions of this Section 10.7 shall not merge on Closing and shall survive the Closing.

10.8 Severability

If any provision of this Agreement shall be determined to be illegal, invalid or unenforceable, that provision shall be severed from this Agreement and the remaining provisions shall continue in full force and effect.

10.9 Governing Law

Notwithstanding anything to the contrary in the Leases, this Agreement and, except as hereinafter provided, all Closing Documents shall be governed by and interpreted and enforced in accordance with the laws of the Province of Ontario and the federal laws of Canada applicable therein. The Lease Amendment and Surrender Agreements shall be governed by the Laws of the applicable Province in which the Premises are located and to which such agreements relate.

10.10 Submission to Jurisdiction

- (a) Each party to this Agreement submits to the exclusive jurisdiction of the Court in any action, application, reference or other proceeding arising out of or related to this Agreement or any of the Leases or the Guarantees and agrees that all claims in respect of any such actions, application, reference or other proceeding shall be heard and determined in the CCAA Proceedings before the Court.
- (b) Each of the parties irrevocably waives, to the fullest extent it may effectively do so, the defence of an inconvenient forum to the maintenance of such action, application or proceeding.
- (c) A final judgment in any such action, application or proceeding is conclusive and may be enforced in other jurisdictions by suit on the judgment or in any other manner specified by law.

10.11 Counterparts and Execution

This Agreement and any amendments thereto (and any other agreements, notices or documents contemplated thereby) may be executed and delivered by facsimile or electronic transmission (including electronic transmission via the Internet) and in any number of counterparts and all such facsimile or electronic copies and counterparts shall be deemed to be an original hereof and for all purposes constitute one agreement, be binding on the parties hereto, provided each party

hereto has executed and delivered at least one counterpart, and each may be relied upon by each party hereto as such for any and all purposes.

10.12 French Language

The parties hereto have requested that this Agreement be drafted in English only. Les parties aux présentes ont demandé à ce que la présente convention soit redigée en anglais seulement.

10.13 Statute References

Any reference in this Agreement to any statute or any section thereof shall, unless otherwise expressly stated, be deemed to be a reference to such statute or section as amended, restated or re-enacted from time to time.

10.14 Headings

The division of this Agreement into Sections, the insertion of headings is for convenience of reference only and are not to be considered in, and shall not affect, the construction or interpretation of any provision of this Agreement.

10.15 References.

Where in this Agreement reference is made to an article or section, the reference is to an article or section in this Agreement unless the context indicates the reference is to some other agreement. The terms "this Agreement", "hereof", "hereunder" and similar expressions refer to this Agreement and not to any particular Article, Section or other portion hereof and include any agreement supplemental hereto.

10.16 Number and Gender

Unless the context requires otherwise, words importing the singular include the plural and vice versa and words importing gender include all genders.

10.17 Use of the word "including" and "or" etc.

The word "includes" or "including" shall mean "includes without limitation" or "including without limitation", respectively. The word "or" is not exclusive.

10.18 Business Days

If any payment is required to be made or other action is required to be taken pursuant to this Agreement on a day which is not a Business Day, then such payment or action shall be made or taken on the next Business Day. All actions to be made or taken by a particular Business Day must be made or taken by no later than 5:00 p.m. (Toronto time) on a Business Day and any action made or taken thereafter shall be deemed to have been made and received on the next Business Day.

10.19 Currency and Payment Obligations

Except as otherwise expressly provided in this Agreement all dollar amounts referred to in this

Agreement are stated in Canadian Dollars.

10.20 Accounting Principles

All calculations made or referred to herein shall be made in accordance with the accounting standards required by the applicable Leases.

10.21 Notice

- (a) Notwithstanding anything to the contrary in the Lease, any notice, consent or approval required or permitted to be given in connection with this Agreement or the Lease (in this Section referred to as a "Notice") shall be in writing and shall be sufficiently given if delivered (whether in person, by courier service or other personal method of delivery), or if transmitted by facsimile or e-mail to the address for notice of the relevant party set out in Schedule "M".
- (b) A Notice is deemed to be given and received (i) if sent by personal delivery or same day courier, on the date of delivery if it is a Business Day and the delivery was made prior to 5:00 p.m. (local time in the place of receipt) and otherwise on the next Business Day, (ii) if sent by overnight courier, on the next Business Day, or (iii) if transmitted by facsimile with confirmation of transmission by the originating facsimile before 5:00 p.m. (local time in the place of receipt), on the same Business Day and otherwise on the next Business Day following such transmission with confirmation, or (iv) if sent by email, on the day of transmission or, if such day is not a Business Day, on the next following Business Day. Any party may change its address for service from time to time by providing a Notice in accordance with the foregoing. Any subsequent Notice must be sent to the party at its changed address. Any element of a party's address that is not specifically changed in a Notice will be assumed not to be changed. Sending a copy of a Notice to a party's legal counsel as contemplated above is for information purposes only and does not constitute delivery of the Notice to that party. The failure to send a copy of a Notice to legal counsel does not invalidate delivery of that Notice to a party.

10.22 Risk of Loss

The Real Property Interests shall be and remain until Closing at the risk of the Tenant. In the event of material damage by fire or other hazard to the Real Property Interests or any part thereof occurring before the Closing Date, the Landlord Entities shall take the proceeds of any insurance available or actually paid to the Tenant, and applying same on account of the Consideration and completing the transaction otherwise in accordance with the terms hereof.

[Remainder of Page Intentionally Left Blank]

IN WITNESS WHEREOF the Tenant has executed this Lease Transaction Agreement.

TARGET CANADA CO.

Зу;

Name: Aaron Alt Title: CEO IN WITNESS WHEREOF each of the Landlord Entities has executed this Lease Transaction Agreement. HILLCREST HOLDINGS INC. By: Gáyajrýs E. Smart Name: Title: By: Name: Kieran F. Mulroy Title: Vice President MONTEZ HILLCREST INC. By: Name: Title: By: Name: Title: OXFORD PROPERTIES RÉTAIL HOLDINGS II INC. By: Name: Gawain SE Smart Title: By: Name: Kieran F. Mulroy Title: Vice President CPPIB UPPER CANADA MALL INC. By: Name: Title: By: Name:

Title:

IN WITNESS WHEREOF each of the Landlord Entities has executed this Lease Transaction Agreement.

HILLCREST HOLDINGS INC.

By:			reinch.
•	Name:		
	Title:		
By:			
<i>-</i>	Name:	and the second second to the second to the second s	seembler
	Title:		
	i itic.		
RAON	merz HII	LCREST INC.	
WO	TEZ III	ECREST INC.	
	ille s		
D		Letty	
By:			nujerinismi
	Name:	Manfred Lau	
	Title	A.S.O,	
Ву:		Trevor McIntyre	(Mirrossin)
	Name!	Trevor McIntyre	
	Title:	A.S.O.	
OXF	ORD PRO	PERTIES RETAIL	
HOI	DINGS II	INC.	
By:			
	Name:		appenden
	Title:		
Ву:	Title.		
Dy.	Name:	the state of the s	between ned a
	Title:		
	ritie:		
CPF	IB OPPER	R CANADA MALL INC.	
Ву:	300000000000000000000000000000000000000		
	Name:		
	Title:		
By:			
•	Name:		
	Title:		

IN WITNESS WHEREOF each of the Landlord Entities has executed this Lease Transaction Agreement.

HILLCREST HOLDINGS INC.

By:		
	Name:	
	Title:	
By:		
	Name:	
	Title:	
MOI	VTEZ HII	LCREST INC.
MIOI	VIII III	ECKEST INC.
By:		
-	Name:	
	Title:	
By:		
	Name:	
	Title:	
OVE		PERTIES RETAIL
	DINGS II	
HOL	DINGS II	ine.
By:		
•	Name:	
	Title:	
By:		
	Name:	
	Title:	
~		
CPP	IB UPPER	CANADA MALL INC.
	1	0 01
By:	\sim \mathcal{M}_{\odot}	trevel 1
_	Name:	Sharm Powell
	Title:	Authorized Signatory /
By:		
• ,	Name	Marco Ding
	Title:	
		Authorized Signatory

	RS REALTY MANAGEMENT
CORI	PORATION
By:	Y N
	Name: Gawaio S. E. Smart
	Title: $\mathcal{J}_{2}\mathcal{J}_{3}\mathcal{J}_{0}$.
By:	1/ Nachon
	Name: Kieran F. Mulroy
	Title: Vice President
SOM.	ARE ONE PROPERTY
COR	PORATION
COM	
By:	<u> </u>
	Name: Gawain'S E Smart
	Title: 19 ident
By:	1 VI (Street)
	Name: / Kieran F. Mulroy
	Title: Vice President
WINIA.	GSWAY GARDEN HOLDINGS INC.
L III	GSVAI GARDEN HODDINGS INC.
Ву:	<u> </u>
J-	Name: Gawain Smart
	Title: Vice Resident
Ву:	OF //Wyon
	Name:
	Title: Kieran F. Mulroy Vice President
	AICE Liegingur
8 3D C	GALERIES DE LA CAPITALE
	LDINGS INC.
ILVI	Direction of the control of the cont
	4/-
Ву:	See
٠ ر س	Name: Gawain E. Smart
	Title: Vice/Fresident
Ву:	10 Var
•	Name: Karen Tsang
	Title: Assistant Secretary

OMERS REALTY MANAGEMENT CORPORATION

	Name:	
	Title:	
By:		
•	Name:	
	Title:	
SOU	ARE ONE	PROPERTY
	PORATIO	
		
By:		
•	Name:	
	Title:	
By:		
_ ,	Name:	
	Title:	
	11,101	
KING	GSWAY (GARDEN HOLDINGS INC.
By:		
J -	Name:	
	Title:	
By:	Titto.	
L) y.	Name:	
	Title:	
	Tuic.	
TTC	CALEDII	ES DE LA CAPITALE
	DINGS I	
1101	DINGS II	
By:	N	1 hevel
IJy.	Name:	Sharm Powell
		Authorized Signatory
D	Title:	
By:		
	NT	of the land of
	Name:	Marco Ding

	· ·
IVA	NHOE CAMBRIDGE II INC
. 2 7 2 8.	
***	///// / / / / / / / / / / / / / / / /
By:	NC Claude Sirgis
	Title: Co-Chef de l'Exploitation et Vice-président exécutif, Québec
By:	Title: No Chief Operating Officer and Executive Vice Procedent, Quebec
٠,٠	Name: Lorna J. Telfer
	Title: Executive Vice President
	Legal Affairs and Corporate Secretary
CAT	4
INC	RREFOUR DE L'ESTRIE HOLDINGS,
MIN C.	•
By:	
	Name:
70	Title:
Ву:	N T
	Name: Title:
	. Title.
PLA	CE LAURIER HOLDINGS INC/
_	
By:	- Valude Strois
	Name: Co-Chef de l'Exploitation et Vice-président exécutif, Québec
Ву:	Title: Co-Chief Operating Officer and Executive Vice Concident, Quebec
1.Jy.	Name: Vana J. Telfer.
	Title: Lomay Teller.
	Executive High Stability Secretary Legal Affairs and Corporate Secretary
	MAC MALL LIMITED
	TNERSHIP, by its general partner
4239	474 CANADA INC
1	
By:	Chande Sirals
•	Name: Co-Chef de Exploitation et Vice-président exécutif, Québec
	Title: Ca-Chief Operating Officer and Executive Vice Crasident, Quebec
By:	MAN
	Name:
	Title: Lorna J. Telfer Executive Vice President,
	Legal Affairs and Corporate Secretary

IVANHOE CAMBRIDGE II INC.

By:		_
•	Name:	
	Title:	
By:		
	Name:	•
	Title:	
		•
		TANK TO THE TANK THE
		DE L'ESTRIE HOLDINGS,
INC.	BY ITS	MANTASER IVANHUE
	CAMB	RAJKE 190.
-		
By:		CARTAGOS Sirois
	Mame:	Co-Chef de l'Exploitation et Vice-président exécutif, Québec
	Tifle:	Co-Chief Operating Officer and Executive Vice President, Qu
Вуг		LO ELINO
	Name:	Emilio Elisio
	Title:	Vice-président, Affaires juridiques
		et Chef adjoint du contentieux
		THE
PLA	CE LAURU	ER HOLDINGS INC.
	•	· ·
Ву:	~ ~	AND THE RESIDENCE OF THE PROPERTY OF THE PROPE
	Name:	•
_	Title:	•
By:		
	Name:	
	Title:	
		÷ .
		· · · · · · · · · · · · · · · · · · ·
		L LIMITED
		by its general partner,
4239	474 CANAI	DAINC.
		•
By:		
	Name:	
	Title:	,
By:		,
	Name:	
	Title:	

OAKRIDGE CENTRE VANCOUVER
HOLDINGS, INC.

By:

Name: Co-Chef de l'Exploitation et Vice-président exécutif, Québec
Title: Co-Chief Operating Officer and Executive Vice President, Quebec
Name:
Title:

Lorna J. Telfer
Executive Vice President,
Segal Affairs and Corporate Secretary

INTERVENTION

TARGET CORPORATION HEREBY INTERVENES to this Agreement for the purpose of acknowledging and agreeing to provide the Release of Tenant Claims in Section 2.4 and of covenanting to comply with the provisions of Sections 9.5 and 10.1.

TARGET CORPORATION

Name: Corey Haaland

Title: Senior Vice President

INTERVENTION

ALVAREZ & MARSAL CANADA INC., solely in its capacity as court appointed monitor of Target Canada Co. pursuant to the Initial Order and not in its personal or corporate capacity agrees to comply with the provisions of Sections 5.6(h), 8.4, 8.5, 9.5 and 10.1 hereof.

> ALVAREZ & MARSAL CANADA INC. solely in its capacity as court appointed Monitor of Target Canada Co. and not in its personal or

corporate capacity

By:

Name: Douglas R. McIntosh

Title: President

INTERVENTION

TARGET CORPORATION HEREBY INTERVENES to this Agreement for the purpose of acknowledging and agreeing to provide the Release of Tenant Claims in Section 2.4 and of covenanting to comply with the provisions of Sections 9.5 and 10.1.

TARGET CORPORATION

Ву:			
	Name:	Corey Haaland	
	Title:	Senior Vice President	

INTERVENTION

ALVAREZ & MARSAL CANADA INC., solely in its capacity as court appointed monitor of Target Canada Co. pursuant to the Initial Order and not in its personal or corporate capacity agrees to comply with the provisions of Sections 5.6(h), 8.4, 8.5, 9.5 and 10.1 hereof.

ALVAREZ & MARSAL CANADA INC.

solely in its capacity as court appointed Monitor of Target Canada Co. and not in its personal or corporate capacity

By:

Name: Douglas R. McIntosh

Title: President

SCHEDULE "A" DEFINED TERMS

- "Affiliate" of any Person means, at the time such determination is being made, any other entity controlling, controlled by or under common control with such first entity, in each case, whether directly or indirectly, and "control" and any derivation thereof means the possession, directly or indirectly, of the power to direct or significantly influence the management, policies, business or affairs of an entity whether through the ownership of voting securities or otherwise.
- "Agent" has the meaning set out in the Sale Order.
- "Agency Agreement" means the Agency Agreement dated January 29, 2015 approved by the Court under the Sale Order.
- "Agreement" means this Lease Transaction Agreement and all schedules and instruments in amendment or confirmation of it.
- "Appeals" has the meaning set out in Section 3.4.
- "Applicants" has the meaning specified in recital "B".
- "Approval Order" means an order issued by the Court approving this Agreement and the Transaction, in substantially the form attached as Schedule "L".
- "ARQST" means An Act respecting the Quebec Sales Tax, CQLR, c. T-0.1, as amended, restated, supplemented or substituted from time to time.
- "BCPSTA" means the *Provincial Sales Tax Act* (British Columbia), SBC 2012, Chapter 35, as amended, restated, supplemented or substituted from time to time.
- "BC PST" means the provincial sales tax under the BCPSTA.
- "Business Day" means any day of the year, other than a Saturday, Sunday or any days on which major banks are closed for business in Toronto, Ontario and in Montréal, Québec.
- "CCAA" means the *Companies' Creditors Arrangement Act*, R.S.C. 1985, c. C-36, as amended, restated, supplemented or substituted from time to time.
- "CCAA Proceedings" mean the proceedings pursuant to which the Applicants (including the Tenant) filed for and were granted protection by the Court under CCAA.
- "Claims" means claims, demands, complaints, grievances, actions, applications, suits, causes of action, Orders, charges, indictments, prosecutions, information or other similar processes, assessments or reassessments, judgments, debts, liabilities, expenses, costs, damages or losses, contingent or otherwise, whether liquidated or unliquidated, matured or unmatured, disputed or undisputed, contractual, legal or equitable, including loss of value, professional fees, including fees and disbursements of legal counsel, and all costs incurred in investigating or pursuing any of the foregoing or any proceeding relating to any of the foregoing.
- "Closing" has the meaning specified in Section 8.1.

- "Closing Date" has the meaning specified in Section 8.1.
- "Closing Documents" means those documents and deliveries, except the Net Consideration, to be delivered in connection with the Closing as contemplated in this Agreement including, without limitation, those set out in Section 8.2.
- "Confidential Information" has the meaning specified in Section 10.1(a).
- "Consideration" has the meaning specified in Section 2.1.
- "Court" means the Ontario Superior Court of justice (Commercial List).
- "Deposit" has the meaning set out in Section 2.1(d).
- "Encumbrance" means any restriction, charge, hypothec, lien, encumbrance, claim, prior claim, liability (direct, indirect, absolute or contingent), obligation, trust, deemed trust, right of retention, judgment, writ of seizure or execution, notice of sale, contractual right, or any other right or interest, whether or not registered, published or filed, statutory or otherwise, secured or unsecured.
- "ETA" the Excise Tax Act, R.S.C., 1985, c. E-15, as amended, restated, supplemented or substituted from time to time.
- "Excluded Assets" means those assets (in each case, as of the Closing Date) described in Schedule "D".
- "FF&E" means the tools, signs referred to in Schedule "J1", furniture, machinery, equipment, furnishings and trade fixtures located at any of the Premises and owned or leased by the Tenant or its subtenants or licensees, if any, including, without limitation, in all cases, the items listed in Schedule "J1" and excluding, without limitation, in all cases, the items listed in Schedule "J2".
- "GST/HST" means goods and services tax or harmonized sales tax under the ETA.
- "Governmental Authorities" means governments, regulatory authorities, governmental departments, agencies, agents, commissions, bureaus, officials, ministers, Crown corporations, courts, bodies, boards, tribunals or dispute settlement panels or other law or regulation-making organizations or entities:
- (a) having or purporting to have jurisdiction on behalf of any nation, province, territory or state or any other geographic or political subdivision of any of them; or
- (b) exercising, or entitled or purporting to exercise any administrative, executive, judicial, legislative, policy, regulatory or taxing authority or power;
- "Guarantee" means a guarantee or indemnity provided by Target to an individual Landlord Entity of certain liabilities and obligations of the Tenant under an individual applicable Lease, and "Guarantees" mean all of such guarantees and indemnities collectively.
- "HBC Entities" mean collectively, Zellers Inc. and Hudson's Bay Company and their respective successors and assigns.

- "Interim Period" means the period between the close of business on the date of this Agreement and the Closing.
- "Initial Order" has the meaning set out in recital B.
- "Inventory" means all inventory, stock, supplies and all other items to be sold from any of the Premises, excluding, for greater certainty, the items listed in Schedule "J2".
- "Landlord" means, in respect of a Lease or a Guarantee, the landlord of that Lease or the recipient of the Guarantee.
- "Landlord Entity" means each of the entities listed under the column "Landlord Entity(ies)" in Schedule "B", each together with any other entity included in the same row being all the Landlord(s) of the Lease at the Location set out opposite to its name, and "Landlord Entities" means all such entities, collectively, but subject to Section 1.1.
- "Landlord Releasees" mean collectively, the Landlord Entities and their respective Affiliates, directors, officers, employees, agents, shareholders, members, partners, general partners, limited partners, and successors and permitted assigns.
- "Laws" means any and all applicable laws, including all statutes, codes, ordinances, decrees, rules, regulations, municipal by-laws, judicial or arbitral or administrative or ministerial or departmental or regulatory judgments, orders, decisions, ruling or awards, and general principles of common and civil law and equity, binding on or affecting the entity referred to in the context in which the word is used.
- "Leasebacks" has the meaning set out in Section 5.6.
- "Leaseback Period" means, in respect of each Leaseback, the period from and including the Closing Date to and including the Vacancy Date of such Leaseback.
- "Leaseback Premises" has the meaning set out in Section 5.6.
- "Leases" means the Tenant's leases and other related agreements for the Premises, as same are assigned, amended, restated, renewed or supplemented from time to time, including but not limited to those documents listed on Schedule "B1".
- "Lease Amendment and Surrender Agreement" means the lease amendment and surrender agreement substantially in the form attached as Schedule "I"
- "Locations" means the locations of the Tenant listed in Schedule "B".
- "Monitor" has the meaning set out in recital B.
- "Monitor's Certificate" means the certificate filed with the Court by the Monitor certifying receipt of confirmation from the Landlord Entities and the Tenant that all conditions of Closing in Sections 7.1, 7.2 and 7.3 of this Agreement have been satisfied or waived.
- "Net Consideration" has the meaning set out in Section 8.2(c)(i).

"Orders" means orders, injunctions, judgments, administrative complaints, decrees, rulings, awards, assessments, directions, instructions, penalties or sanctions issued, filed or imposed by any Governmental Authority or arbitrator.

"Outside Date" has the meaning ascribed to such expression in Section 9.4.

"Permitted Encumbrances" means, collectively: (a) any Encumbrances encumbering the fee simple, ownership or Landlord Entity interest of the properties on which the Premises are located, but excluding, for greater certainty, any Encumbrances which were created or caused by the Tenant or were consented to by the Tenant (unless at the request of a Landlord Entity); and any Encumbrances charging the leasehold interest (or the rights of the Tenant as lessee) in and to the properties on which the Premises are located; (b) the Leasebacks; and (c) the Site Plan Agreement.

"Person" means a natural person, a partnership of any type, a corporation, a joint venture, a syndicate, a chartered bank, a trust, a trust company, a government or an agency thereof, an unincorporated association, or an executor, administrator or other legal representative.

"Premises" means the lands and premises which are leased by the Tenant at the Locations.

"Properties" or "Property" has the meaning set out in Section 2.3.

"QST" means Quebec sales tax under the ARQST.

"Real Property Interests" means all properties, assets, interests and rights of the Tenant which are related to the operation at each of the Premises, which for greater certainty include the assets in Schedule "C" but do not include Excluded Assets or any items listed in Schedule "J1".

"Release Matters" has the meaning set out in Section 2.3.

"Release of Landlord Claims" has the meaning set out in Section 2.3.

"Release of Tenant Claims" has the meaning set out in Section 2.4.

"Representatives" means, with respect to a party, its directors, officers, employees, agents or advisors, lenders, solicitors or other consultants retained for the purpose of the Transaction contemplated hereunder.

"Sale Order" means the Order of the Court granted on February 4, 2015 approving the sale under the Agency Agreement through the Tenant's Agent of certain assets of the Tenant including chattels, personal property, Inventory, FF&E and any other Excluded Assets.

"Site Plan Agreement" means the site plan agreement and related documents listed in Schedule "F".

"Target" means Target Corporation.

"Tax Certificate, Undertaking and Indemnity" has the meaning set out in Section 6.4(a).

"Taxes" means land transfer, mutation, sales, goods and services, harmonized sales, use, value-

added, excise, stamp or similar taxes, including, without limitation, QST, GST/HST and BC PST.

"Tax Refunds" has the meaning set out in Section 3.4.

"Tenant Affiliate Releasees" mean collectively, the Applicants and Target and all predecessors in interest to the Tenant under any of the Leases (other than the HBC Entities) and each of their respective Affiliates, directors, officers, employees, agents, shareholders, members, partners, general partners, limited partners, successors and permitted assigns.

"Tenant Releasees" mean collectively, the Tenant Affiliate Releasees and the HBC Entities.

"Transaction" means, collectively, the surrender of the Real Property Interests and Leasebacks contemplated in this Agreement.

"Vacancy Date" means, in respect of a Leaseback, June 30, 2015, unless advanced by the Tenant in accordance with Section 5.6(i).

SCHEDULE "B" LOCATIONS & CORRESPONDING LANDLORD ENTITIES

Store#	Locations	Address	Postal Code	City	Province	Landlord Group	Landlord Entity(ies)
3590	Carrefour De L'Estrie	3050 boul. de Portland	J1L 1K1	Sherbrooke	QC	IC / CPPIB	Carrefour de L'Estrie Holdings, Inc.
3672	Conestoga Mall	550 King St N	N2L 5W6	Waterloo	ON	IC	Ivanhoe Cambridge II Inc.
3666	Hillcrest Mall	9350 Yonge St	L4C 5G2	Richmond Hill	ON	Oxford / Montez	Hillcrest Holdings Inc. & Montez Hillcrest Inc.
3561	Kingsway Garden Mall	1 Kingsway Garden Mall, Suite 60	T5G 3A6	Edmonton	AB	Oxford	Kingsway Garden I-Ioldings Inc.
3696	Les Galeries De La Capitale	5401 boul. des Galeries	G2K 1N4	Quebec City	QC	Oxford / CPPIB	Les Galeries De La Capitale Holdings Inc.
3717	Metropolis At Metrotown	4545 Central Blvd	V5H 4J1	Burnaby	BC	IC	Ivanhoe Cambridge II Inc.
3697	Mic Mac Mall	21 MicMac Rd	B3A 4K7	Halifax	NS	IC	Mic Mac Mall Limited Partnership by its general partner, 4239474 Canada Inc.
3711 (not open)	Oakridge Centre	650 West 41st Avenue	V5Z 2M9	Vancouver	BC	IC	Oakridge Centre Vancouver Holdings, Inc.
3755	Place Laurier	2700 boul. Laurier	G1V 2L8	Quebec City	QC	IC	Place Laurier Holdings Inc.
3636	Square One	100 City Center Dr	L5B 2C9	Mississauga	ON	Oxford / AIMCo	OMERS Realty Management Corporation & Square One Property Corporation
3565	Upper Canada Mall	17600 Yonge St	L3Y 4Z1	Newmarket	ON	Oxford / CPPIB	Oxford Properties Retail Holdings II Inc. & CPPIB Upper Canada Mall Inc.

SCHEDULE "B1" LEASES

SCHEDULE "B1"

T#	Named	List of All Leasing and Related Documents
Store ID	Landlord	
	on Lease	
3590 Z 0199	Carrefour de l'Estrie	Letter dated July 23, 2009 from Zellers to Carrefour de l'Estrie Holdings Inc. providing notice of Zellers exercising their right to extend the term of the lease to August 31, 2015.
	Holdings	Estoppel Certificate dated January 22, 2009 between Compagnie d'Assurance Standard Life du Canada and Carrefour de l'Estrie
Carrefour de l'Estrie	Inc.	Notice to tenants that Carrefour de l'Estrie Inc. has transferred their interest in the leases to Carrefour de l'Estrie Holdings Inc. and that Redcliff has been hired to manage the shopping centre as of December 2005
		Tenant Certificate dated November 8, 2005 executed by Zellers Inc.
Sherbrooke		Letter dated August 13, 2004 extending the lease until August 31, 2010. Letter dated June 11, 2004 from the Office of the Senior Vice President Real Estate and Devalorment for Zellong dated.
Québec		Zellers is able to proceed with a pharmacy at this location.
		Contract for Rental of a Space on Common Pylon dated November 27, 2002 between Carrefour de l'Estrie Inc. and Zellers Inc.
		Letter dated November 23, 1990 whereby the contributions to the Merchant's Association (now the Promotion Fund) were
		increased.
		Memorial for Registration of Lease between Zellers Inc. and Carrefour de l'Estrie Inc dated March 6, 1986.
		Lease dated August 16, 1985 between Zellers Inc. and Carrefour de l'Estrie Inc.
		Offer of Lease and revisions dated August 1984. The recurred Lease dated Auril 27 1084.
		Contract for rental of a space on common polon between Zellers Inc. and Carrefour de l'Estrie Inc. dated December 2, 2002
		From the state of
		Assignment and Assumption of Lease Agreement dated May 27, 2011 between Zellers and Target Canada Co. First Amendment to Lease dated May 27, 2011 between Target Canada Co. and Carrefour de 1'Estrie Holdings Inc (the "First
		Amendment").
		Registration/Notice of First Amendment to Lease dated June 17, 2011 by Target Canada Co as number 18 234 130.
		Letter Agreement amending First Amendment to the Lease dated September 16, 2011 from Carrefour De L'estrie Holdings Inc. to Target Canada Co.
		Letter from Target Canada Co. to Carrefour de l'Estrie Holdings Inc. dated February 1, 2012 regarding exercising the internal
		expansion option
		Notice to tenants of change of property manager name dated June 4, 2012 by Triovest Realty Advisors Inc. Notice of Amalgamation by Triovest Realty Advisors Inc. dated Inne 8, 2012
		Letter Agreement dated November 20, 2012 between Carrefour De l'Estrie Holdings Inc. and Target Canada Co. regarding the
		development period.
		Pre-Open month-to-month lease between Carrefour de l'Estrie Holdings Inc. and Target Canada Co. dated March 28, 2013. Notice and direction to Tenante (Target Canada Co.) dated Among 7, 2013 to Canada. De 19 Canada Co.
		recess and discussive formula (1 anger Canada Co.) dated August 1, 2013 by Catterout De 1 Estre Holdings Inc. regarding the

		new property manager. Lease renewal letter dated January 13, 2014 from Target Canada Co. to Carrefour De l'Estrie Holdings Inc. regarding lease
		renewal. Letter from Ivanhoe Cambridge to Target Canada Co. dated January 29, 2014.
3666 Z0321	Montez Hillcrest	Letter dated April 3, 1973 from S.S. Kresge Company Limited to The Fairview Corporation Limited amending and enclosing an executed copy of the Basis of Agreement (Offer to Lease) dated April 2, 1973.
Hillcrest Mall	Inc. and Hillcrest	Lease dated August 8, 1974 between The Cadillac Fairview Corporation Limited, as landlord, and S.S. Kresge Company, Limited, as tenant ("Lease").
Richmond Hill	Holdings Inc	Letter dated August 9, 1974 from S.S. Kresge Company Limited to The Cadillac Fairview Corporation Limited notifying
	TITO.	Certificate of the Secretary-Treasurer of the York Regional Land Division Committee dated October 8, 1975 confirming
Ontario		Planning Act consent. Letter dated February 24, 1975 from counsel to S.S. Kresge Company Limited, enclosing a certified copy of Kresge's authority
		to hold land in Ontario and attidavits. Consent Agreement dated October 19, 1989 between The Cadillac Fairview Corporation Limited (Landlord) and K Mart Canada
		Limited (Tenant) whereby Tenant consented to construction by Landlord.
		Estoppel Certificate dated September 3, 1991 from Kmart Canada Limited to The Cadillac Fairview Corporation Limited. Letter Agreement dated February 21, 1994 between Kmart Canada Limited and The Cadillac Fairview Corp. with respect to
		consent for installation of a rooftop satellite dish.
		Estoppel Certificate dated April 26, 1994 from Kmart Canada Limited to The Cadillac Fairview Corporation Limited, Hillcrest Leaseholds Inc and Onfrea Inc.
		Acknowledgement Agreement dated May 11, 1994 between Ontrea Inc., Hillcrest Leaseholds Inc., and Kmart Canada Limited
		whereby the parties agreed to assume and/or confirm the obligations under the Lease.
		Letter dated December 10, 1997 from Kmart's solicitor to Kmart Canada Co. regarding solicitor opinion as to whether Kmart
		could turn the store into a bed and bath merchandising store operating under a different name.
		Letter dated April 27, 1998 from The Cadillac Fairview Corporation Limited to Zellers Inc. enclosing the renewal agreement
		whereby Lease was renewed for term of 5 years commencing September 1, 1999 and ending August 31, 2004.
		terms and conditions on which Premises was expanded.
		Subsurface Investigation and Remedial Excavation Report dated June 9, 2000 from Pinchin Environmental for Zellers Auto
		Centre #321, Richmond Hill, Ontario confirming decommissioning of auto site complies with environmental laws.
		Lease Amending Agreement dated July 20, 2000 between Ontrea Inc., as landlord, and 3584747 Canada Inc., as tenant.
		Letter dated September 13, 2000 to 3584747 Canada Inc. explaining discrepancy in Gross Leasable Area.
		Exiter value vialen 1, 2001 ambumb diamente Opening Date, consultenon and wante and Oross Exasable falsa of expansion
		promises. Letter dated August 15, 2005 from Cadillac Fairview to Hudson's Bay Company clarifying maintenance, repair and renovation
		requirements at the Shopping Centre.
		Lease Amending Agreement dated May 13, 2007 between Ontrea Inc. and Zellers Inc.

Assignment and Assumption Agreement between Zellers Inc. and Target Canada Co. dated June 9, 2011. Sixth Amendment to Lease dated June 9, 2011 ("6th Amendment") between Target Canada Co., Hillcrest Holdings Inc. and Montez Hillcrest Inc. Registration of sixth amendment to lease dated June 9, 2011 receipted as YR1660627. Notice of Assignment of Lessee Interest in Lease dated June 9, 2011 between Zellers Inc. and Target Canada Co. receipted as YR1660625. Notice of algae build up on roof dated March 28, 2012 from Hillcrest Mall Management Inc. to Target Canada Co. Seventh Amendment to Lease dated July 19, 2012 between Target Canada Co., Hillcrest Holdings Inc. and Montez Hillcrest Inc. Eight Amendment to Lease dated December 6, 2012, between Target Canada Co., Hillcrest Holdings Inc. and Montez Hillcrest Inc. Building Development Agreement dated 2012 entered into between Target Canada Co. and Hillcrest Mall Management Inc. License Agreement dated March 21, 2013 between Target Canada Co. and Hillcrest Mall Management Inc.		Certificate of Committee of Adjustment, dated June 28, 1978, in respect of approval of Lease being in excess of 21 years ("Certificate of Committee Adjustment"). Letter, dated September 29, 1978, from K Mart Canada Limited to Cambridge Leaseholds Limited in respect of the opening	Assignment of Lease, dated November 15, 1978, between Cambridge Leaseholds Limited, as assignor, and The Prudential Insurance Company of America, as assignee, registered on December 7, 1978 as instrument no. 645830. Certificate and Agreement, dated December 5, 1978, between K Mart Canada Limited and Cambridge Leaseholds Limited in respect of various tenancy terms ("Certificate and Agreement"). Notice of Lease, dated August 29, 1979, registered on May 1, 1980 as instrument no. 680771. Priority Agreement, dated January 1, 1980, between The Prudential Insurance Company of America and K Mart Canada Limited, registered on August 11, 1980 as instrument no. 686212. Minutes of Settlement, between Cambridge Leaseholds Limited and K Mart Canada Limited in respect of payment of realty taxes.	Emeral Assignment of Leases, dated June 14, 1991, between Select Properties Limited, as assignor, and Cambridge Leaseholds Limited, as assignee in respect of the sale of Select Properties Limited's 50% interest to Cambridge Leaseholds Limited. Tenant Estoppel Certificate, dated December 4, 1991, from K Mart Canada Limited ("1991 Estoppel"). Amendment of Lease Agreement, dated March 31, 1994, between Cambridge Leaseholds Limited and K Mart Canada Limited, in respect of building expansion. Letter, dated May 4, 1995, from IPCF Properties Inc. to K Mart Canada Limited, in respect of consent to expansion. Supplemental Information Sheet, dated January 3, 1996, in respect of the terms to amendment. Certificate of Amendment, registered on May 1, 1997 as instrument no. 1335351, whereby K Mart Canada Limited changed its
	Ivanhoe Cambridge II Inc.			
	3672 Z0334 Conestoga Mall	Kitchener	Ontario	

Certificate of Amalgamation, registered on May 1, 1997 as instrument no. 1335352, whereby K Mart Canada Limited amalgamated with 3002070 Nova Scotia Company to continue as K Mart Canada Co.

Notice of Lease, registered on May 1, 1997 as instrument no. 1335353, registering the Amendment of Lease Agreement, dated March 31, 1994.

Nova Scotia and Caisse De Depot et Placement du Quebec in respect of purchase by Cambridge Shopping Centres Limited of Agreement, dated June 9, 1997, between Cambridge Shopping Centres Limited, The Toronto-Dominion Bank, The Bank of Letter, dated June 27, 1997, from Cambridge Shopping Centres Limited to Kmart Canada Co. enclosing Confirmation and Markborough Properties Inc.

accepted by Hudson's Bay Company ("HBC"), Zellers Inc and 3009354 Nova Scotia Co. ("1998 Letter Amending Agreement") Amendment of Lease, dated January 11, 1999, between Cambridge Shopping Centres Limited, 3009354 Nova Scotia Co. and Amending Letter, dated May 14, 1998, from Cambridge Shopping Centres Limited in respect of amendments to Lease and Zellers Inc. ("1999 Amendment"). Letter, dated February 16, 2000, from Cambridge Shopping Centres Limited to Hudson's Bay Company in respect of conversion of Merchants Association to Promotion Funds accepted by Hudson's Bay Company and Zellers Inc. on March 9, 2000 ("2000

Acknowledgement and Assumption Agreement, dated August 15, 2001, between Ivanhoe Cambridge II Inc., 3859681 Canada Notice to Tenants, dated August 15, 2001, from Ivanhoe Cambridge I Inc, in respect of assignment of landlord's interest to Inc. and Zellers Inc., whereby Ivanhoe Cambridge II Inc. assumes the obligations as landlord of Ivanhoe Cambridge I Inc. Ivanhoe Cambridge II Inc.

Letter, dated January 15, 2003, from 3859681 Canada Inc. to Ivanhoe Cambridge in respect of the exercise of the first renewal. Letter, dated January 30, 2003, from Ivanhoe Cambridge to 38559681 Canada Inc. in respect of letter received outlining corporate change in transition of the tenant from 3009354 Nova Scotia Co. to 38559681 Canada Inc.

Letter agreement, dated June 10, 2004, in respect of reduced parking ratios at various Ivanhoe centres and right of expansion. Letter Agreement, dated June 10, 2004, from Hudson's Bay Company to Ivanhoe Cambridge.

Certificate and Articles of Amalgamation, dated February 1, 2005, whereby Zellers Inc. and 3859681 amalgamate to continue as Letter Agreement, dated September 24, 2004, confirmed by a letter, dated October 27, 2004, to Hudson's Bay Company. Zellers Inc.

Lease amending agreement, dated February 18, 2005.

Lease Extension Notice, dated December 10, 2007, in respect of renewal to August 31, 2013, with corresponding letter. Fenant Estoppel Certificate, dated April 9, 2009.

Letter Agreement between Target Canada Co. and Ivanhoe Cambridge Inc. dated April 29, 2011.

First Amendment to Lease dated May 27, 2011 between Target Canada Co. and Ivanhoe Cambridge II Inc. Assignment and Assumption Agreement dated May 27, 2011 between Zellers Inc. and Target Canada Co.

Notice of Assignment of Lessee Interest in Lease date May 27 2011 by Zellers Inc. and Target Canada Co. receipted as Sublease Agreement between Target Canada Co. and Zellers Inc. dated May 27, 2011.

Notice of Lease Amending Agreement dated May 27, 2011 by Target Canada Co. receipted as WR 616408.

WR616406 and WR616407.

Notice from Target Canada Co. to Ivanhoe Cambridge Inc. dated June 7, 2011 regarding assignment of Zellers leases. Letter from Hudson's Bay Company regarding entry of Target Canada Co. into Shopping Centres, undated

		Letter from Loblaw to Target Canada Co. regarding purchase price dated August 20, 2012. Acknowledgement and Acceptance dated June 3, 2013 from Ivanhoe Cambridge II Inc. to Target Canada Co.
3561 Z0150	Kingsway Garden Holdings	Lease dated February 26, 1974 between Westgreen Developments (North) Ltd., Zeller's (Western) Limited, and Zeller's Limited (the "Lease"). Lease Amending Agreement dated September 26, 1986
Kingsway Mall	Inc.	Leasehold Construction Agreement dated September 26, 1986 Letter Agreement dated November 5, 1985.
Edmonton		Lease Amending Agreement dated October 12, 1976 between Westgreen Developments (North) Ltd., Zeller's (Western) Limited, and Zeller's Limited (the "1976 Amendment").
Alberta		Letter Agreement dated June 26, 1974 between Zeller's (Western) Limited, Zeller's Limited, and Westgreen Developments (North) Ltd. regarding amendments to the Lease (the "1974 Amendment").
		Letter from Hudson's Bay Corporation to Kingsway Garden Holdings Inc. dated March 10, 2009. Letter dated September 9, 2009 from Oxford Properties Group to "Business Partner" regarding the official change of name of
	,	the shopping centre to "Kingsway Mall". Letter Agreement dated February 24, 2006 regarding HVAC obligations
		Estoppel Certificate dated May 29, 1998. Estoppel Certificate dated June 6th, 1991.
		Estoppel Certificate dated June 26, 1986.
		Draft Lease Amending Agreement dated June 2005. [Not executed.] Letter dated February 24, 2006 regarding exercise of the right to renew.
		Lease extension houce from Zeliers inc. to mingsway Garden holdings like, dated rebruary 6, 2011.
		Assignment and Assumption of Lease dated June 9, 2011 between Zellers Inc. and Target Canada Co. Recognition, Non-disturbance and Attornment Agreement dated June 9, 2011 between OMERS Realty Canada Inc, Kingsway
		Garden Holdings, Inc. and Target Canada Co. Caveat Forbidding Registration dated June 9, 2011 by Target Canada Co.
		Fifth Amendment to Lease dated June 9, 2011 between Target Canada Co. and Kingsway Garden Holdings Inc.
		Notice of Assignment of Lease from Larget Canada Co. to Kingsway Gardens Holdings Inc. dated June 14, 2011. Sixth Amendment to Lease dated July 19, 2012 between Target Canada Co. and Kingsway Garden Holdings Inc.
		Agreement between Target Canada Co. and Sears Canada Inc. dated January 14, 2013.
		Seventh Amendment to Lease dated February 13, 2013 between Target Canada Co. and Kingsway Garden Holdings Inc. Building Development Agreement dated February 15, 2013 between Target Canada Co. and Kingsway Garden Holdings Inc.
		Notice of Sublease from Target Canada Co. dated January 31, 2013. Ticense A greenent dated February 25, 2014 between Target Canada Co. and Kinosway Garden Holdings Inc. regarding a traffic
		counter.

3696 Z0410	Les Galeries de la	Lease dated May 11, 2001 between Zellers Inc. and Les Galeries de la Capitale Inc. Letter Aereement dated March 27, 2003 regarding square footage.
T ac Golamac	Capitale	Letter Agreement dated March 7, 2002 regarding possession date and exercise of option to extend Lease term. Tenut Advantagement dated 25, 2000 fem. 7, 21, 22, 21, 21, 21, 21, 21, 21, 21, 21
De La Capitale	Inc.	Form of Tenant Certificate dated November 8, 2005 from Zellers Inc to Osmington Inc. Registration of I ease #10 375 077 dated May 2, 2003 from Descourses Naturalise Ondshee
Québec		
Québec		First Amendment to Lease dated May 27, 2011 between Target Canada Co. and Les Galeries de la Capitale Holdings Inc. (1st Amendment)
		Registration of First Amendment to Lease between Target Canada Co. and Les Galeries de la Capitale Holdings Inc. dated June 1, 2011.
		Notice of Assignment of the Lease to Target Canada Co. addressed to Les Galeries de la Capitale Holdings Inc. and dated June 7, 2011.
		Second Amendment to Lease dated May 31, 2013 between Target Canada Co. and Les Galeries de la Capitale Holdings Inc.
		Assignment and Assumption of Lease dated May 27, 2011 between Zellers Inc. and Target Canada Co. Notice of amendment to commercial lease registered June 1, 2011 between Target Canada Co. and Les Galeries de la Capitale
		Holdings Inc.
		Notice of Amendment to Commercial Lease dated March 20, 2012 between Target Canada Co. and Les Galeries de la Capitale Holdings Inc.
		Notice of Sublease expiration date to Zellers Inc. by Target Canada Co. dated May 16, 2012.
		Nouce of Amalgamation from Triovest dated June 8, 2012.
		Assumption Agreement dated May 51, 2015 between Target Canada Co. and OMERS Realty Corporation. Covenant of Transferee dated April 5, 2013 from Target Canada Co. to Les Galeries de la Capitale Holdings Inc.
		Pre-Open Lease dated March 28, 2013 between Target Canada Co. and Les Galeries de la Capitale Holdings Inc.
		Notice of Potential Strike Impact from Target Canada Co to Les Galeries de la Capitale Holdings Inc. dated July 24, 2013.
		Notice to Tenants (Target Canada Co.) from Les Galeries de la Capitale Holdings Inc. dated June 3 rd , 2013 regarding
		appointment of Orot Management Limited ratmership as property manager. License agreement dated November 20, 2013 between Target Canada Co. and Les Galeries de la Capitale Holdings Inc. in
		respect of traffic counters.
		Snow Removal Agreement dated February 20, 2014 between Target Canada Co. and Oxford Properties. Notice of Sublease from Target Canada Co. to Target Canada Dromerty I D. given to Les Galaries de la Canitale Holdings Inc.
		dated February 28, 2013.
3697	Mic Mac	Letter of Intent between Zellers Inc. and Cambridge Shopping Centres Limited dated October 28, 1998.
Z0411	Mall	Lease dated August 14, 1999 between Zellers Inc. and Cambridge Shopping Centres Limited.
1	Limited	Tenant Estoppel Certificate dated July 28, 1999 to CIBC Mellon Trust Company, The Toronto-Dominion Bank and Caisse de
Mic Mac Mall	Partnership	Depot Et Placement Du Quebec.
Dartmouth	oy its General	Letter dated March 1 2001 from Ivanhoe Cambridoe recardino Gross I easable Area and Kent Commencement Date. Letter dated March 1 2001 from Ivanhoe Cambridoe recardino Gross I easable Area
Car miloam	Constant	Lower during 1, 2001 noin Ivamice Cambridge regarding Oross Leasable Area.

Letter dated August 16, 2001 from Ivanhoe Cambridge I Inc. regarding Change of landlord. Letter Amendment dated May 8, 2003 between Zellers Inc. and Ivanhoe Cambridge II Inc. Letter Agreement dated June 10, 2004 among Hudson's Bay Company, Zellers Inc., and Ivanhoe Cambridge II Inc. Tenant Certificate dated June 16, 2004 to any person becoming second lender or purchaser of the property. Letter Agreement dated September 24, 2004 between Hudson's Bay Company and Zellers Inc. and Ivanhoe Cambridge II Inc. Letter dated October 27, 2004 from Ivanhoe Cambridge II Inc. to Hudson's Bay Company and Zellers Inc. Release dated October 27, 2004 from Zellers Inc. to Mic Mac Mall Limited Partnership, 4239474 Canada Inc, Ivanhoe Cambridge II Inc. Release dated December 7, 2004 from Zellers Inc. to Mic Mac Mall Limited Partnership, 439474 Canada Inc, Ivanhoe Cambridge Inc. and Ivanhoe Cambridge II Inc. Release dated January 3, 2005 regarding rent and operating costs payable under the (Atlantic) Supercentre Lease. Letter Agreement between Target Canada co. and Ivanhoe Cambridge Inc. and Target Canada Co. Notice of Assignment of Icase by Target Canada Co. to Ivanhoe Cambridge Inc. dated June 7, 2011. First Amendment to Lease dated May 27, 2011 between Target Canada Co. and Mic May 27, 2011. Notice of First Amendment to lease registered on July 8, 2011 at the Halifax County Registry as document number 98673941. Notice by Zellers Inc. to Target Corporation dated May 16, 2012 regarding the cessation of pharmacy operations. Notice to Zellers of sublease expiration dated April 15, 2013. Notice of Sublease from Target Canada Co. dated April 15, 2013. Notice of Sublease from Target Canada Co. dated April 15, 2013.	Lease Term Sheet dated March 18, 1993. Lease dated June 9, 1993. Notice to extend the term of the lease by 5 years ending June 8, 2013 dated January 16, 1997. Estoppel Certificate dated November 28, 2001. Assignment of Zellers Inc. Lease dated January 21, 2002. Tenant Acknowledgement dated June 8, 2008. Letter Agreement between Target Canada Co. and Ivanhoe Cambridge Inc. and Target Canada Co. Tenant Acknowledgement to Lease dated May 27, 2011 between Zellers Inc. and Target Canada Co. First Amendment to Lease dated May 27, 2011 between Target Canada Co. and Ivanhoe Cambridge Inc. Assignment and Assumption of Lease by Target Canada Co. to Ivanhoe Cambridge Inc. Sublease Agreement between Target Canada Co. and Zellers Inc. dated May 27, 2011. Notice of Assignment and Assumption of Leases by Target Canada Co. to Ivanhoe Cambridge Inc dated June 7, 2011. Letter from Target Canada Co. to Zellers Inc. regarding sublease expiration date dated May 16, 2012. License Agreement dated November 2013 between Target Canada Co. and Ivanhoe Cambridge II Inc. regarding traffic counters. License Agreement dated December 19, 2013 between Target Canada Co. and Ivanhoe Cambridge II Inc. regarding traffic counters. Notice of Sublease from Target Canada Co. dated April 15, 2013.
Partner 4239474 Canada Inc.	Ivanhoe Cambridge II Inc.
Nova Scotia	3717 Z0440 Metropolis at Metrotown Centre Burnaby British Columbia

3711 (not	Oakridge	Letter Agreement dated November 7, 1991 between Woodward Stores Limited and Cambridge Western Lessaholds Limited
open)	Centre	regarding expansion of Woodwards Department Store
Z0430	Vancouver	Agreement to Lease dated May 22, 1993 between Hudson's Bay Company and Oakridge Centre Vancouver Holdings Inc.
	Holdings	regarding replacement of Woodwards store with Zellers store
Oakridge	Inc.	Lease signed December 18, 2001, but effective June 9, 1993.
Centre		Indemnity dated June 18, 1993 in favour of Cambridge Western Leaseholds Limited and OMERS Realty Corporation from Zellers Inc.
Vancouver		Letter Agreement dated Time 10, 2004 hetween Hindson's Ray Company. Zellers Inc., and Ivanhoe Cambridge II Inc.
		Letter Agreement between Hidson's Bay Company and Zellers Inc. and Ivanhoe Cambridge II Inc.
British		Letter from Ivanhoe Cambridge II Inc. to Hudson's Bay Company dated October 27, 2004.
Columbia		First Amendment to Lease dated May 27, 2011 between Target Canada Co. and Oakridge Centre Vancouver Holdings Inc., by
	-	its manager Ivanhoe Cambridge Inc.
		Assignment and Assumption of Lease Agreement dated May 27, 2011 between Zellers Inc. and Target Canada Co.
		Sublease Agreement between Target Canada Co. and Zellers Inc. dated May 27, 2011.
		second Amendment to Lease dated November 28, 2011.
		Notice of Sublease from Target Canada Co. dated April 15, 2013.
		Services Agreement between Oakridge Centre Vancouver Holdings, Inc. and Target Canada Co. dated April 30, 2013.
		Letter dated March 3, 2014 addressing Rent abatement during Redevelopment Period.
		Letter dated March 3, 2014 by Target Canada Co. requesting reimbursement from Ivanhoe Cambridge Inc. regarding erroneous
		payments.
3755	Place	Offer to Lease between La Société Immobilière Marathon and Zellers Limited, dated June 3, 1982.
Z0481	Laurier	Lease between La Société Immobilière Marathon Limitée and Zellers Inc., dated November 3, 1982.
	Holdings	Memorial for Registration of Lease between La Société Immobilière Marathon Limitée and Zellers Inc., dated March 7, 1986
Place Laurier	Inc.	and registered at the Quebec Land Registry Office under number 1177851.
;		Non-disturbance Agreement between Zellers Inc. and Banque Nationale du Canada, dated April 4, 1986.
Québec		Letter Agreement between La Société Immobilière Marathon Limitée and Zellers Inc. amending Article 36 of Lease, dated March 15, 1993.
Québec		Agreement between Marathon Realty Ltd. and Zellers Inc. concerning a satellite dish installation, dated November 25, 1993.
		Letter Agreement between La Société Immobilière Marathon Limitée and Zellers Inc. amending Article 3 of Lease, dated April
		21, 1994.
		Letter Agreement between Centrixx Realty Holdings Limited, Omers Realty Corporation and Zellers Inc. amending the Lease, dated October 28, 1996
		Notice regarding first option to extend the term dated November 6 1006
		Lease Amending Agreement between CTX Properties Inc. Omers Realty Compration and Zellers Inc. dated November 17
		1999 (Short form of amendment prepared solely for registration murroses)
		Notice for Registration of the Rights Resulting from a Lease, dated April 25, 2000 and registered at the Ouebec Land Registry
		Office under number 1727258.
		Letter Agreement between Zellers Inc. and Place Laurier Holdings Inc. amending the Lease, dated September 8, 2000.

		Letter Agreement between Zellers Inc., Hudson's Bay Company and Ivanhoe Cambridge II Inc. dated June 10, 2004. Letter Agreement between Zellers Inc., Hudson's Bay Company and Ivanhoe Cambridge II Inc. dated September 24, 2004. Notice I after by Frankog Cambridge II Inc. dated October 27, 2004.
-1000		Memorandum of Agreement of Addendum to Lease between Societé de Gestion Place Laurier Inc. by its manager Ivanhoe Cambridge Inc., as Landlord, and Zellers Inc., as Tenant, dated April 7, 2010.
~~~		Letter of Amendment between Zellers Inc., Hudson's Bay Company and Société de Gestion Place Laurier Inc. by its manager Ivanhoe Cambridge Inc. dated April 9, 2010.
		Notice Letter from Hudson's Bay Company to Société de Gestion Place Laurier Inc. c/o Ivanhoe Cambridge Inc. dated May 17, 2010.
		Letter of Amendment between Zellers Inc., Hudson's Bay Company and Société de Gestion Place Laurier Inc. by its manager Ivanhoe Cambridge Inc. dated May 25, 2010.
		Notice Letter from Ivanhoe Cambridge to Hudson's Bay Company dated June 16, 2010.
		Email notice from Ivanhoe Cambridge to Hudson's Bay Company dated December 22, 2010.  Letter Agreement between Target Canada Co. and Ivanhoe Cambridge Inc. dated April 29, 2011
		First Amendment to Lease dated May 27, 2011 between Target Canada Co. and Place Laurier Holdings Inc./Ivanhoe Cambridge Inc.
	eu.	Assignment and Assumption of Lease Agreement dated May 27, 2011 between Zellers Inc. and Target Canada Co.
		Sublease Agreement between Target Canada Co. and Zellers Inc. dated May 27, 2011.  Notice of Amendment to Commercial Lease registered on line 1, 2011, lease between Tornet Canada Co. and Disco I comits.
		Holdings Inc./Ivanhoe Cambridge Inc.
	www.	Notice of Assignment of Zellers Lease to Target Canada Co. by Target Canada Co. to Ivanhoe Cambridge Inc. dated June 7, 2011.
		Letter from Target Canada Co. to Zellers Inc. dated March 20, 2012 regarding sublease expiration date.
		Subsease between 1 auget Canada Co. and 1 auget Canada Froperty Lr dated March 4, 2013. Notice of Potential Strike Impact dated July 24, 2013 from Target Canada Co. to Ivanhoe Cambridge Inc.
		Temporary Occupancy Agreement between Place Laurier Holdings Inc. by its manager Ivanhoe Cambridge Inc. and Target Canada Co. dated October 11, 2013.
		Temporary Occupancy Extension Agreement between Place Laurier Holdings Inc. by its manager Ivanhoe Cambridge Inc. and Target Canada Co. dated July 11, 2013.
		Temporary Occupancy Extension Agreement between Place Laurier Holdings Inc. by its manager Ivanhoe Cambridge Inc. and Target Canada Co., dated November 11, 2013.
		Notice of Assignment of Sublease from Target Canada LP to Target Canada Property LLC given to Place Laurier Holdings Inc.
		by its manger Ivanhoe Cambridge Inc. dated November 25, 2013.
		Temporary Occupancy License Agreement between Place Laurier Holdings Inc. by its manager Ivanhoe Cambridge Inc. and Target Canada Co. dated Fehrnary 13, 2013
3636 <b>Z</b> 0279	[156 Square One	Letter Agreement dated November 25, 1985 between The T. Eaton Company Limited and Mascan Corporation and Hammerson Canada Inc.
	Limited]	Letter Agreement dated August 11, 1986 between The T. Eaton Company Limited and Mascan Corporation and Hammerson
Square One	and OMERS	Canada Inc. (Guarantor of Mascan Corporation's obligations) to which a schedule of lease terms was attached.

Shopping Centre	Realty Managemen	Purchase Agreement dated October 8, 1999 between Hudson's Bay Company and The T. Eaton Company Limited.  Agreement dated November 19, 1999 containing inter alia amendments to August 11, 1986 Letter Agreement.  Consent to assignment dated November 23, 1000
Mississauga	Corporation	Assignment and Assumption of Lease Agreement dated December 2, 1999; Lease dated July 1, 2002.
Ontario		Notice and Direction from OMERS Realty Management Corporation to all tenants of the shopping centre dated December 17, 2007 regarding change in property management.
	-14	Assignment and Assumption of Lease Agreement dated June 9, 2011 between Zellers and Target Canada Co. Notice of Assignment of Lease interest in Lease dated June 0, 2011 by Zellers Inc. and Target Canada Co.
		First Amendment to Lease dated June 9, 2011 between Target Canada Co. and OMERS Realty Management Corporation and 156 Square One Limited (the "First Amendment").
		Notice of Assignment of Zellers Leases to Target Canada Co. dated June 14, 2011 by Target Canada Co. to OMERS Realty Management Cornoration and 156 Square One Limited
		Second Amendment to Lease dated July 12, 2012 between Target Canada Co. and OMERS Realty Management Corporation
		and 156 Square One Limited.  Agreement regarding partial surrender of lease dated December 3, 2013, between Target Canada Co, and OMERS Realty
		Management Corporation and 156 Square One Limited.
		[NTD: registered title transferred by 156 Square One Limited as beneficial owner to Square One Property Corporation
		as trustee but no notice of assignment]
3565	CPPIB	Lease dated December 10, 1973 between Zeller's Limited and Regional Shopping Centres Limited.
Z0157	Upper	Assignment Agreement dated December 11, 1973 between Prudential Insurance Company of America as Assignee and Regional
;	Canada Mall	Shopping Centres Limited as Assignor.
Upper Canada	Inc.	Sublease dated November 12, 1974 between Zeller's Drug Store Limited as Subtenant and Zeller's Limited as landlord.
INIAII	Oxford	Area Ceruncate dated May 17, 1987. Letter amending Lease dated November 13, 1987, amended by letter dated May 5, 1988, between Cambridge Leaseholds
Newmarket	Properties	Limited and Zellers Inc.
Ontario	Retail   Holdings II	Amendment of Lease dates May 6, 1988 between Zellers Inc., the Prudential Insurance Company of America and Regional
	Inc.	Estoppel Certificate dated July 26, 1988 to the Prudential Insurance Company of America.
		Estoppel Certificate dated November 5, 1990 to the Prudential Insurance Company of America.
		Estoppel Certificate dated December 6, 1993 to OMERS Realty Corporation.
		Letter dated August 22, 1995 regarding Tax Dispute.
		Letter regarding Matters of Mutual Interest dated June 10, 2004 between Zellers Inc., Hudson's Bay Company and Ivanhoe
		Camoungs II IIIC.
		Ectical Agreement from Ivaluities Cambridge to Induson's Bay and Zellers inc. dated September 24, 2004. Sublease dated October 28, 2001 between Zellers Inc. and 3859681 Canada Inc
		Letter dated October 17, 2005 between Ivanhoe Cambridge Inc. and Zellers Inc.
		Lease Amending Agreement dated May 17, 2007 between Zellers Inc. and Upper Canada Mall Limited.
		Area Certificate dated March 28, 2008.

	Estoppel Certificate dated June 17, 2010 to OMERS Realty Corporation.
10.8 PA	Assignment and Assumption of Lease Agreement dated June 9, 2011 between Zellers Inc. and Target Canada Co.
	Fourth Amendment to Lease dated June 9, 2011 between Target Canada Co. and Upper Canada Mall Limited.
	Application to Register Fourth Amendment to Lease dated June 9, 2011 and receipted as YR1660622.
	Recognition, Non-Disturbance and Attornment Agreement dated June 9, 2011 between Oxford Properties Retail Holdings Inc.,
	Oxford Properties Retail Holdings II Inc., Upper Canada Mall Limited and Target Canada Co. to recognize subtenant's rights
	under the sublease and the non-disturbance of Target Canada Co.'s possession of the subleased premises.
	Planning Act statement by Zellers Inc. dated June 9, 2011.
	Notice of Assignment of Lessee Interest in Lease dated June 9, 2011 by Zellers Inc. and Target Canada Co. receipted as
	YR1660621.
	Notice of Lease registered on June 9, 2011 by Target Canada Co. receipted as YR1660623.
	Notice regarding Sublease Expiration date dated December 27, 2011 by Target Canada Co. to Zellers Inc.
	Fifth Amendment to Lease dated July 19, 2012 between Target Canada Co. and Upper Canada Mall Limited.
	License agreement dated October 10, 2012 between Target Canada Co. and Upper Canada Mall Limited.
	Notice dated June 4, 2013 to all tenants in respect of changing the landlord to Oxford Properties Retail Holdings II Inc. and
	CPPIB Upper Canada Mall Inc.

# SCHEDULE "C" REAL PROPERTY INTERESTS

- 1. The Tenant's right, title and interest, including its leasehold interest at common law and its rights as tenant at civil law, in and to the Leases and the Premises, including, if any, the benefit of all easements, restrictive covenants, access rights, licences to use any common areas or rooftop areas of the buildings or shopping centres of which the Premises form part, rights to renew or extend the term of any Lease, options and similar rights of first offer, first refusal or first opportunity or otherwise to lease or purchase (if any), parking rights and signage rights; and
- 2. The FF&E which are left on the Premises on the Vacancy Date and which are not required to be removed by the Tenant pursuant to Schedule "J1".

# SCHEDULE "D" EXCLUDED ASSETS

- 1. All chattels and personal property which are owned by the Tenant or its subtenants, licensees or other parties;
- 2. All intellectual property or proprietary rights, whether registered or not, and any intangible property, owned, used or held by the Tenant;
- 3. All items, materials and signs bearing the logo, trade-mark, trade-name or business name or other mark or design of the Tenant;
- 4. All Inventory;
- 5. All FF&E which, by virtue of the terms and provisions of the applicable Lease, is not, or does not become, the property of the applicable Landlord upon the surrender of such Lease;
- 6. All FF&E listed in Schedule "J1" which are to be removed from any of the Premises, either prior to Closing or during the term of any Leaseback;
- 7. Any leasehold improvements, equipment, mechanical or electrical systems or fixtures which are not listed in Schedule "J2" which the Tenant desires to remove;
- 8. Any property belonging to the subtenants, franchisees or licensees of the Tenant or other occupants of the Premises;
- 9. All insurance policies of the Tenant; and
- 10. Any and all assets not located at a Location or any asset not used directly and exclusively at the Locations.

# SCHEDULE "E" INTENTIONALLY DELETED.

# SCHEDULE "F" SITE PLAN AGREEMENT

The Site Plan Agreement and related documents in respect of Location 3672, Conestoga Mall, comprised of:

Registered Site Plan Agreement dated August 21, 2012 between Ivanhoe Cambridge I Inc. and The Corporation of the City of Waterloo;

RSPA Compliance Agreement dated August 13, 2012 between Target Canada Co. and Ivanhoe Cambridge II Inc.;

Undertaking made on August 2, 2012 by Target Canada Corp. to The Corporation of the City of Waterloo.

# SCHEDULE "G" PRE-PAID LEASEBACK RENT

Project Bullseye

# Schedule of Gross Rents for PSA As of February 25, 2015

Schedule of Oxford Mall Gross Rents

		Les Galeries				
Rental Item	Upper Canada Mall	de la Capitale	Kingsway Mall	Square One	Hillcrest Mall	Total
Bace Rent	220,922	288,739	389,179	189,930	177,062	1,265,832
Operating Costs	50,479	26,564	168,791	43,332	96,630	355,796
Real Estate Tayes	100,475	244,467	103,551	152,829	104,705	706,027
Hillins	. •	49,131	70,171	142,323	9839	268,461
Other(1)	5.806	4,405	4,184	7,873	3,953	23,220
Total \$ (to tune 30) (Excl. Sales Tax)	374,682	613,305	735,876	536,287	359,187	2,619,337
Sales Tax	48,709	91,842	36,794	69,717	46,694	293,756
Total \$ (to lune 30) (Incl. Sales Tax)	423,391	705,147	772,669	606,005	405,881	2,913,093

(1) Includes Marketing and Promotion

Schedule of Ivanhoe Cambridge Mall Gross Rents

		Metropolis			Carrefour		
Rental Item	Oakridge	at Metrotown	Conestoga	Laurier	de L'Estrie	Mic Mac Mall	Total
Bace Rent		188,017	93,124	258,605	314,056	300,345	1,154,146
Observation Costs	•	139,328	27,562	75,710	70,414	41,589	354,603
Operating Costs Real Estate Taxes	,	56,674	150,828	314,322	165,881	105,562	793,267
	,	68,417	1,479	33,549	28,655	1	132,100
Other (1)	•	3.801	2,340	6,740	7,074	1,603	21,557
Total & (to line 30) (Excl Sales Tax)		456,237	275,333	688,925	586,080	449,099	2,455,673
Color Tox	,	22,812	35,793	103,166	87,765	67,365	316,902
Jacob Hav Total \$ (to June 30) (Incl. Sales Tax)	1	479,048	311,126	792,091	673,845	516,464	2,772,574
Total \$ (to June 30) (Incl. Sales Tax)	4	479,048	311,126	792,091	673,845		516,464

Notes: (1) Includes Marketing and Promotion

# SCHEDULE "H" INTENTIONALLY DELETED

# SCHEDULE "I" FORM OF LEASE AMENDMENT AND SURRENDER AGREEMENT

### LEASE AMENDMENT AND SURRENDER AGREEMENT

# [FOR PREMISES IN THE PROVINCE OF QUÉBEC: LEASE TERMINATION AGREEMENT]

THIS AGREEMENT is made as of the <b>Date</b> ").	day of	, 2015 (the "Closing
BETWEEN:		
TARGET CANADA CO.		
(the "Tenant")		
- and -		
•		
(the "Landlord")		

### **RECITALS:**

- A. Pursuant to a lease dated  $\bullet$ , as same is assigned, amended, restated, renewed or supplemented from time to time, including but not limited to those documents listed in Schedule "A" attached hereto (collectively, the "Lease"), the Landlord leased to the Tenant certain premises (the "Premises") at  $\bullet$  in the City of  $\bullet$ , in the Province of  $\bullet$  as more particularly described in the Lease (the "Location").
- B. [NTD: Pursuant to a [NTD: insert particulars of guaranty/indemnity], Target Corporation ("Target") provided the Landlord with a [NTD: guaranty/indemnity] of certain liabilities and obligations of the Tenant under the Lease (the "Guarantee").]
- C. The Tenant and certain of its Affiliates (collectively, the "Applicants") applied for and were granted protection by the Ontario Superior Court of Justice (Commercial List) (the "Court") under the *Companies' Creditors Arrangement Act*, R.S.C. 1985, c. C-36, as amended (the "CCAA"), and Alvarez & Marsal Canada Inc. (the "Monitor") was appointed the monitor of the Tenant, pursuant to an Order of the Court dated January 15, 2015, as amended and restated on February 11, 2015, and as further amended, restated and/or amended and restated from time to time (collectively, the "Initial Order").
- D. The Tenant, as tenant, and the Landlord and other entities, as Landlord Entities, entered into a lease transaction agreement dated ●, 2015 (the "Lease Transaction Agreement"), whereby, among other things, the Tenant agreed to surrender to the Landlord, and the Landlord agreed to accept a surrender of, the Lease and the Premises and the Real Property Interests relating thereto.
- E. The Lease Transaction Agreement was approved by the Court pursuant to the Order dated (the "Approval Order").

- F. The Landlord and the Tenant are entering into this Agreement to provide for the surrender of the Lease and the Premises and the Real Property Interests relating thereto in accordance with the Lease Transaction Agreement.
- G. Unless otherwise expressly provided for herein, all capitalized terms when used in this Agreement have the same meaning given to such terms in the Lease Transaction Agreement, or if no meaning is given in the Lease Transaction Agreement, in the Lease.

**THEREFORE**, in consideration of the premises and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties agree as follows:

# ARTICLE 1 TERMINATION AND SURRENDER

[Note: As contemplated in the Lease Transaction Agreement, the Landlord Entities and Tenant may enter into a separate agreement prior to this Agreement to cancel options to renew/extend.]

# 1.1 Amendment and Early Termination of Lease.

The Lease is hereby amended and the Landlord and the Tenant hereby agree that the Lease has expired and is terminated, as of 11:59 p.m. on the day immediately preceding the Closing Date (the "Termination Date"), and neither the Tenant [nor Target] nor the Landlord shall have any liabilities or obligations under the Lease [and/or Guaranty], financial or otherwise, as of and as from the Termination Date.

[For Premises situated in Quebec: The Landlord and the Tenant hereby agree that the Lease has expired and is terminated, as of 11:59 p.m. on the date immediately preceding the Closing Date (the "Termination Date") as if it was the last day of the term of the Lease and neither the Tenant [nor Target] nor the Landlord shall have any further liabilities or obligations under the Lease [and/or Guarantee], financial or otherwise, as of and as from the Termination Date.]

# 1.2 Surrender by Tenant.

The Tenant hereby surrenders to the Landlord, as of the Termination Date, and the Landlord hereby accepts such surrender from the Tenant, the Lease and the Premises demised by the Lease the Lease and the Real Property Interests relating thereto and all the Tenant's rights, title and interest thereunder, with the intent that the unexpired residue of the term of the Lease including, without limitation, any rights or options to renew or extend hereby merge and are extinguished in the reversion expectant thereon, on the terms and conditions set out in the Lease Transaction Agreement.

[Note: Section 1.2 to be deleted from Agreement for Premises situated in Québec.]

# 1.3 Adjustments.

All adjustments under the Lease shall be dealt with in accordance with the Lease Transaction Agreement.

Page 3

### 1.4 Release.

The parties acknowledge and agree that, pursuant to and on the terms and conditions set out in the Lease Transaction Agreement, the Landlord, the Tenant and Target have executed and delivered mutual releases in respect of the Real Property Interests at the Location, the Lease, the Premises and/or the properties of which the Premises form a part.

# 1.5 Paramountcy.

The rights and obligations of the parties respectively with respect to the Real Property Interests at the Location, the Lease, the Premises and/or the properties of which the Premises form a part shall be governed by the Lease Transaction Agreement. In the event of any conflict, inconsistency or ambiguity between the provisions of this Agreement and of the Lease Transaction Agreement, then the provisions of the Lease Transaction Agreement shall govern and be paramount, and any such provision in this Agreement shall be deemed to be amended to the extent necessary to eliminate any such conflict, inconsistency, ambiguity or difference.

# ARTICLE 2 GENERAL

## 2.1 Time of the Essence.

Time shall be of the essence of this Agreement.

### 2.2 Enurement.

This Agreement shall be binding upon and enure to the benefit of the parties and their respective successors and permitted assigns.

# 2.3 Entire Agreement.

This Agreement, the Lease Transaction Agreement, the Closing Documents and the Approval Order constitute the entire agreement between the parties with respect to the transactions contemplated in this Agreement and supersedes all prior agreements, understandings, negotiations and discussions, whether oral or written, of the parties with respect to the subject matter of this Agreement. There are no representations, warranties, covenants, conditions or other agreements, express or implied, collateral, statutory or otherwise, between the parties in connection with the subject matter of this Agreement, except as specifically set forth in this Agreement, or in the Lease Transaction Agreement, or in any of the Closing Documents or in the Approval Order. The parties have not relied and are not relying on any other information, discussion or understanding in entering into and completing the transactions contemplated by this Agreement.

### 2.4 Waiver.

(a) No waiver of any of the provisions of this Agreement shall be deemed to constitute a waiver of any other provision (whether or not similar), nor shall such waiver be binding unless executed in writing by the party to be bound by the

waiver.

(b) No failure on the part of the Tenant or the Landlord to exercise, and no delay in exercising any right under this Agreement shall operate as a waiver of such right; nor shall any single or partial exercise of any such right preclude any other or further exercise of such right or the exercise of any other right.

#### 2.5 Further Assurances.

Each of the parties covenants and agrees to do such things, to attend such meetings and to execute such further conveyances, transfers, documents and assurances as may be deemed necessary or advisable from time to time in order to effect the surrender of the Lease and the Premises and carry out the terms and conditions of this Agreement in accordance with their true intent.

#### 2.6 Severability.

If any provision of this Agreement shall be determined to be illegal, invalid or unenforceable, that provision shall be severed from this Agreement and the remaining provisions shall continue in full force and effect.

#### 2.7 Governing Law.

Except as otherwise provided for in the Lease, this Agreement shall be governed and construed in accordance with the laws applicable to the Lease and the federal laws of Canada applicable in the province in which the Premises are located.

#### 2.8 CCAA Proceedings.

Each party to this Agreement submits to the exclusive jurisdiction of the Court in any action, application, reference or other proceeding arising out of or related to this Agreement or the Lease or the Guaranty and agrees that all claims in respect of any such actions, application, reference or other proceeding shall be heard and determined in the CCAA Proceedings before the Court.

#### 2.9 French Language.

The parties hereto have requested that this Agreement be drafted in English only. / Les parties aux présentes ont demandé à ce que la présente convention soit redigée en anglais seulement.

#### 2.10 References.

Where in this Agreement reference is made to an article or section, the reference is to an article or section in this Agreement unless the context indicates the reference is to some other agreement. The terms "this Agreement", "hereof", "hereunder" and similar expressions refer to this Agreement and not to any particular Article, Section or other portion hereof and include any agreement supplemental hereto.

#### 2.11 Number and Gender.

Unless the context requires otherwise, words importing the singular include the plural and vice versa and words importing gender include all genders.

#### 2.12 Use of the word "including" and "or" etc.

The word "includes" or "including" shall mean "includes without limitation" or "including without limitation", respectively. The word "or" is not exclusive.

#### 2.13 Notice.

Notwithstanding anything to the contrary in the Lease, any notice, consent or approval required or permitted to be given in connection with this Agreement or the Lease shall be in writing and shall be sufficiently given if delivered in accordance with the Lease Transaction Agreement.

#### 2.14 Counterparts and Execution.

This Agreement and any amendments thereto (and any other agreements, notices or documents contemplated thereby) may be executed and delivered by facsimile or electronic transmission (including electronic transmission via the Internet) and in any number of counterparts and all such facsimile or electronic copies and counterparts shall be deemed to be an original hereof and for all purposes constitute one agreement, be binding on the parties hereto, provided each party hereto has executed and delivered at least one counterpart, and each may be relied upon by each party hereto as such for any and all purposes.

[Signature pages follow.]

# Lease Amendment and Surrender Agreement [Note: For Premises in Québec: Lease Termination Agreement] [Location]

Schedules

IN WITNESS OF WHICH the Tenant has duly executed this Agreement.

T	٨	p	GET	CA	NA	D	A	CO	).

By:			 	 
	Name:			
	Title:			

# Lease Amendment and Surrender Agreement [Note: For Premises in Québec: Lease Termination Agreement] [Location] Signature Page

IN WITNESS OF WHICH the Landlord has duly ex	executed this Agreement.
----------------------------------------------	--------------------------

[ LANDLORD]
-------------

ву:		
	Name:	
	Title:	
By:		
	Name:	
	Title:	

[Location] Schedules

## SCHEDULE "A" – LEASE

[NTD: list lease documents]

# SCHEDULE "J1" ITEMS WHICH SHALL BE REMOVED FROM THE PREMISES

All shelving units back of house & in the store

Coolers

Checkout counters

Non affix kiosks

Branding Signage (Interior & Exterior)

All office equipment inventory

Shopping carts & corrals (even if located outside of the Premises)

Refrigeration units

Walk-in cooler

Security systems including all video and security cameras and equipment (provided Tenant shall not remove cabling in place) and entrance sensors used for merchandise security

All computers and related systems and information storage media

All point-of-sales systems and all appurtenances thereto

# SCHEDULE "J2" ITEMS WHICH CANNOT BE REMOVED FROM THE PREMISES

Elevators with support devices

Escalators with support devices

Anything HVAC related

Ceiling lights and tiles

Flooring

All washrooms (sinks, toilets, urinals & stall partition) and fixtures

All doors (hardware) / loading exterior and interior

All roofing systems

Fire safety systems & equipment

Electrical equipment

Trash Compactors or Bailers

Generators

Subject to Schedules "E" and "J1", property and assets owned by the Landlord Entities

### SCHEDULE "K" FORM OF LEASEBACK

#### SHORT TERM LEASE

2015 (41 - "TEEC - 41-12 Do4")

THIS LEASE	is made as of the	day of	, 2015 (the	Effective Date ).
BETWEE	N:			
	•			
	(the "Landlord")			
	- and -			
	TARGET CANADA	CO.		
	(the "Tenant")			

#### **RECITALS:**

- A. Pursuant to a lease dated , as same is assigned, amended, restated, renewed or supplemented from time to time, including but not limited to those documents listed in Schedule "A" attached hereto (collectively, the "Original Lease"), the Landlord leased to the Tenant certain premises described in the Original Lease (the "Premises") at in the City of ●, in the Province of as more particularly described in the Original Lease (the "Location").
- B. The Tenant and certain of its Affiliates (collectively, the "Applicants") applied for and were granted protection by the Ontario Superior Court of Justice (Commercial List) (the "Court") under the Companies' Creditors Arrangement Act, R.S.C. 1985, c. C-36, as amended (the "CCAA"), and Alvarez & Marsal Canada Inc. (the "Monitor") was appointed the monitor of the Tenant, pursuant to an Order of the Court dated January 15, 2015, as amended and restated on February 11, 2015, and as further amended, restated and/or amended and restated from time to time (collectively, the "Initial Order").
- C. The Tenant, as tenant, and the Landlord and other entities, as Landlord Entities, entered into a lease transaction agreement dated February ●, 2015 (said agreement as amended from time to time being collectively, the "Lease Transaction Agreement"), whereby, among other things, the Tenant agreed to surrender the Original Lease and the Premises and the Real Property Interests relating thereto to the Landlord and the Landlord agreed to accept such surrender and to grant a Leaseback of the Premises to the Tenant and the Tenant agreed to continue to lease the Premises from the Landlord in accordance with the terms and conditions of this Lease.
- D. The Lease Transaction Agreement was approved by the Court pursuant to the Order dated (the "Approval Order") and the Closing of the Transaction contemplated under the Lease Transaction Agreement took place on the Effective Date.
- E. In connection with the Closing of the Transaction contemplated under the Lease Transaction Agreement, the Original Lease was surrendered as of 11:59 p.m. on the day immediately preceding the Effective Date.
- F. The Landlord and the Tenant exchanged mutual releases in respect of the Original Lease,

the Premises and the Real Property Interests relating thereto as same was surrendered and the properties of which the Premises form a part.

- G. The Landlord and the Tenant are entering into this Lease to provide for the Leaseback of the Premises in accordance with the Lease Transaction Agreement.
- H. Unless otherwise expressly provided for herein, all capitalized terms when used in this Lease have the same meaning given to such terms in the Lease Transaction Agreement, or if no meaning is given in the Lease Transaction Agreement, in the Original Lease.

**THEREFORE**, in consideration of the premises and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties agree as follows:

#### ARTICLE 1 LEASE AND TERM

#### 1.1 Lease of Premises for the Term.

- (a) The Landlord hereby leases the Premises to Tenant, and Tenant hereby leases the Premises from Landlord, for the Term (as defined below), together with all of the same right, title and interest as the Tenant had, including its leasehold interest at common law and its rights as tenant at civil law, in and to the Premises on the terms and conditions set out in Section 2.1 and elsewhere in this Lease, including, if any, the benefit of all easements, restrictive covenants, access rights, licences to use any common areas or rooftop areas of the buildings or shopping centres of which the Premises form part, parking rights and signage rights, except as otherwise expressly provided for in Article 2. However, for greater certainty, nothing herein shall be construed or interpreted to revive or constitute novation of the Original Lease which the parties acknowledge was surrendered.
- (b) The "Term" shall mean the period commencing on the Effective Date, and expiring on June 30, 2015, as such date may be advanced pursuant to an Early Vacancy Notice delivered in accordance with Section 2.7 (the "Expiration Date"). For greater certainty, the Term shall commence immediately as at and from the date and at the time the Original Lease was surrendered, with no gap between the expiry or termination of the term of the Original Lease and the commencement of the Term, and shall expire and be terminated, as of 11:59 p.m. on the Expiration Date.

#### 1.2 Quiet Enjoyment

Subject to Tenant complying with all terms of this Lease, the Landlord covenants with the Tenant for quiet enjoyment.

# ARTICLE 2 TERMS OF LEASE

#### 2.1 Original Lease Terms

The Landlord leases the Premises to the Tenant, and the Tenant leases the Premises from the

Landlord, on the same terms and conditions as in the Original Lease as applicable under the CCAA Proceedings for the Premises immediately prior to the surrender of the Original Lease (which terms and conditions are incorporated herein by reference), including all Court Orders made in the CCAA Proceedings immediately prior to the surrender of the Original Lease, including, the Initial Order and the Sale Order, except as set out in this Article 2.

#### 2.2 Gross Rent

The Landlord acknowledges and agrees that the Tenant has pre-paid to the Monitor, as escrow agent, in accordance with Section 8.4 of the Lease Transaction Agreement, the entire Pre-Paid Leaseback Rent. The Landlord further acknowledges and agrees that the portion of the Pre-Paid Leaseback Rent relating to this Lease represents a fixed and non-adjustable gross rent on account of all basic rent, minimum and additional rent for the Term without adjustment plus applicable GST/HST, QST and / or BCPST, and that the Monitor shall make regular gross rent payments (together with applicable GST/HST, QST and/or BCPST) to the Landlord payable under this Lease from the Pre-Paid Leaseback Rent as per the Initial Order during the Term on the first and fifteenth day of each calendar month during the Term. Any part of the portion of the Pre-Paid Leaseback Rent relating to this Lease which is not paid or not required to be paid under this Lease shall be refunded by the Monitor to the Tenant on the Business Day following the Expiration Date. For greater certainty, there shall be no adjustment or readjustment before, on or after the Expiration Date.

#### 2.3 Rights and Obligations, Termination

The Tenant shall have no less rights, and subject to this Article 2, no less obligations, under this Lease to use and occupy the Premises than in the Original Lease as applicable under the CCAA Proceedings for the Premises immediately prior to the surrender of the Original Lease. Notwithstanding anything to the contrary in this Lease but subject to the Initial Order, the Landlord shall not be entitled to terminate this Lease prior to the Expiration Date by any reason other than a default or event of default during the Term beyond the applicable cure period giving rise to a right of termination pursuant to the terms and conditions of this Lease.

#### 2.4 Use

- (a) The Tenant may occupy the Premises for the purpose of selling or removing, either directly or through its agents, any and all chattels, personal property, Inventory, FF&E and any other Excluded Assets or items listed on Schedule "J1" of the Lease Transaction Agreement, but excluding any items listed in Schedule "J2", from the Premises in accordance with the Initial Order and the Sale Order and the provisions of this Article 2.
- (b) The Tenant shall have no obligation to stock, open, staff, operate or continuously operate any business at the Premises, and may, in its sole discretion, by itself or through its Agent continue the Sale as defined in and as approved or permitted by the Initial Order and the Sale Order, provided that the Tenant shall continue to maintain its existing insurance coverage (including any self-insurance, if existing immediately prior to the surrender of the Original Lease) on the Premises until the Expiration Date. As part of such winding-down, Tenant shall not sell or remove or permit the sale or removal from the Premises of any of the items listed in

Schedule "J2" of the Lease Transaction Agreement.

#### 2.5 Transfers

- (a) The Tenant shall not have any rights to assign its rights, title, interest or obligations pursuant to this Lease nor to grant any sublease or licence in respect of the Premises unless permitted by the Initial Order, the Sale Order, the Approval Order or any other Order of the Court made on notice to the Landlord.
- (b) However, the Tenant shall be permitted, without any consent from the Landlord: (A) to have existing third party pharmacists occupy a portion of the Premises and remain in such Leaseback Premises until no later than March 30, 2015, all in accordance with the accommodation confirmed in the Endorsement of R.S.J Morawetz dated February 18, 2015; and (B) to conduct through its Agent the Sale as defined in the Sale Order on the Premises in accordance with such Order and the Initial Order.
- (c) The Landlord shall not assign its right, obligations or interests in this Lease during the Term, except if such assignment is to an Affiliate of the Landlord, in which case the Landlord shall cause any such Affiliate assignee to be bound by the covenants, obligations and liabilities of the Landlord under this Lease, and to comply with the terms of this Lease for the Premises until the Expiration Date. However, the foregoing shall not prohibit the Landlord from assigning its interest in this Lease and/or the Premises as security to a *bona fide* lender which is taking security.

#### 2.6 Other Terms of Lease

- (a) The Tenant shall accept the Premises on an "as is, where is" basis.
- (b) To the extent that the Tenant is required to do so under the Original Lease, the Tenant shall be responsible for maintaining and paying for utility services (including electricity) to the Premises during the Term. To the extent that the Tenant is required to do so under the Original Lease, the Landlord shall continue to provide utilities to the Premises during the Term.
- (c) The Tenant shall not provide any acknowledgement, status certificate or estoppel in respect of this Lease.
- (d) The amendments to the Original Lease as set out in Section 1.2(a)(ii) of the Lease Transaction Agreement, if any, shall not apply to this Lease.
- (e) The Guaranty, if any, shall not apply to this Lease.
- (f) The Tenant shall not register any notice of this Lease in any land registry office or similar registers. If Tenant registers such a notice in breach of the foregoing, Tenant shall have same discharged at Tenant's sole cost and expense, failing which Landlord may have same discharged at Tenant's cost and expense.
- (g) The Landlord shall have no obligation under this Lease to pay any allowance or

inducement to Tenant, to carry out any work or improvements in the Premises (except for any maintenance, repair and replacement which is the responsibility of the Landlord under this Lease) or to pay any commissions.

(h) The Tenant shall deliver to the Landlord Entity certificates of insurance attesting the existence of the insurance coverage required pursuant to the terms and conditions of this Lease or shall confirm that the existing insurance coverage pursuant to the Leases remains in full force and effect.

### 2.7 Tenant's Right to Terminate Early

(a) Notwithstanding the duration of the Term, the Tenant may, without penalty or being in default, provide notice (an "Early Vacancy Notice") to the Landlord on or before the date indicated below to advance the Expiration Date to the corresponding date indicated below in respect of each Early Vacancy Notice date:

If Early Vacancy Notice is given on or before:	Expiration Date is advanced to:
April 15, 2015	April 30, 2015
April 30, 2015	May 15, 2015
May 15, 2015	May 30, 2015
May 30, 2015	June 15, 2015

If the Tenant has not given an Early Vacancy Notice in respect of the Premises on or before May 30, 2015, then the Expiration Date shall be June 30, 2015. For the avoidance of doubt, Tenant may only send an Early Vacancy Notice in respect of all of (and not part of) the Premises.

(b) On the Expiration Date (as such Expiration Date may be advanced pursuant to an Early Vacancy Notice), the Tenant shall vacate the entire Premises, this Lease shall be at an end and all of the obligations of the Tenant, including to maintain insurance or pay rent for the Premises, shall terminate.

#### 2.8 End of Term

(a) On or before the expiration or earlier termination of the Term, the Tenant shall vacate the Premises, remove any chattels, personal property, Inventory, FF&E and any other Excluded Assets in accordance with the Initial Order and the Sale Order; for greater certainty, the Tenant or the Agent shall remove each and every item of the type listed in Schedule "J1" of the Lease Transaction Agreement and shall not remove any of the items of the type listed in Schedule "J2" of the Lease Transaction Agreement. If the Tenant fails to remove any item of the type listed in Schedule "J1" of the Lease Transaction Agreement, then the Landlord may, at its election, either exercise any and all remedies to cause Tenant to remove same,

or obtain possession and ownership of same without notice or compensation to Tenant.

- (b) On or before the expiration or earlier termination of the Term, the Tenant shall leave the Premises in a broom-swept condition as required by the provisions of the Initial Order and the Sale Order, and the Landlord shall accept the Premises in such condition, and notwithstanding anything to the contrary in this Lease, the Tenant shall not be responsible for making any repairs, replacements, renovations, alterations, improvements or upgrades in or to the Premises except as provided for in the Initial Order or the Sale Order and other than to leave the Premises in a broom-swept condition and in the condition required by the provisions of the Initial Order and the Sale Order.
- (c) On or before the Expiration Date, the Tenant shall terminate all its contracts and agreements for the supply of any utilities (including electricity, gas, water, fuel, telephone service, internet services, security and surveillance services or otherwise) to the Premises.

#### 2.9 Releases

Within five (5) Business Days of the expiration or earlier termination of this Lease, the Landlord shall execute and deliver to the Tenant a full and final release in respect of any Claims relating to or arising from this Lease in a form substantially similar to the Release of Landlord Claims attached as Schedule "N" to the Lease Transaction Agreement, and the Tenant shall execute and deliver to the Landlord a release in respect of any Claims relating to or arising from this Lease in a form substantially similar to the Release of Tenant Claims attached as Schedule "O" to the Lease Transaction Agreement, which releases may be subject to any Claims identified therein relating to or arising from this Lease notice of which has been delivered to the other party prior to the delivery of such releases. This obligation in this Section 2.9 to execute and deliver such releases and the releases shall not merge but shall survive the expiry or termination of this Lease; and

#### 2.10 Paramountcy

In the event of any conflict, inconsistency or ambiguity between the provisions of this Lease and of the Lease Transaction Agreement, then the provisions of the Lease Transaction Agreement shall govern and be paramount, and any such provision in this Lease shall be deemed to be amended to the extent necessary to eliminate any such conflict, inconsistency, ambiguity or difference.

# ARTICLE 3 LANDLORD REPRESENTATIONS & WARRANTIES

#### 3.1 Landlord Representations

The Landlord represents and warrants as to all of the following matters, and acknowledges and confirms that the Tenant is relying on such representations and warranties in connection with the entering into of this Lease:

- (a) The Landlord has been duly incorporated or constituted and is validly subsisting under the Laws of the jurisdiction of its incorporation or constitution, and has all requisite corporate capacity, power and authority to carry on its business as now conducted by it and to own its properties and assets and is qualified to carry on business under the Laws of the jurisdictions where it carries on a material portion of its business.
- (b) The execution, delivery and performance by the Landlord of this Lease:
  - (i) has been duly authorized by all necessary corporate action on the part of the Landlord; and
  - (ii) does not (or would not with the giving of notice, the lapse of time or the happening of any other event or condition) require any consent or approval under, result in a breach or a violation of, or conflict with, any of the terms or provisions of its respective constating documents or by-laws.
- (c) This Lease has been duly executed and delivered by the Landlord and constitutes legal, valid and binding obligations of the Landlord, enforceable against it in accordance with its terms subject only to any limitation under applicable Laws relating to (i) bankruptcy, winding-up, insolvency, arrangement and other similar Laws of general application affecting the enforcement of creditors' rights, and (ii) the discretion that a court may exercise in the granting of equitable remedies such as specific performance and injunction.

The representations and warranties in this Section 3.1 shall survive the execution of this Lease.

# ARTICLE 4 GENERAL

#### 4.1 Time of the Essence.

Time shall be of the essence of this Lease.

#### 4.2 Enurement.

This Lease shall be binding upon and enure to the benefit of the parties and their respective successors and permitted assigns.

#### 4.3 Entire Agreement.

This Lease, the Lease Transaction Agreement, the Initial Order, the Sale Order and the Approval Order constitute the entire agreement between the parties with respect to the transactions contemplated in this Lease and together these documents supersede all prior agreements, understandings, negotiations and discussions, whether oral or written, of the parties with respect to the subject matter of this Lease. There are no representations, warranties, covenants, conditions or other agreements, express or implied, collateral, statutory or otherwise, between the parties in connection with the subject matter of this Lease, except as specifically set forth in this Lease, the Lease Transaction Agreement, the Initial Order, the Sale Order, or the Approval Order. The parties have not relied and are not relying on any other information, discussion or

understanding in entering into and completing the transactions contemplated by this Lease.

#### 4.4 Waiver.

- (a) No waiver of any of the provisions of this Lease shall be deemed to constitute a waiver of any other provision (whether or not similar), nor shall such waiver be binding unless executed in writing by the party to be bound by the waiver.
- (b) No failure on the part of the Tenant or the Landlord to exercise, and no delay in exercising any right under this Lease shall operate as a waiver of such right; nor shall any single or partial exercise of any such right preclude any other or further exercise of such right or the exercise of any other right.

#### 4.5 Further Assurances.

Each of the parties covenants and agrees to do such things, to attend such meetings and to execute such further conveyances, transfers, documents and assurances as may be deemed necessary or advisable from time to time in order to effect the Leaseback of the Premises and carry out the terms and conditions of this Lease in accordance with their true intent.

#### 4.6 Severability.

If any provision of this Lease shall be determined to be illegal, invalid or unenforceable, that provision shall be severed from this Lease and the remaining provisions shall continue in full force and effect.

#### 4.7 Governing Law.

Except as otherwise provided for in the Lease, this Lease shall be governed by the laws applicable to the Original Lease.

#### 4.8 CCAA Proceedings

Each party to this Lease submits to the exclusive jurisdiction of the Court in any action, application, reference or other proceeding arising out of or related to this Lease and agrees that all claims in respect of any such actions, application, reference or other proceeding shall be heard and determined in the CCAA Proceedings before the Court.

#### 4.9 French Language.

The parties hereto have requested that this Lease be drafted in English only. Les parties aux présentes ont demandé à ce que la présente convention soit redigée en anglais seulement.

#### 4.10 References.

Where in this Lease reference is made to an article or section, the reference is to an article or section in this Lease unless the context indicates the reference is to some other agreement. The terms "this Lease", "hereof", "hereunder" and similar expressions refer to this Lease and not to any particular Article, Section or other portion hereof and include any agreement supplemental hereto.

#### 4.11 Number and Gender.

Unless the context requires otherwise, words importing the singular include the plural and vice versa and words importing gender include all genders.

#### 4.12 Use of the word "including" and "or" etc.

The word "includes" or "including" shall mean "includes without limitation" or "including without limitation", respectively. The word "or" is not exclusive.

#### 4.13 Notice.

Notwithstanding anything to the contrary in this Lease, any notice, consent or approval required or permitted to be given in connection with this Lease shall be in writing and shall be sufficiently given if delivered in accordance with the Lease Transaction Agreement.

#### 4.14 Counterparts and Execution.

This Lease and any amendments thereto (and any other agreements, notices or documents contemplated thereby) may be executed and delivered by facsimile or electronic transmission (including electronic transmission via the Internet) and in any number of counterparts and all such facsimile or electronic copies and counterparts shall be deemed to be an original hereof and for all purposes constitute one agreement, be binding on the parties hereto, provided each party hereto has executed and delivered at least one counterpart, and each may be relied upon by each party hereto as such for any and all purposes.

[Remainder of Page Intentionally Left Blank]

IN WITNESS OF WHICH the Landlord has duly executed this Lease.

## [● LANDLORD]

By:			
_	Name:	 	 
	Title:		
By:			
	Name:	 	 
	Title:		

IN WITNESS OF WHICH the Tenant has duly executed this Lease.

## TARGET CANADA CO.

By:		 	 	 
-	Name:			
	Title:			

## SCHEDULE "A" – ORIGINAL LEASE

[NTD: list lease documents]

### SCHEDULE "L" FORM OF APPROVAL ORDER

# ONTARIO SUPERIOR COURT OF JUSTICE COMMERCIAL LIST

THE HONOURABLE	)	THURSDAY, THE $5^{TH}$
REGIONAL SENIOR JUSTICE	)	DAY OF MARCH, 2015
MORAWETZ	)	

IN THE MATTER OF THE COMPANIES' CREDITORS ARRANGEMENT ACT, R.S.C. 1985, c. C-36, AS AMENDED

AND IN THE MATTER OF A PLAN OF COMPROMISE OR ARRANGEMENT OF TARGET CANADA CO., TARGET CANADA HEALTH CO., TARGET CANADA MOBILE GP CO., TARGET CANADA PHARMACY (BC) CORP., TARGET (ONTARIO) CORP., TARGET PHARMACY CANADA **CANADA TARGET** CORP.. PHARMACY CANADA **CANADA TARGET** CORP., and **PHARMACY** (SK) PROPERTY LLC (collectively the "Applicants")

## APPROVAL AND VESTING ORDER – LEASE TRANSACTION AGREEMENT

THIS MOTION, made by the Applicants, pursuant to the *Companies' Creditors Arrangement Act*, R.S.C. 1985, c. c-36, as amended (the "CCAA") for an order, *inter alia*, approving: the lease transaction (the "Transaction") contemplated by a Lease Transaction Agreement (the "Lease Transaction Agreement") among Target Canada Co. ("TCC"), as Tenant, and Hillcrest Holdings Inc., Montez Hillcrest Inc., Oxford Properties Retail Holdings II Inc., CPPIB Upper Canada Mall Inc., OMERS Realty Management Corporation, Square One Property Corporation, Kingsway Garden Holdings Inc., Les Galeries de la Capitale Holdings Inc., Ivanhoe Cambridge II Inc., Carrefour de L'estrie Holdings, Inc., Place Laurier Holdings Inc., Mic Mac Mall Limited Partnership and Oakridge Centre Vancouver Holdings, Inc. (collectively, the "Landlord Entities") dated •, 2015 and certain related relief, was heard this day at 330 University Avenue, Toronto, Ontario.

ON READING the Notice of Motion of the Applicants, the Affidavit of Mark Wong sworn on ●, 2015 including the exhibits thereto (the "Wong Affidavit"), the Affidavit of Timothy Pohl sworn on ●, 2015 and the ● Report (the "Monitor's ● Report") of Alvarez & Marsal Canada Inc., in its capacity as Monitor (the "Monitor"), filed, and on hearing the submissions of respective counsel for the Applicants and the Partnerships listed on Schedule "A" hereto, the Monitor, Target Corporation, the Landlord Entities, and such other counsel as were present, no one else appearing although duly served as appears from the Affidavit of Service of [Robert Carson] sworn ●, 2015, filed:

#### SERVICE AND DEFINITIONS

- 1. THIS COURT ORDERS that the time for service of the Notice of Motion and the Motion Record herein is hereby abridged and validated so that this Motion is properly returnable today and hereby dispenses with further service thereof.
- 2. THIS COURT ORDERS that any capitalized term used and not defined herein shall have the meaning ascribed thereto in the Amended and Restated Initial Order in these proceedings dated January 15, 2015 (the "Initial Order"), or in the Lease Transaction Agreement, as applicable.

#### APPROVAL OF THE LEASE TRANSACTION AGREEMENT

3. THIS COURT ORDERS AND DECLARES that the Transaction is hereby approved and ratified and that the execution of the Lease Transaction Agreement by TCC is hereby approved and ratified with such minor amendments as TCC (with the consent of the Monitor) and the Landlord Entities may agree to in writing. TCC is hereby authorized and directed to take such additional steps and execute such additional documents as may be necessary or desirable for the completion of the Transaction, including the surrender by TCC of its right, title and interest in and to the Real Property Interests, the Premises and the Leases (collectively, the "Surrendered Assets") to the applicable Landlord Entities and the Monitor shall be authorized to take such additional steps in furtherance of its responsibilities under the Lease Transaction Agreement. The legal descriptions and applicable land registry offices with respect to the Surrendered Assets are as set out on Schedule "C" hereto.

- 4. THIS COURT ORDERS AND DECLARES that upon the delivery of a Monitor's certificate to the Landlord Entities substantially in the form attached as Schedule "B" hereto (the "Monitor's Certificate"), all of TCC's right, title and interest in and to the Surrendered Assets and the right, title and interest, if any, of any predecessor in interest of TCC in and to the Surrendered Assets, to the extent same was assigned or otherwise transferred to TCC, shall be surrendered to the Landlord Entities free and clear of and from any and all security interests (whether contractual, statutory, or otherwise), hypothecs, mortgages, trusts or deemed trusts (whether contractual, statutory, or otherwise), liens, executions, leases, notices of lease, subleases, licenses, levies, restrictions, rights of retention, judgments, notices of sale, contractual rights, options, liabilities (direct, indirect, absolute or contingent), obligations, charges, or other financial or monetary claims, whether or not they have attached or been perfected, registered or filed and whether secured, unsecured or otherwise in respect of the Surrendered Assets (collectively, the "Claims"), including, without limiting the generality of the foregoing:
  - (a) the Administration Charge, the KERP Charge, the Directors' Charge, the Financial Advisor Subordinated Charge, the DIP Lender's Charge, the Agent's Charge and Security Interest (collectively, the "CCAA Charges");
  - (b) all charges, security interests or claims evidenced by registrations pursuant to the Personal Property Security Act (Ontario), the Civil Code of Québec, the Alberta Personal Property Security Act, the British Columbia Personal Property Security Act, the Nova Scotia Personal Property Security Act or any other personal property registry system; and
  - (c) those Claims listed on Schedule "C" hereto;

(all of which are collectively referred to as the "Encumbrances", which term shall not include the Permitted Encumbrances listed on Schedule "D" hereto)

and, for greater certainty, this Court orders that all of the Claims and Encumbrances affecting or relating to the Surrendered Assets are hereby expunged and discharged as against the Surrendered Assets and the real property described in Schedule "C".

5. THIS COURT ORDERS that upon the registration in the applicable land registry office of a certified copy of this Order in the manner prescribed by the applicable land registry

office, the applicable Land Registrar is hereby directed to specifically discharge, delete and expunge from title to the applicable real property described in Schedule "C" all of the Encumbrances listed in Schedule "C" hereto.

- 6. THIS COURT ORDERS that for the purposes of determining the nature and priority of Claims, the net proceeds received on the Closing of the Transaction shall stand in the place and stead of the Surrendered Assets and that from and after the delivery of the Monitor's Certificate all Claims and Encumbrances shall attach to the net proceeds therefrom with the same priority as they had with respect to the Surrendered Assets immediately prior to the Closing of the Transaction, as if the Transaction had not been completed.
- 7. THIS COURT ORDERS AND DIRECTS the Monitor to file with the Court a copy of the Monitor's Certificate, forthwith after delivery thereof.
- 8. THIS COURT ORDERS that nothing herein or in the Lease Transaction Agreement affects:
  - (a) the accommodation provided by TCC to the pharmacists as confirmed by the Endorsement of R.S.J. Morawetz dated February 18, 2015 for the period that shall end on or before March 30, 2015;
  - (b) the rights and obligations of TCC and the Agent under the Agency Agreement dated January 29, 2015 (the "Agency Agreement"); and
  - (c) the terms of the Approval Order Agency Agreement granted, February 4, 2015 (the "Agency Agreement Approval Order") including the Sales Guidelines attached as Schedule "B" thereto (and for greater certainty, any reference to Leases in the Agency Agreement and the Sales Guidelines shall include the corresponding Leasebacks).
- 9. THIS COURT ORDERS that TCC, as Tenant, is hereby authorized to enter into the Leasebacks and to continue to occupy each of the Leaseback Premises in accordance with and on the terms and conditions of the Lease Transaction Agreement, and the Landlord Entities shall allow TCC to continue to occupy the Premises in accordance with and subject to the Lease

Transaction Agreement and the Leasebacks to permit the completion of the Sale (as defined in the Agency Agreement Approval Order).

#### **SEALING**

- 10. THIS COURT ORDERS that:
  - (a) Confidential Appendix A to the Monitor's Report, being the un-redacted version of the Lease Transaction Agreement ("Confidential Appendix A") and Confidential Appendix B to the Monitor's Report, being the Northwest analysis in connection thereto ("Confidential Appendix B") shall be sealed, kept confidential and not form part of the public record, but rather shall be placed, separate and apart from all other contents of the Court file, in a sealed envelope attached to a notice that sets out the title of these proceedings and a statement that the contents are subject to a sealing order; and
  - (b) Confidential Appendix A shall only be opened and made public upon the Closing of the Transaction and Confidential Appendix B shall remain under seal until further Order of the Court.

#### GENERAL PROVISIONS

- 11. THIS COURT ORDERS that, notwithstanding:
  - (a) the pendency of these proceedings;
  - (b) any applications for a bankruptcy order now or hereafter issued pursuant to the Bankruptcy and Insolvency Act (Canada) in respect of TCC and any bankruptcy order issued pursuant to any such applications; and
  - (c) any assignment in bankruptcy made in respect of TCC;

the surrender of the Surrendered Assets to the Landlord Entities pursuant to this Order shall be binding on any trustee in bankruptcy that may be appointed in respect of TCC and shall not be void or voidable by creditors of TCC, nor shall it constitute nor be deemed to be a fraudulent preference, assignment, fraudulent conveyance, transfer at undervalue, or other reviewable transaction under the *Bankruptcy and Insolvency Act* (Canada) or any other applicable federal or provincial legislation, nor shall it constitute oppressive or unfairly prejudicial conduct pursuant to any applicable federal or provincial legislation.

- 12. THIS COURT ORDERS AND DECLARES that, pursuant to section 142 of the Courts of Justice Act (Ontario), no person shall be liable for any act done in good faith in accordance with any Order issued in this proceeding. Any person who takes any action whatsoever in reliance on this Order prior to the commencement of any appeal hereof or the expiry of any appeal period, including completion of the Transaction, shall not be prejudiced in any manner by any such subsequent appeal.
- 13. THIS COURT ORDERS AND DECLARES that the Transaction is exempt from the application of the *Bulk Sales Act* (Ontario) and any equivalent legislation in any other jurisdiction in which all or any part of the Surrendered Assets are located.
- 14. THIS COURT HEREBY REQUESTS the aid and recognition of any court, tribunal, regulatory or administrative bodies, having jurisdiction in Canada or in the United States of America, to give effect to this Order and to assist TCC, the Monitor and their respective agents in carrying out the terms of this Order. All courts, tribunals, regulatory and administrative bodies are hereby respectfully requested to make such orders and to provide such assistance to TCC and to the Monitor, as an officer of this Court, as may be necessary or desirable to give effect to this Order, to grant representative status to the Monitor in any foreign proceeding, or to assist TCC and the Monitor and their respective agents in carrying out the terms of this Order.

# SCHEDULE "A" PARTNERSHIPS

Target Canada Pharmacy Franchising LP

Target Canada Mobile LP

Target Canada Property LP

#### **SCHEDULE "B"**

Court File No. CV-15-10832-00CL

# ONTARIO SUPERIOR COURT OF JUSTICE COMMERCIAL LIST

THE HONOURABLE	)	THURSDAY, THE $5^{TH}$
REGIONAL SENIOR JUSTICE	)	DAY OF MARCH, 2015
MORAWETZ	)	

IN THE MATTER OF THE *COMPANIES' CREDITORS* ARRANGEMENT ACT, R.S.C. 1985, c. C-36, AS AMENDED

AND IN THE MATTER OF A PLAN OF COMPROMISE OR ARRANGEMENT OF TARGET CANADA CO., TARGET CANADA HEALTH CO., TARGET CANADA MOBILE GP CO., TARGET CANADA PHARMACY (BC) CORP., TARGET CANADA **PHARMACY** (ONTARIO) CORP., **TARGET** CANADA **PHARMACY** CORP., **TARGET CANADA** CORP., **PHARMACY** (SK) and TARGET **CANADA** PROPERTY LLC (collectively the "Applicants")

#### **MONITOR'S CERTIFICATE**

#### **RECITALS**

A. All undefined terms in this Monitor's Certificate have the meanings ascribed to them in the Order of the Court dated March 5, 2015 (the "Approval Order") approving the Lease Transaction Agreement entered into among Target Canada Co. ("TCC") and Hillcrest Holdings Inc., Montez Hillcrest Inc., Oxford Properties Retail Holdings II Inc., CPPIB Upper Canada Mall Inc., OMERS Realty Management Corporation, Square One Property Corporation, Kingsway Garden Holdings Inc., Les Galeries de la Capitale Holdings Inc., Ivanhoe Cambridge II Inc., Carrefour de L'estrie Holdings, Inc., Place Laurier Holdings Inc., Mic Mac Mall Limited Partnership and Oakridge Centre Vancouver Holdings, Inc. (collectively, the "Landlord Entities") dated ●, 2015 (the "Lease Transaction Agreement"), a copy of which is attached as Exhibit ● to the Affidavit of [Mark Wong] dated ●, 2015.

Draft

B. Pursuant to the Approval Order, the Court approved the Lease Transaction Agreement and provided for the surrender to the Landlord Entities of TCC's right, title and interest in and to the Surrendered Assets, which surrender is to be effective with respect to the Surrendered Assets upon the delivery by the Monitor to the Landlord Entities and TCC of a certificate confirming (i) the conditions to Closing as set out in sections 7.1, 7.2 and 7.3 of the Lease Transaction Agreement have been satisfied or waived by the Landlord Entities and TCC, as applicable; and (ii) the Transaction has been completed to the satisfaction of the Monitor.

#### THE MONITOR CERTIFIES the following:

- 1. The conditions to Closing as set out in section 7.1, 7.2 and 7.3 of the Lease Transaction Agreement have been satisfied or waived by the Landlord Entities and TCC, as applicable; and
- 2. The Transaction has been completed to the satisfaction of the Monitor.
- 3. This Certificate was delivered by the Monitor at _____ [TIME] on _____ [DATE].

ALVAREZ & MARSAL CANADA INC., in its capacity as Court-appointed Monitor of Target Canada Co., *et al.* and not in its personal or corporate capacity

Per:					
	Name:				
	Title				

### SCHEDULE "C"

No.	Location/Address	Province	Land Registry Office	Legal Description	Encumbrances to be Expunged/Deleted
1.	Carrefour De L'Estrie 3050 boul. de Portland, Sherbrook	QC	Land Registry Office for the Registration Division Of Sherbrooke	An immovable located in the City of Sherbrooke, Province of Québec, known and described as being composed of lot numbers 1 395 254, 1 512 226, 1 512 227, 1 512 228, 1 512 229, 1 512 230, 1 512 231, 1 512 232, 2 016 500, 2 016 501, 3 015 081, 3 015 082, 3 027 252 and 3 027 253 of the Cadastre of Québec, Registration Division of Sherbrooke.	The Lease in favour of Zellers Inc. dated August 16, 1985, as amended from time to time and registered by Memorial at the Land Register for the Registration Division of Sherbrooke under the number 327 573, as assigned to Target Canada Co. by an Assignment and Assumption of Lease Agreement dated May 27, 2011 and as further amended by a Lease Amending Agreement dated May 27, 2011 and registered by Notice at the said Register under the number 18 234 130.
2.	Conestoga Mall 550 King St. N., Waterloo	ON	LAND TITLES OFFICE FOR WATERLOO REGION NO. 58	PIN 22283-0265 (LT)  PARTS BLOCKS A, B AND C, PLAN 1434, PART LOT 8, GERMAN COMPANY TRACT, TOWNSHIP OF WATERLOO DESIGNATED AS PARTS 1 AND 2 ON REFERENCE PLAN 58R2220 CITY OF WATERLOO EXCEPT PART 1 ON REFERENCE PLAN 58R10992, PARTS 2 TO 6 INCL., ON REFERENCE PLAN 58R2116 EXCEPT PARTS 1, 2 AND 5 ON REFERENCE PLAN 58R7763 AND PART 1 ON REFERENCE PLAN 58R7763 AND PART 1 ON REFERENCE PLAN 58R17644 (BEING PART OF CONESTOGA ROAD CLOSED BY BY-LAW 633771 AS AMENDED BY 843068); S/T 413351, 543074, 543076, 559295, 575653E, 633780, 645237; SUBJECT TO AN EASEMENT IN GROSS OVER PART OF BLOCK C ON PLAN 1434 DESIGNATED AS PARTS 1 AND 2 ON REFERENCE PLAN 58R17528 AS IN	(a) Instrument No. 680771 Notice of Lease registered May 1, 1980, between Cambridge Leaseholds Limited (Landlord) And K Mart Canada Limited K Mart Canada Limitée (Tenant);  (b) Instrument No. 1335353 Notice of Lease registered May 1, 1997, between Cambridge Leaseholds Limited (Landlord) and Kmart Canada Co./Kmart Canada Cie. (Tenant);  (c) Instrument No. WR616405 Application To Change Name- Instrument registered May 27, 2011, by Zellers Inc.;  (d) Instrument No. WR616406 Notice of Assignment of Lessee Interest in Lease registered May 27, 2011, by Zellers Inc. to Target Canada Co. respecting Notice of Lease 680771;  (e) Instrument No. WR616407 Notice of Assignment of Lessee Interest in Lease registered May 27, 2011, by Zellers Inc. to Target Canada Co. respecting Notice of Lease 1335353;  (f) Instrument No. WR616408 Application (General) Notice of Lease Amending Agreement registered May 27, 2011, by Target Canada Co. respecting Notices of Lease 680771 and 1335353.

No.	Location/Address	Province	Land Registry Office	Legal Description	Encumbrances to be Expunged/Deleted
				WR710905; SUBJECT TO AN EASEMENT IN GROSS OVER PART OF BLOCK C ON PLAN 1434 DESIGNATED AS PARTS 7 AND 8 ON REFERENCE PLAN 58R17644 AS IN WR734991	
3.	Hillcrest Mall 9350 Yonge St., Richmond Hill	ON	LAND TITLES OFFICE FOR YORK REGION NO. 65	PIN 03146-0066(LT)  PARCEL N-1, SECTION M1436; BLOCK N, PLAN M1436, S/T PT BLOCK N, PLAN M1436, PT 1, 66R10783, AS IN LA791337, (S/T LA471061, LA472563, LA481308, LA481408, LA492413, LA525777, LA525907, LA544907, LA572974, LA572975); TOWN OF RICHMOND HILL	(a) Instrument No. LA472563 registered February 3, 1975 being a Notice of Lease between The Cadillac Fairview Corporation Limited, as Landlord and S. S. Kresge Company, Limited, as Tenant. (b) Instrument No. YR1660625 registered June 9, 2011 being an Application to Change Name by Zellers Inc. as it relates to Instrument No. LA472563. (c) Instrument No. YR1660626 registered June 9, 2011 being a notice of Assignment of Lessee Interest in Instrument No. LA472563 by Zellers Inc. to Target Canada Co. (d) Instrument No. YR1660627 registered June 9, 2011 being a Notice of Lease Amending Agreement relating to Instrument No. LA472563.
4.	Kingsway Garden Mall 1 Kingsway Garden Mall, Ste. 60, Edmonton	AB	Edmonton Land Title Office	Title: 142 391 858  Lot 276, Block 7 Plan 1425119  LOT 276 BLOCK 7 PLAN 1425119 - CIVIC ADDRESS: 1 KINGSWAY GARDEN MALL, EDMONTON, AB	(a) Lease 752 165 749 to Target Canada Co. registered November 17, 1975 (b) Caveat re Lease Interest, Etc. 112 175 476 by Target Canada Co. registered June 13, 2011
5.	Les Galeries De La Capitale 5401 boul. des Galeries, Quebec City	QC	Land Registry Office for the Registration Division Of Quebec	An immovable located in the the City of Québec (Borough of Des Rivières), Province of Québec, known and described as being composed of lot numbers 1 145 420, 4 154 209, 4 154 210, 1 147 680, 4 154 208, 4 009 213, 3 583 645, 4 680 540, 5 336 012 and 5 336 013 of the Cadastre of Québec,	The Lease in favour of Zellers Inc. dated May 11, 2001, as amended from time to time and registered by Notice at the Land Register for the Registration Division of Québec under the number 10 375 977, as assigned to Target Canada Co. by an Assignment and Assumption of Lease Agreement dated May 27, 2011 and as further amended by a Lease Amending Agreement dated May 27, 2011 and registered by Notice at the said Register under the number

No.	Location/Address	Province	Land Registry Office	Legal Description	Encumbrances to be Expunged/Deleted
				Registration Division of Québec.	18 177 905.
6.	Metropolis At Metrotown 4545 Central Blvd., Burnaby	ВС	New Westminster Land Title Office	Title BB249940: PID 025-702-254  LOT 1 PLAN BCP6303 - CIVIC ADDRESS: 4545/4575 CENTRAL BLVD, BURNABY & 4800/4820/4880 KINGSWAY, BURNABY	(a) Lease CA3544455 to Target Canada Co. registered January 13, 2014
7.	Mic Mac Mall 21 MicMac Rd., Dartmouth	NS	Land Registration Office for the County of Halifax	All that lot of land situate on the northeastern side of Micmac Boulevard in the City of Dartmouth, County of Halifax, Province of Nova Scotia shown as Lot G-1 on a plan (Servant, Dunbrack, McKenzie & MacDonald Ltd. Plan No. 14-727-0) of survey of Lot G-1, Lot K-1, Lot M-1 and Parcel MB-1, Consolidation and Subdivision of Block L-5, Block L-6 and Parcel L-B (Glencairn Crescent), Lands Acquired by Markborough Properties Inc., signed by Carl K. Hartlen, N.S.LS., dated November 25, 1996; revised March 4, 1997; approved by the Development Officer for the Halifax Regional Municipality March 11, 1997 and filed at the Registry of Deeds for Halifax County as plan number 31984 in drawer 353, said Lot G-1 bearing PID number 00616946.  PID 40173676:  ALL that certain lot of land situated on the southwestern side of Circumferential Highway (Route No. 111) in the City of Dartmouth,	(a) Instrument No. 98673941 registered July 8, 2011 being a Lease and Confirmatory Lease and Amendment to Lease between 4239474 Canada Inc., as Landlord and Target Canada Co., as Tenant.

No.	Location/Address	Province	Land Registry Office	Legal Description	Encumbrances to be Expunged/Deleted
				County of Halifax, Province of Nova Scotia shown as Lot M-1 on a plan (Servant, Dunbrack, McKenzie & MacDonald Ltd. Plan No. 14-727-0) of survey of Lot G-1, Lot K-1, Lot M-1 and Parcel MB-1, Consolidation and Subdivision of Block L-S, Block L-6 and Parcel L-B (Glencairn Crescent), Lands Acquired by Markborough Properties Inc., signed by Carl K. Hurtle; N.S.L.S., dated November 25, 1996; revised March 4, 1997; approved by the Development Officer for the Halifax Regional Municipality March 11, 1997, registered at the Halifax County Registry of Deeds as Plan 31984 in Drawer 353 and said lot	
				being more particularly described in Instrument No. 98673941.  PID 40173684:	
				All that certain block of land on the southwestern side of the Circumferential Drive in the City of Dartmouth, Province of Nova Scotia, shown as Block L-2 on a plan (Servant, Dunbrack & McKenzie Ltd. Plan Number Dr. 4 No. 189) showing Boundaries of a Portion of MicMac Village, signed by Walter B. Servant, N.S.L.S. dated May 5, 1972 being registered at the Halifax County Registry of Deeds as Plan No. 11315 in Drawer 161 and which lot is described as in	
				Instrument No. 98673941.  PID 40173692:  ALL that certain block of	

No.	Location/Address	Province	Land Registry Office	Legal Description	Encumbrances to be Expunged/Deleted
				land on the northwestern side of Manor Drive in the City of Dartmouth, Province of Nova Scotia, shown as Block L-3 on a plan (Servant, Dunbrack & McKenzie Ltd. Plan Number Dr. 4 No. 189) showing Boundaries of a Portion of MicMac Village, signed by Walter E. Servant, N.S.L.S. dated May 5, 1972 being registered at the Halifax County Registry of Deeds as Plan No. 11315 in Drawer 161 which said lot is described in Instrument No. 98673941.	
				All that certain lot of land situate, lying and being on the northern side of Glen Manor Drive in the City of Dartmouth, Province of Nova Scotia, and being Block L-1 as shown on a subdivision plan of MicMac Village, signed by Walter E. Servant, N.S.L.S., dated May 5, 1972 being registered at the Halifax County Registry of Deeds as Plan No. 11315 in Drawer 161 and which said Block L-1 may be more particularly described in Instrument No. 98673941.	
8.	Oakridge Centre 650 West 41 st Avenue, Vancouver	BC	New Westminster Land Title Office	(a) Title BR215693 and (b) Title CA1626173 PID 003-128-687 LOT 7 PLAN 20424 - CIVIC ADDRESS: 650 WEST 41ST AVENUE, VANCOUVER	<ul> <li>(a) Lease CA3544453 to Target Canada Co. registered January 13, 2014</li> <li>(b) Lease CA3544453 to Target Canada Co. registered January 13, 2014</li> </ul>
9.	Place Laurier 2700 boul. Laurier, Quebec City	QC	Land Registry Office for the Registration Division Of	An immovable located in the the City of Québec (Borough of Sainte-Foy- Sillery-Cap-Rouge),	The Lease in favour of Zellers Inc. dated November 3, 1982, as amended from time to time and registered by Memorial at the Land Register for the

No.	Location/Address	Province	Land Registry Office	Legal Description	Encumbrances to be Expunged/Deleted
			Québec	Province of Québec, known and described as being composed of lot number 2 012 430 of the Cadastre of Québec, Registration Division of Québec.	Registration Division of Québec under the number 1 177 851, as amended by a Lease Amending Agreement registered by Notice at the said Register under the number 1 727 258, as assigned to Target Canada Co. by an Assignment and Assumption of Lease Agreement dated May 27, 2011 and as further amended by a Lease Amending Agreement dated May 27, 2011 and registered by Notice at the said Register under the number 18 177 903.
10.	Square One 100 City Center Dr., Mississauga	ON	LAND TITLES OFFICE FOR REGION OF PEEL NO. 43	PIN 13142-0060(LT) Property Description: BLOCKS 1, 16, 19, 20, 21, PLAN 43M-1010 EXCEPT PARTS 1, 2, 3, PLAN 43R-20341, PARTS 1 TO 16, PLAN 43R-33444, PARTS 1 TO 6, PLAN 43R-33815, PARTS 1, 2, 3, 4, 5, 6, 7, 8, 11, 12, 13, 14, 15, 17, 18, 19, 20 PL 43R-35320; SUBJECT TO AN EASEMENT AS IN VS202846 PARTIAL RELEASE BY LT1099979, LT1365500, LT1394307, PR1890934, PR1890937; SUBJECT TO AN EASEMENT AS IN VS288971 BY PARTIAL RELEASE LT1099979, LT1365499, LT1394303, LT1493567, PR189035, PR1890936, PR1890936, PR1890938, PR2477270, PR2477271, PR2477272; SUBJECT TO AN EASEMENT AS IN LT1280198; SUBJECT TO AN EASEMENT AS IN LT1280200; SUBJECT TO AN EASEMENT AS IN LT1280200; SUBJECT TO AN	(a) Instrument No. PR999611 registered January 17, 2006 being a Notice of Lease in favour of Zellers Inc.  (b) Instrument No. PR2017694 registered June 9, 2011 being an Assignment of Lease No. PR999611 by Zellers Inc. to Target Canada Co.  (c) Instrument No. PR2017695 registered June 9, 2011 being a Notice of Lease Amending Agreement relating to Instrument No. PR999611.  (e) Instrument No. PR2665203 registered January 26, 2015 being a Construction Lien in favour of PCL Constructors Canada Inc., as lien claimant.

No.	Location/Address	Province	Land Registry Office	Legal Description	Encumbrances to be Expunged/Deleted
				PR665471;	
				SUBJECT TO AN EASEMENT AS IN PR2477258;	
				TOGETHER WITH AN EASEMENT OVER PT BLOCKS 1, 16, 19, 20 PL 43M-1010 DES PARTS 1, 2, 3, 4, 5, 6, 7, 8, 11, 12, 13, 14, 15, 17, 18, 19, 20 PL 43R-35320 AS IN PR2477266;	
				TOGETHER WITH AN EASEMENT OVER PT BLOCK 16 PL 43M-1010 DES PARTS 1, 2, 3 PLAN 43R-20341 AS IN PR2477266;	
				CITY OF MISSISSAUGA	

No.	Location/Address	Province	Land Registry Office	Legal Description	Encumbrances to be Expunged/Deleted
				PIN 13142-0061(LT) Property Description: BLOCKS 1, 16, 19, 20, 21 PL 43M-1010 EXCEPT PARTS 1, 2, 3, PL 43R-20341, PARTS 1 TO 16 PL 43R-33444, PARTS 1 TO 6 PL 43R-33815, PARTS 1, 2, 3, 4, 5, 6, 7, 8, 11, 12, 13, 14, 15, 17, 18, 19, 20 PLAN 43R-35320; SUBJECT TO AN EASEMENT AS IN VS202846 PARTIAL RELEASE BY LT1099979, LT1365500, LT1394307, PR1890934, PR1890937; SUBJECT TO AN EASEMENT AS IN VS288971 PARTIAL RELEASE BY LT1099979, LT1365499, LT1394303, LT1493567, PR189035, PR1890936, PR1890938, PR2477270, PR2477271, PR2477272; SUBJECT TO AN EASEMENT AS IN LT1280198; SUBJECT TO AN EASEMENT AS IN LT1280200; SUBJECT TO AN EASEMENT AS IN LT1280200; SUBJECT TO AN EASEMENT AS IN PR665471; SUBJECT TO AN EASEMENT AS IN PR665471; SUBJECT TO AN EASEMENT AS IN PR665471; SUBJECT TO AN EASEMENT AS IN PR2477258; TOGETHER WITH AN EASEMENT OVER PT BLOCKS 1, 16, 19, 20 PL 43M-1010 DES PARTS 1, 2, 3, 4, 5, 6, 7, 8, 11, 12, 13, 14, 15, 17, 18, 19, 20 PL 43R-35320 AS IN PR2477266; TOGETHER WITH AN EASEMENT OVER PT BLOCK 16 PL 43M-1010 DES PARTS 1, 2, 3 PL 43R-20341 AS IN PR2477266; CITY OF	(a) Instrument No. PR999611 registered January 17, 2006 being a Notice of Lease in favour of Zellers Inc.  (b) Instrument No. PR2017694 registered June 9, 2011 being an Assignment of Lease No. PR999611 by Zellers Inc. to Target Canada Co.  (c) Instrument No. PR2017695 registered June 9, 2011 being a Notice of Lease Amending Agreement relating to Instrument No. PR999611.  (e) Instrument No. PR2665203 registered January 26, 2015 being a Construction Lien in favour of PCL Constructors Canada Inc., as lien claimant.
	LEGAL_1:33944758.4			MISSISSAUGA (East of Parts 1. 2. 3. Plan	

No.	Location/Address	Province	Land Registry Office	Legal Description	Encumbrances to be Expunged/Deleted
				PIN 13142-0062(LT)  Property Description:  BLOCKS 1, 16, 19, 20, 21 PL 43M-1010 EXCEPT PARTS 1, 2, 3, PL 43R-20341, PARTS 1 TO 16 PL 43R-33444, PARTS 1 TO 6 PL 43R-33815, PARTS 1, 2, 3, 4, 5, 6, 7, 8, 11, 12, 13, 14, 15, 17, 18, 19, 20 PL 43R-35320;  SUBJECT TO AN EASEMENT AS IN VS202846 PARTIAL RELEASE BY LT1099979, LT1365500, LT1394307, PR1890934, PR1890937;  SUBJECT TO AN EASEMENT AS IN VS288971 PARTIAL RELEASE BY LT1099979, LT1365499, LT1394303, LT1493567, PR189035, PR1890936, PR1890938, PR2477270, PR2477271, PR2477272;  SUBJECT TO AN EASEMENT AS IN LT1280198;  SUBJECT TO AN EASEMENT AS IN LT1280200;  SUBJECT TO AN EASEMENT AS IN PR665471;  SUBJECT TO AN EASEMENT AS IN PR667471;  SUBJECT TO AN EASEMENT AS IN PR67477258;  TOGETHER WITH AN EASEMENT OVER PT BLOCKS 1, 16, 19, 20 PL 43M-1010 DES PARTS 1, 2, 3, 4, 5, 6, 7, 8, 11, 12, 13, 14, 15, 17, 18, 19, 20 PL 43R-35320 AS IN PR2477266;  TOGETHER WITH AN EASEMENT OVER PT	(a) Instrument No. PR999611 registered January 17, 2006 being a Notice of Lease in favour of Zellers Inc.  (b) Instrument No. PR2017694 registered June 9, 2011 being an Assignment of Lease No. PR999611 by Zellers Inc. to Target Canada Co.  (c) Instrument No. PR2017695 registered June 9, 2011 being a Notice of Lease Amending Agreement relating to Instrument No. PR999611.  (d) Instrument No. PR2579401 registered August 8, 2014 being a Partial Surrender of Lease No. PR999611, as amended in PR2017695, by Target Canada Co. as to Parts 3, 4, 11, 12, 13, 14, 16, 17, 23, 25, 28 to 31, Plan 43R-35814.  (e) Instrument No. PR2665203 registered January 26, 2015 being a Construction Lien in favour of PCL Constructors Canada Inc., as lien claimant.  (f) Instrument No. PR2676585 registered February 23, 2015 being a Partial Discharge of Construction Lien by PCL Constructors Canada Inc., relating to Instrument No. PR2665203, as to Parts 1-31, Plan 43R-35814.
				BLOCK 16 PL	

No.	Location/Address	Province	Land Registry Office	Legal Description	Encumbrances to be Expunged/Deleted
				43M-1010 DES PARTS 1, 2, 3 PL 43R-20341 AS IN PR2477266;	
				SUBJECT TO AN EASEMENT OVER BLOCKS 1,16,19,20,21, PL 43M1010, EXCEPT PARTS 1 TO 3, PL 43R20341, PARTS 1 TO 16, PL 43R33444, PARTS 1 TO 6, PL 43R33815, PARTS 1 TO 8, 11 TO 15, 17 TO 20, PL 43R35320 IN FAVOUR OF PT BLOCKS 1,16,19,20, PL 43M1010, DES PARTS 1 TO 8, 11 TO 15, 17 TO 20, PL 43R35320 & PT BLOCK 16, PL 43M1010, DES PARTS 1 TO 3, PL 43R20341 AS IN PR2479858; CITY OF MISSISSAUGA (South of Parts 1, 2, 3, 4, 5, 6, 7, 8, 11, 12, 13, 14, 15, 17, 18, 19, 20, Plan 43R-35320)	
				PIN 13142-0055 (LT)	(a) Instrument No. PR999611
				PT BLK 16 PL 43M-1010 DES PTS 1, 2, 3, 4, 5, 6, 7 PL 43R-33444; S/T EASEMENT VS202846 PARTIALLY RELEASED BY LT1099979, PR1890934, PR1890937; S/T EASEMENT VS288971 PARTIALLY RELEASED BY LT1099979, PR1890935, PR1890936, PR1890935, PR1890936, PR1890938; SUBJECT TO AN EASEMENT AS IN LT1280198; SUBJECT TO AN EASEMENT AS IN LT1280200; SUBJECT TO AN EASEMENT IN GROSS OVER PT 2 PL 43R- 33444 AS IN PR1805934;	registered January 17, 2006 being a Notice of Lease in favour of Zellers Inc.  (b) Instrument No. PR2017694 registered June 9, 2011 being an Assignment of Lease No. PR999611 by Zellers Inc. to Target Canada Co.  (c) Instrument No. PR2665203 registered January 26, 2015 being a Construction Lien in favour of PCL Constructors Canada Inc., as lien claimant.

No.	Location/Address	Province	Land Registry Office	Legal Description	Encumbrances to be Expunged/Deleted
				SUBJECT TO AN EASEMENT IN GROSS OVER PTS 1,2,3,4,5,6 & 7, PL 43R33444 AS IN PR2023121; CITY OF MISSISSAUGA	
	STREET OWNED BY CITY OF TORONTO			PIN 13142-0054  PTS BLKS 1, 16, 20 PL 43M-1010 DES PTS 8, 9, 10, 11, 12, 13, 14, 15, 16 PL 43R-33444 TO BE ESTABLISHED AS PART OF THE PUBLIC HIGHWAY SYSTEM TO BE PART OF HAMMERSON DRIVE AND PART OF SQUARE ONE DRIVE BY BY-LAW NO. 0201- 2011 AS IN PR2036706; CITY OF MISSISSAUGA	(a) Instrument No. PR999611 registered January 17, 2006 being a Notice of Lease in favour of Zellers Inc.  (b) Instrument No. PR2017694 registered June 9, 2011 being an Assignment of Lease No. PR999611 by Zellers Inc. to Target Canada Co.  (c) Instrument No. PR2665203 registered January 26, 2015 being a Construction Lien in favour of PCL Constructors Canada Inc., as lien claimant.  (d) Instrument No. PR2676585 registered February 23, 2015 being a Partial Discharge of Construction Lien by PCL Constructors Canada Inc., relating to Instrument No. PR2665203, as to all of Pin: 13142-0054.
	STREET OWNED BY CITY OF TORONTO			PIN 13142-0057  PT BLK 1, 16, 19, 20, 21, 43M1010, DES PTS 1 TO 6 PL 43R33815; S/T 288971VS, PARTIALLY RELEASED BY LT1099979 & PR1890935, PR1890936 & PR1890938; SUBJECT TO AN EASEMENT AS IN LT1280198; SUBJECT TO AN EASEMENT AS IN LT1280200; CITY OF MISSISSAUGA	(a) Instrument No. PR999611 registered January 17, 2006 being a Notice of Lease in favour of Zellers Inc.  (b) Instrument No. PR2017694 registered June 9, 2011 being an Assignment of Lease No. PR999611 by Zellers Inc. to Target Canada Co.  (c) Instrument No. PR2665203 registered January 26, 2015 being a Construction Lien in favour of PCL Constructors Canada Inc., as lien claimant.  (d) Instrument No. PR2676585 registered February 23, 2015 being a Partial Discharge of Construction Lien by PCL Constructors Canada Inc., relating to Instrument No. PR2665203, as to all of Pin: 13142-0057.
11.	Upper Canada Mall	ON	LAND TITLES OFFICE FOR	PIN 03554-0344(LT) PT LT 96 CON 1 (WYS)	(a) Instrument No. R165585 registered October 25, 1974 being a Notice of Lease between Regional

No.	Location/Address	Province	Land Registry Office	Legal Description	Encumbrances to be Expunged/Deleted
	17600 Yonge St., Newmarket		YORK REGION NO. 65	(EG), PARTS 1 & 2, 65R899; PT LT 97 CON 1 (WYS) (EG) PT 1, 65R7420; SAVE AND EXCEPT PARTS 6 & 7, 65R18177; EXCEPT PT 1, 65R18177, SAVE & EXCEPT PT 1, 65R19397.  S/T EASE IN FAVOUR OF PT LOTS 96 & 97 CON 1 (WYS) PT 1, 65R19397, OVER PARTS 2 & 4, 65R19397 AS IN R719694. (AMENDED BY M.MUDIE 2002/1/12)  T/W EASE OVER PT LT 97 CON 1 WYS, PT 1 65R31382 AS IN YR1345574;  S/T EASE OVER PT LT 97 CON 1 WYS, PT 2 65R31382 IN FAVOUR OF PT LT 97 CON 1 WYS, PT 2 65R31382 IN FAVOUR OF PT LT 97 CON 1 WYS, PARTS 1, 2, 6 & 7 65R18178 & IN FAVOUR OF PT LT 97 CON 1 WYS, PARTS 11, 12 & 15 65R18178 AS IN YR1345575;  TOWN OF NEWMARKET	Shopping Centres Limited, as Landlord and Zeller's Limited, as Tenant.  (b) Instrument No. R534641 registered Feb 13, 1990 being a Notice of Agreement to Amend a Lease by Regional Shopping Centres Limited, as Lessor and Zellers Inc., as Lessee relating to Instrument No. R165585.  (c) Instrument No. YR1660620 registered June 9, 2011 being an Application to Change Name relating to Zellers Inc. as it relates to Instrument No. R165585.  (d) Instrument No.YR1660621 registered June 9, 2011 being notice of Assignment of Lessee Interest in Instrument No. R165585 by Zellers Inc. to Target Canada Co.  (e) Instrument No. YR1660622 registered June 9, 2011 being a Notice of Lease Amending Agreement relating to Instrument Nos. R165585 and R534641.
				PIN 03554-0078(LT)  LAND TITLES OFFICE FOR YORK REGION NO. 65	(a) Instrument No. YR1660623 registered June 9, 2011 being a Notice of Lease in favour of Target Canada Co.
				Property Description:	
:				PT LT 96 CON 1 W YONGE ST PT 5, 65R19397,	
				T/W R719692 & R719693;	
				S/T EG15610, EG20073;	
				T/W EASE OVER PT LT 97 CON 1 WYS, PT 1 65R31382 AS IN YR1345574;	
				TOWN OF	

No.	Location/Address	Province	Land Registry Office	Legal Description	Encumbrances to be Expunged/Deleted
				NEWMARKET	

### SCHEDULE "D" - PERMITTED ENCUMBRANCES

"Permitted Encumbrances" means, collectively: (a) any Encumbrances (as defined in the Lease Transaction Agreement) encumbering the fee simple, ownership or Landlord Entity interest of the properties on which the Premises are located, but excluding, for greater certainty, any Encumbrances (as defined in the Lease Transaction Agreement) which were created or caused by the Tenant or were consented to by the Tenant (unless at the request of a Landlord Entity); and any Encumbrances (as defined in the Lease Transaction Agreement) charging the leasehold interest (or the rights of the Tenant as lessee) in and to the properties on which the Premises are located; (b) the Leasebacks; and (c) the Site Plan Agreement.

### SCHEDULE "M" ADDRESSES FOR NOTICE

#### A. For notices to the Tenant:

Target Canada Co. 5570 Explorer Drive Mississauga, ON L4W 0C3

Attn:

Aaron Alt

Email:

aaron.alt@target.com

#### With a copy to:

Osler, Hoskin & Harcourt LLP 100 King Street West 1 First Canadian Place Suite 6200, P.O. Box 50 Toronto, ON M5X 1B8

Attn:

Tracy Sandler

Email:

tsandler@osler.com

#### **B.** For Notices to the Landlord Entities:

Oxford Properties Group 200 Bay Street, Suite 900 Royal Bank Plaza, North Tower Toronto, Ontario M5J 2J2

Attention:

Executive Vice-President, Chief Legal Counsel

Facsimile:

(416) 868-3751

#### For lease matters:

Oxford Properties Group 200 Bay Street, Suite 900 Royal Bank Plaza, North Tower Toronto, Ontario M5J 2J2

Attention:

Senior Vice-President, Real Estate Management

Facsimile:

(416) 868-3751

#### With a copy to:

200 Bay Street, Suite 900 Royal Bank Plaza, North Tower Toronto, Ontario M5J 2J2 Attention:

Vice President, Real Estate Management Legal

Facsimile:

(416) 868-3751

#### And to:

Ivanhoe Cambridge 1001, Square Victoria

Suite C-500

Montréal, Québec H2Z 2B5

Attention:

Claude Sirois, Co-Chief Operating Officer

and Executive Vice President, Quebec

Facsimile:

1-514-841-7762

Email:

Claude.Sirois@ivanhoecambridge.com

With a copy to:

Ivanhoe Cambridge 1001, Square Victoria

Suite C-500

Montréal, Québec H2Z 2B5

Attention:

Claude Gendron, Executive Vice President,

Legal Affairs and General Counsel

Facsimile:

1-514-841-7675

Email:

Claude.Gendron@ivanhoecambridge.com

With a copy to:

Ivanhoe Cambridge 1001, Square Victoria Suite C-500

Montréal, Québec H2Z 2B5

Attention:

Lorna J. Telfer, Executive Vice President,

Legal Affairs and Secretary

Facsimile:

1-514-841-7675

Email:

Lorna.Telfer@ivanhoecambridge.com

#### And, in all cases, with a copy to:

Fasken Martineau DuMoulin LLP 333 Bay Street, Suite 2400 Toronto, Ontario M5H 2T6

Attention:

Aubrey E. Kauffman

Facsimile:

(416) 364-7813

Email:

akauffman@fasken.com

#### And a copy to:

Fasken Martineau DuMoulin LLP The Stock Exchange Tower 800 Square Victoria **Suite 3700** Montréal Quebec H4Z 1E9

Attention:

Nicolas Leblanc and Luc Morin

Facsimile:

(514) 397 - 7600

Email:

nleblanc@fasken.com and lmorin@fasken.com

#### And a copy to:

Thornton Grout Finnigan LLP **Suite 3200** TD West Tower 100 Wellington Street West, P.O. Box 329 Toronto-Dominion Centre

Toronto, Ontario M5K 1K7

Attention:

D.J. Miller

Facsimile:

(416) 304-1313

Email:

DJMiller@tgf.ca

#### C. For notices to the Monitor:

Alvarez & Marsal Canada Inc. Royal Bank Plaza, South Tower 200 Bay Street, Suite 2900 P.O. Box 22 Toronto, ON, M5J 2J1

Attn:

Doug McIntosh

Email:

dmcintosh@alvarezandmarsal.com

#### With a copy to:

Goodmans LLP Barristers & Solicitors Bay Adelaide Centre 333 Bay Street, Suite 3400 Toronto, ON M5H 2S7

Attn:

Jay Carfagnini, Ken Herlin & Melaney Wagner

Email:

jcarfagnini@goodmans.ca, kherlin@goodmans.ca &

mwagner@goodmans.ca

## SCHEDULE "N" RELEASE OF LANDLORD CLAIMS

#### RELEASE OF LANDLORD CLAIMS

TO:

TARGET CANADA CO. (the "Tenant"), TARGET CANADA HEALTH CO., TARGET CANADA MOBILE GP CO., TARGET CANADA PHARMACY (BC) CORP., TARGET CANADA PHARMACY (ONTARIO) CORP., TARGET CANADA PHARMACY CORP., TARGET CANADA PHARMACY (SK) CORP., TARGET CANADA PROPERTY LLC, FRANCHISING LP, CANADA PHARMACY CANADA PROPERTY LP, TARGET CANADA MOBILE LP AND TARGET CORPORATION AND ALL PREDECESSORS IN INTEREST TO THE TENANT UNDER ANY OF THE LEASES (OTHER THAN ZELLERS INC. AND HUDSON'S BAY COMPANY AND THEIR RESPECTIVE SUCCESSORS AND ASSIGNS (collectively, the "HBC Entities", and individually, an "HBC Entity")) AND EACH OF THEIR RESPECTIVE AFFILIATES, DIRECTORS, OFFICERS, EMPLOYEES, AGENTS, SHAREHOLDERS, MEMBERS, PARTNERS, GENERAL PARTNERS, LIMITED PARTNERS, SUCCESSORS AND PERMITTED ASSIGNS (collectively, the "Tenant Affiliate Releasees", and individually, a "Tenant Affiliate Releasee", and collectively with the HBC Entities, the "Tenant Releasees", and individually, a "Tenant Releasee")

FROM:

HILLCREST HOLDINGS INC., MONTEZ HILLCREST INC., OXFORD PROPERTIES RETAIL HOLDINGS II INC., CPPIB UPPER CANADA MALL INC., OMERS REALTY MANAGEMENT CORPORATION, SQUARE ONE PROPERTY CORPORATION, KINGSWAY GARDEN HOLDINGS INC., LES GALERIES DE LA CAPITALE HOLDINGS INC., IVANHOE CAMBRIDGE II INC., CARREFOUR DE L'ESTRIE HOLDINGS, INC., PLACE LAURIER HOLDINGS INC., MIC MAC MALL LIMITED PARTNERSHIP, and OAKRIDGE CENTRE VANCOUVER HOLDINGS, INC. (collectively, the "Landlord Entities", and individually, a "Landlord Entity")

RE:

Lease Transaction Agreement between the Tenant and the Landlord Entities dated February, 26 2015 (as amended, modified, restated and/or supplemented from time to time, the "Lease Transaction Agreement")

#### WHEREAS:

A. One of the Tenant Releasees, the Tenant, and certain of its Affiliates applied for and together with the limited partnerships listed on Schedule "A" to the Initial Order (as defined below) were granted protection by the Ontario Superior Court of Justice (Commercial List) (the "Court") under the Companies' Creditors Arrangement Act, R.S.C. 1985, c. C-36, as amended (the "CCAA"), and Alvarez & Marsal Canada Inc. (the "Monitor") was appointed the monitor of the Tenant, pursuant to an Order of the Court dated January 15, 2015, as amended and restated on February 11, 2015, and as further amended, restated and/or amended and restated from time to time (collectively, the "Initial Order").

- B. On March [5], 2015, the Court issued an Approval Order approving the Lease Transaction Agreement, pursuant to which the Tenant shall surrender and the Landlord Entities shall accept a surrender from the Tenant of those Real Property Interests at the Locations, subject to a Leaseback of all but one of the Premises to the Tenant, on the terms and conditions set out in the Lease Transaction Agreement.
- C. Section 2.3 of the Lease Transaction Agreement contemplates that each of the Landlord Entities shall execute and deliver a release on the Closing Date to the Tenant Releasees pursuant to which the Landlord Entities will: (i) release and forever discharge the Tenant Releasees from all claims arising out of or relating to: (a) the Real Property Interests; (b) the Leases; and (c) the Premises ((a) (c), collectively the "Release Matters"); and (ii) release and forever discharge the Tenant Affiliate Releasees from all claims arising out of or relating to the properties of which any of the Premises forms a part.
- D. The Tenant has surrendered to the Landlord Entities all of its right, title and interest in and to the Leases and the Landlord has accepted same. Accordingly, the Landlord Entities desire to execute and deliver this Release to each of the Tenant Releasees in satisfaction of the foregoing obligation.
- E. Unless otherwise provided for herein, all capitalized terms used in this Release have the meaning ascribed to them in the Lease Transaction Agreement.

**NOW THEREFORE** in consideration of the payment of TEN DOLLARS (\$10.00) and for other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the undersigned agree as follows:

Each Landlord Entity, on its own behalf and on behalf of its Affiliates that own or owned 1. a direct or indirect interest in the real properties of which any of the Premises form a part (collectively, the "Properties", and individually, a "Property") and on behalf of each beneficial owner that now owns an interest in its respective Properties and all of their respective successors and assigns (collectively the "Releasors", and individually, a "Releasor"), hereby forever fully and unconditionally remises, releases, acquits, waives and forever discharges each of the Tenant Releasees from any and all actual or potential claims, demands, complaints, grievances, actions, applications, suits, causes of action, Orders, charges, indictments, prosecutions, informations or other similar processes, assessments or reassessments, judgments, debts, liabilities, expenses, costs, dues, accounts, bonds, damages or losses, contingent or otherwise, whether liquidated or unliquidated, matured or unmatured, disputed or undisputed, contractual, legal or equitable, including loss of value, professional fees, including fees and disbursements of legal counsel, and all costs incurred in investigating or pursuing any of the foregoing or any proceeding relating to any of the foregoing (collectively, the "Claims"), whether known or unknown, howsoever arising of every nature and kind whatsoever that the Releasors ever had, now have or hereafter can, shall or may have against any of the Tenant Releasees in any way relating to or arising from any of the Release Matters but excluding Claims against the Tenant in any way relating to or arising from: (a) the Tenant's obligations or liabilities under the Lease Transaction Agreement which are expressly stated to survive the Closing; or (b) the Leasebacks or any other Closing Documents.

- 2. Each of the Releasors hereby forever fully and unconditionally remises, releases, acquits, waives and forever discharges each of the Tenant Affiliate Releasees from any and all actual or potential Claims, whether known or unknown, howsoever arising of every nature and kind whatsoever that the Releasors ever had, now have or hereafter can, shall or may have against any of the Tenant Affiliate Releasees in any way relating to or arising from any of the Properties.
- 3. For greater certainty, none of the Releasors shall have any Claims in the CCAA Proceedings, or in any subsequent bankruptcy proceedings in respect of any of the Tenant Affiliate Releasees, in connection with the matters which are herein released.
- 4. Each of the Releasors agrees not to make any claims or demands or take any proceedings against any other person, corporation or entity which might claim over against any of the Tenant Releasees, or who might claim contribution or indemnity from any of the Tenant Releasees, in connection with the matters which are herein released. In the event that any of the Releasors hereafter makes any claims or demands or takes any other proceedings against any of the Tenant Releasees or against any person who may claim over or claim contribution or indemnity against any of the Tenant Releasees with respect to any of the matters herein released, this Release may be raised as an estoppel and complete bar to any such claim, demand or proceeding.
- 5. Each of the Releasors acknowledges and declares that: (a) it has had an adequate opportunity to read and consider this Release and to obtain such advice in regard to it as it should consider advisable; (b) it fully understands the nature and effect of this Release; and (c) this Release has been duly executed voluntarily.
- 6. This Release shall not be deemed to be any admission of liability on the part of the Tenant Releasees, and liability is specifically denied by them.
- 7. All of the foregoing shall enure to the benefit of each of the Tenant Releasees and their respective heirs, attorneys, guardians, estate trustees, executors, successors, assigns and representatives and be binding upon each of the Releasors and their respective heirs, attorneys, guardians, estate trustees, executors, successors, assigns and representatives.
- 8. This Release shall be governed by the laws of the Province of Ontario and the federal laws of Canada applicable therein.
- 9. The terms of this Release constitute the entire agreement between the parties relating to the subject matter hereof.
- 10. This Release may be executed by the parties in counterparts and may be executed and delivered by facsimile or electronic transmission and all such counterparts and facsimiles or electronic transmissions shall together constitute one and the same agreement.

[Remainder of Page Intentionally Left Blank]

IN WITNESS OF	WHICH	the parties	have duly	executed th	is Release	this	day	of March,
2015.							_	

HIL	LCREST HOLDINGS INC.
By:	
Σ,	Name:
	Title:
By:	
J	Name:
	Title:
MO	NTEZ HILLCREST INC.
By:	
Dj.	Name:
	Title:
By:	
_ ;	Name:
	Title:
HOL	ORD PROPERTIES RETAIL  LDINGS II INC.
By:	
	N
	Name:
Ву:	Title:
Ву:	Title: Name:
By:	Title:
	Title: Name:
CPP.	Title:  Name: Title:
	Title:  Name: Title:
CPP.	Title:  Name: Title:  IB UPPER CANADA MALL INC.
CPP.	Title:  Name: Title:  IB UPPER CANADA MALL INC.  Name:
CPP.	Title:  Name: Title:  IB UPPER CANADA MALL INC.  Name:

### OMERS REALTY MANAGEMENT CORPORATION

By:	
	Name:
	Title:
By:	
Dy.	Name:
	Title:
	ARE ONE PROPERTY
COR	RPORATION
By:	
	Name:
	Title:
By:	
,	Name:
	Title:
	11110.
IZINI	GSWAY GARDEN HOLDINGS INC.
IZIIV	GSWAT GARDEN HOLDINGS INC.
D	
By:	
	Name:
	Title:
By:	
	Name:
	Title:
LES	GALERIES DE LA CAPITALE
	DINGS INC.
By:	
Dy.	Name:
<b>D</b>	Title:
By:	
	Name:
	Title:

### IVANHOE CAMBRIDGE II INC.

Ву:	
	Name:
	Title:
By:	11110.
Dy.	N
	Name:
	Title:
	RREFOUR DE L'ESTRIE HOLDINGS,
INC.	,
By:	
	Name:
	Title:
By:	
2).	Name:
	Title:
	Title.
Ву:	Name:
-	Title:
By:	
	Name:
	Title:
	MAC MALL LIMITED
PAR	TNERSHIP, by its general partner,
4239	474 CANADA INC.
By:	
	Name:
	Title:
By:	
<i>J</i> -	Name:
	Title:

## OAKRIDGE CENTRE VANCOUVER HOLDINGS, INC.

ву:		
	Name:	 
	Title:	
By:		
-	Name:	
	Title:	

## SCHEDULE "O" RELEASE OF TENANT CLAIMS

#### **RELEASE OF TENANT CLAIMS**

TO:

HILLCREST HOLDINGS INC., MONTEZ HILLCREST INC., OXFORD PROPERTIES RETAIL HOLDINGS II INC., CPPIB UPPER CANADA MALL INC., OMERS REALTY MANAGEMENT CORPORATION, SQUARE ONE PROPERTY CORPORATION, KINGSWAY GARDEN HOLDINGS INC., LES GALERIES DE LA CAPITALE HOLDINGS INC., IVANHOE CAMBRIDGE II INC., CARREFOUR DE L'ESTRIE HOLDINGS, INC., PLACE LAURIER HOLDINGS INC., MIC MAC MALL LIMITED **PARTNERSHIP AND OAKRIDGE CENTRE** VANCOUVER HOLDINGS, INC. AND THEIR RESPECTIVE AFFILIATES, DIRECTORS, OFFICERS, EMPLOYEES, AGENTS, SHAREHOLDERS, MEMBERS, PARTNERS, GENERAL PARTNERS, LIMITED PARTNERS, SUCCESSORS AND PERMITTED ASSIGNS (collectively, the "Landlord Releasees", and individually, a "Landlord Releasee")

FROM:

TARGET CANADA CO. (the "Tenant") and TARGET CORPORATION ("Target")

RE:

Lease Transaction Agreement between the Tenant and the Landlord Entities dated February, 26 2015 (as amended, modified, restated and/or supplemented from time to time, the "Lease Transaction Agreement")

#### WHEREAS:

- A. One of the Releasors, the Tenant, and certain of its Affiliates applied for and together with the limited partnerships listed on Schedule "A" to the Initial Order (as defined below) were granted protection by the Ontario Superior Court of Justice (Commercial List) (the "Court") under the Companies' Creditors Arrangement Act, R.S.C. 1985, c. C-36, as amended (the "CCAA"), and Alvarez & Marsal Canada Inc. (the "Monitor") was appointed the monitor of the Tenant, pursuant to an Order of the Court dated January 15, 2015, as amended and restated on February 11, 2015, and as further amended, restated and/or amended and restated from time to time (collectively, the "Initial Order").
- B. On March [5], 2015, the Court issued an Approval Order approving the Lease Transaction Agreement, pursuant to which the Tenant shall surrender and the Landlord Entities shall accept a surrender from the Tenant of those Real Property Interests at the Locations, subject to a Leaseback of all but one of the Premises to the Tenant, on the terms and conditions set out in the Lease Transaction Agreement.
- C. Section 2.4 of the Lease Transaction Agreement contemplates that the Tenant and Target shall execute and deliver a release on the Closing Date to the Landlord Releasees pursuant to which the Tenant and Target will release and forever discharge all claims arising out of or relating to: (i) the Real Property Interests; (ii) the Leases; (iii) the Premises; and (iv) the properties of which any of the Premises forms a part ((i)-(iv), collectively, the "Release Matters").

- D. The Tenant has surrendered to the Landlord Entities all of its right, title and interest in and to the Leases and the Landlord has accepted same. Accordingly, the Tenant and Target desire to execute and deliver this Release to the Landlord Releasees in satisfaction of the foregoing obligation.
- E. Unless otherwise provided for herein, all capitalized terms used in this Release have the meaning ascribed to them in the Lease Transaction Agreement.

**NOW THEREFORE** in consideration of the payment of TEN DOLLARS (\$10.00) and for other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the undersigned agree as follows:

- 1. Each of the Tenant and Target, on their own behalf and on behalf of their Affiliates and all of their respective successors and assigns (collectively the "Releasors", and individually, a "Releasor"), hereby forever fully and unconditionally remises, releases, acquits, waives and forever discharges each of the Landlord Releasees from any and all actual or potential claims, demands, complaints, grievances, actions, applications, suits, causes of action, Orders, charges, indictments, prosecutions, informations or other similar processes, assessments or reassessments, judgments, debts, liabilities, expenses, costs, dues, accounts, bonds, damages or losses, contingent or otherwise, whether liquidated or unliquidated, matured or unmatured, disputed or undisputed, contractual, legal or equitable, including loss of value, professional fees, including fees and disbursements of legal counsel, and all costs incurred in investigating or pursuing any of the foregoing or any proceeding relating to any of the foregoing (collectively, the "Claims"), whether known or unknown, howsoever arising of every nature and kind whatsoever that the Releasors ever had, now have or hereafter can, shall or may have against any of the Landlord Releasees, in any way relating to or arising from any of the Release Matters but excluding Claims in any way relating to or arising from: (a) the Landlord Entities' obligations or liabilities under the Lease Transaction Agreement which are expressly stated to survive the Closing; or (b) the Leasebacks or any other Closing Documents.
- 2. Each of the Releasors agrees not to make any claims or demands or take any proceedings against any other person, corporation or entity which might claim over against any of the Landlord Releasees, or who might claim contribution or indemnity from any of the Landlord Releasees, in connection with the matters which are herein released. In the event that any of the Releasors hereafter makes any claims or demands or takes any other proceedings against any of the Landlord Releasees or against any person who may claim over or claim contribution or indemnity against any of the Landlord Releasees with respect to any of the matters herein released, this Release may be raised as an estoppel and complete bar to any such claim, demand or proceeding.
- 3. Each of the Releasors acknowledges and declares that: (a) it has had an adequate opportunity to read and consider this Release and to obtain such advice in regard to it as it should consider advisable; (b) it fully understands the nature and effect of this Release; and (c) this Release has been duly executed voluntarily.
- 4. This Release shall not be deemed to be any admission of liability on the part of the Landlord Releasees, and liability is specifically denied by them.

- 5. All of the foregoing shall enure to the benefit of each of the Landlord Releasees and their respective heirs, attorneys, guardians, estate trustees, executors, successors, assigns and representatives and be binding upon each of the Releasors and their respective heirs, attorneys, guardians, estate trustees, executors, successors, assigns and representatives.
- 6. This Release shall be governed by the laws of the Province of Ontario and the federal laws of Canada applicable therein.
- 7. The terms of this Release constitute the entire agreement between the parties relating to the subject matter hereof.
- 8. This Release may be executed by the parties in counterparts and may be executed and delivered by facsimile or electronic transmission and all such counterparts and facsimiles or electronic transmissions shall together constitute one and the same agreement.

[Remainder of Page Intentionally Left Blank]

<b>IN WITNESS OF WHICH</b> the parties have d 2015.	uly executed this Release this day of March,
	TARGET CANADA CO.
	By: Name: Title:
	TARGET CORPORATION
	By: Name: Title:
	itio.

## SCHEDULE "P" RELEASE OF GUARANTEES

#### **RELEASE OF GUARANTEE**

TO:

TARGET CANADA CO. (the "Tenant"), TARGET CANADA HEALTH CO., TARGET CANADA MOBILE GP CO., TARGET CANADA PHARMACY (BC) CORP., TARGET CANADA PHARMACY (ONTARIO) CORP., TARGET CANADA PHARMACY CORP., TARGET CANADA PHARMACY (SK) CORP., TARGET CANADA PROPERTY LLC, **TARGET** CANADA PHARMACY FRANCHISING LP, CANADA PROPERTY LP, TARGET CANADA MOBILE LP AND TARGET CORPORATION AND ALL PREDECESSORS IN INTEREST TO THE TENANT UNDER ANY OF THE LEASES (OTHER THAN ZELLERS INC. AND HUDSON'S BAY COMPANY AND THEIR RESPECTIVE SUCCESSORS AND ASSIGNS) AND EACH OF THEIR RESPECTIVE AFFILIATES, DIRECTORS, OFFICERS, EMPLOYEES, AGENTS, SHAREHOLDERS, MEMBERS, PARTNERS, GENERAL PARTNERS, LIMITED PARTNERS, SUCCESSORS AND PERMITTED ASSIGNS (collectively, the "Tenant Affiliate Releasees", and individually, a "Tenant Affiliate Releasee")

FROM:

IVANHOE CAMBRIDGE II INC., CARREFOUR DE L'ESTRIE HOLDINGS, INC., PLACE LAURIER HOLDINGS INC., MIC MAC MALL LIMITED PARTNERSHIP, and OAKRIDGE CENTRE VANCOUVER HOLDINGS, INC. (collectively, the "Landlord Entities", and individually, a "Landlord Entity")

RE:

Lease Transaction Agreement between the Tenant, the Landlord Entities and other entities party thereto dated February, 26 2015 (as amended, modified, restated and/or supplemented from time to time, the "Lease Transaction Agreement")

#### **WHEREAS:**

- A. The Tenant and certain of its Affiliates applied for and together with the limited partnerships listed on Schedule "A" to the Initial Order (as defined below) were granted protection by the Ontario Superior Court of Justice (Commercial List) (the "Court") under the Companies' Creditors Arrangement Act, R.S.C. 1985, c. C-36, as amended (the "CCAA"), and Alvarez & Marsal Canada Inc. (the "Monitor") was appointed the monitor of the Tenant, pursuant to an Order of the Court dated January 15, 2015, as amended and restated on February 11, 2015, and as further amended, restated and/or amended and restated from time to time (collectively, the "Initial Order").
- B. On March [5], 2015, the Court issued an Approval Order approving the Lease Transaction Agreement, pursuant to which the Tenant shall surrender and the Landlord Entities shall accept a surrender from the Tenant of those Real Property Interests at the Locations, subject to a Leaseback of all but one of the Premises to the Tenant, on the terms and conditions set out in the Lease Transaction Agreement.

- C. Section 5.4 of the Lease Transaction Agreement contemplates that each Landlord Entity which holds a guarantee and/or indemnity from any of the Tenant Affiliate Releasees in connection with any Lease, including without limitation the Guarantees, shall execute and deliver a release on the Closing Date to the Tenant Affiliate Releasees pursuant to which the Landlord Entities will release and forever discharge all covenants, obligations, liabilities, Claims, indemnities and guarantees of the Tenant Affiliate Releasees arising out of or relating to the Guarantees, the Leases, the Premises, the Real Property Interests and/or the properties of which any of the Premises form a part.
- D. The Landlord Entities desire to execute and deliver this Release to the Tenant Affiliate Releasees in satisfaction of the foregoing obligation.
- E. Unless otherwise provided for herein, all capitalized terms used in this Release have the meaning ascribed to them in the Lease Transaction Agreement.

**NOW THEREFORE** in consideration of the payment of TEN DOLLARS (\$10.00) and for other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the undersigned agree as follows:

- 1. Each Landlord Entity, on its own behalf and on behalf of its Affiliates that own or owned a direct or indirect interest in the real properties of which any of the Premises form a part (collectively, the "Properties", and individually, a "Property") and on behalf of each beneficial owner that now owns an interest in its respective Properties and all of their respective successors and assigns, (collectively the "Releasors", and individually, a "Releasor"), hereby forever fully and unconditionally remises, releases, acquits, waives and forever discharges each of the Tenant Affiliate Releasees from any and all actual or potential claims, demands, complaints, grievances, actions, applications, suits, causes of action, Orders, charges, indictments, prosecutions, informations or other similar processes, assessments or reassessments, judgments, debts, liabilities, expenses, costs, dues, accounts, bonds, damages or losses, contingent or otherwise, whether liquidated or unliquidated, matured or unmatured, disputed or undisputed, contractual, legal or equitable, including loss of value, professional fees, including fees and disbursements of legal counsel, and all costs incurred in investigating or pursuing any of the foregoing or any proceeding relating to any of the foregoing (collectively, the "Claims"), whether known or unknown, howsoever arising of every nature and kind whatsoever that the Releasors ever had, now have or hereafter can, shall or may have against any of the Tenant Affiliate Releasees, and all covenants, obligations, liabilities, indemnities and guarantees of the Tenant Affiliate Releasees, in any way arising from or under the Guarantees, the Leases, the Premises, the Real Property Interests and/or the Properties and acknowledges that each the Guarantees shall be of no further force or effect.
- 2. Each of the Releasors agrees not to make any claims or demands or take any proceedings against any other person, corporation or entity which might claim over against any of the Tenant Affiliate Releasees, or who might claim contribution or indemnity from any of the Tenant Affiliate Releasees, in connection with the matters which are herein released. In the event that any of the Releasors hereafter makes any claims or demands or takes any other proceedings against any of the Tenant Affiliate Releasees or against any person who may claim over or claim contribution or indemnity against any of the Tenant

Affiliate Releasees with respect to any of the matters herein released, this Release may be raised as an estoppel and complete bar to any such claim, demand or proceeding.

- 3. Each of the Releasors acknowledges and declares that: (a) it has had an adequate opportunity to read and consider this Release and to obtain such advice in regard to it as it should consider advisable; (b) it fully understands the nature and effect of this Release; and (c) this Release has been duly executed voluntarily.
- 4. This Release shall not be deemed to be any admission of liability on the part of the Tenant Affiliate Releasees, and liability is specifically denied by them.
- 5. All of the foregoing shall enure to the benefit of each of the Tenant Affiliate Releasees and their respective heirs, attorneys, guardians, estate trustees, executors, successors, assigns and representatives and be binding upon each of the Releasors and their respective heirs, attorneys, guardians, estate trustees, executors, successors, assigns and representatives.
- 6. This Release shall be governed by the laws of the Province of Ontario and the federal laws of Canada applicable therein.
- 7. The terms of this Release constitute the entire agreement between the parties relating to the subject matter hereof.
- 8. This Release may be executed by the parties in counterparts and may be executed and delivered by facsimile or electronic transmission and all such counterparts and facsimiles or electronic transmissions shall together constitute one and the same agreement.

[Remainder of Page Intentionally Left Blank]

IN WITNESS OF WHICH the parties March, 2015.	s have duly executed this Release this day of
	IVANHOE CAMBRIDGE II INC.
	By:
	Name:
	Title:
	By: Name:
	Title:
	CARREFOUR DE L'ESTRIE HOLDINGS, INC.
	By:
	Name:
	Title:
	By: Name:
	Title:
	PLACE LAURIER HOLDINGS INC.
	Ву:
	Name: Title:
	By: Name:
	Title:
	MIC MAC MALL LIMITED PARTNERSHIP, by its general partner, 4239474 CANADA INC.
	Ву:
	Name:
	Title:
	By:
	Name: Title:
	A ALLO.

# OAKRIDGE CENTRE VANCOUVER HOLDINGS, INC.

By:			
	Name:	 	
	Title:		
By:			
	Name:	 	
	Title		

### SCHEDULE "Q" SOA

Closing Date: March 6, 2015 Final Adjustment Date: 11:59 pm, March 5, 2015

Final Due to Final Due to Landlord Tenant	\$ 138,000,000.00	\$ 434,949.59 \$ 211,724.51	\$ 165,268.21 \$ 165,457.50	\$ 128,186,91 \$ 118,999.20	\$ 170,057.26 \$ 136,140.29		\$ 658,903.13 \$ 285,945.36	\$ 120,493.03 \$ 64,670.67	\$ 351,283.46 \$ 161,363.87	\$ 97,959.70 \$ 84,936.66	\$ 1,025,029.65 \$ 449,106.97	\$ 55,203.75 \$ 144,759.22	\$ 3,207,334.69 \$ 139,823,104.28			\$5,685,667.67	\$1,763,454.27	\$ 7,449,121.94
Note	Price as per Agreement	3590 Carrefour de l'Estrie ("CE")	3672 Conestoga Mall ("CM")	3717 Metropolis at Metrotown ("MaM")	3697 Mic Mac Mail ("MMM")	3711 Oakridge Centre ("OC")	3755 Place Laurier ("PL")	3565 Upper Canada Mall ("UCM")	3636 Square One Shopping Centre ("SQ1")	3666 Hillcrest Mall ("HM")	3696 Les Galeries de la Capitale ("GC")	3561 Kingsway Mall ("KM")	TOTAL	AMOUNT PAYABLE TO TENANT	Required Transaction Agreement-Related Payments	Prepaid Rent (Mar 6, 2015 - Jun 30, 2015)	Outstanding Lien Payment Related to Property #3666	TOTAL Required Transaction Agreement Related Payments